UNIVERSITY OF CINCINNATI

Date: 2-Jul-2009

I, Kumar Chokalingam

hereby submit this original work as part of the requirements for the degree of:

Doctor of Philosophy

in Biomedical Engineering

It is entitled:

Transgenic Mouse Model: Examination of Healing, Development and

Mechanical Response of Cells

Student Signature: Kumar Chokalingam

This work and its defense approved by:

Committee Chair:

David Butler, PhD David Butler, PhD

Jason Shearn, PhD Jason Shearn, PhD

Marepalli Rao, PhD Marepalli Rao, PhD

Shawn Hunter, PHD Shawn Hunter, PHD

Transgenic Mouse Model: Examination of Healing, Development and Mechanical Response of Cells

A dissertation submitted to the

Graduate School of the University of Cincinnati

in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY (Ph.D.)

in the Department of Biomedical Engineering of the College of Engineering

2009

by

Kumar Chokalingam

M.S., University of Cincinnati, 2007 B.E. (Hons), Birla Institute of Technology and Science, Rajasthan, India, 1999

Committee Chair: David L. Butler, Ph.D.

ABSTRACT

Tracking real-time changes in Col1 and Col2 gene expression could speed up the FTE process and could lead to new strategies to create zonal insertions in repair tendons. Hence double transgenic mice with fluorescent promoters for Col1 and Col2 gene expression were created and used to examine natural healing of patellar tendons, development of knee structures and the mechanical response of primitive MSCs as well as specialized chondrocytes and fibroblasts.

The natural healing study sought to determine how full thickness, mid-substance punch defect injuries affect spatial and temporal patterns of Col1 and Col2 gene expression and biomechanics compared to age-matched, normal patellar tendons. Col1 expression in injured tendons was highest at two weeks after injury. No detectable Col2 gene expression was seen in the mid-substance healing wound site at any time point. Natural healing resulted in inferior repair properties at both 4 weeks and 6 weeks post injury compared to contralateral sham and age-matched tendons.

Monitoring fluorescent changes in murine knees from embryonic through early post-natal development indicates that Col2 expression precedes Col1 expression. There were temporal and spatial changes in the expression of two related growth factors, TGF-β1 and BMP-2, and their downstream markers, pSmad1,5,8 and pSmad2,3 respectively.

Exposing specialized and primitive cell-scaffold constructs to dynamic *compressive strains* produced temporal changes in gene, protein expression and biomechanics. Compression of *chondrocyte-agarose scaffolds* significantly increased ECFP fluorescence, Col2 mRNA expression, type II collagen content and aggregate modulus. ECFP fluorescence and Col2 mRNA expression were found to be positively correlated. Compression of *MSCagarose constructs* significantly increased ECFP fluorescence, type II collagen content and aggregate modulus. Finally, 28 days of compressive stimulation of *fibroblast-agarose*

i

constructs produced significant decreases in GFP-T fluorescence as well as increases in aggregate modulus. The absence of ECFP expression in any of the fibroblast-agarose constructs indicates that cells were not upregulating the Col2 gene.

In a similar way, *tensile stimulation* of specialized and primitive cells altered both gene expression and tensile stiffness. While tensile stimulation of *MSC-collagen constructs* increased GFP-T fluorescence, Col1 gene expression and linear stiffness, tensile stimulation of chondrocyte-collagen constructs decreased ECFP fluorescence and linear stiffness. Absence of GFP-T expressing cells in the chondrocyte-collagen constructs indicates that the cells were not upregulating the Col1 gene.

In summary, natural healing of patellar tendon results in inferior tissue even after 6 weeks of healing. Murine knees exhibit temporal and spatial variations in Col1 and Col2 gene expression. In-vitro mechanical stimulation studies show that specialized cells respond more rapidly and to a greater extent than MSCs when stimulated with signals that they normally experience in-vivo. MSCs respond more rapidly and to a greater extent than specialized cells when they are stimulated with signals they do not normally experience in-vivo. Future studies must identify appropriate mechanical cues to control/modulate MSC, chondrocyte, and fibroblast differentiation in-vitro so that multiple cellular phenotypes can be spatially and temporally produced, resulting in the synthesis of functional extracellular matrices after surgery.

ii

ACKNOWLEDGEMENTS¹

While my name is solely listed as the author of this dissertation, it could have never been completed without the guidance, help, support and efforts of numerous individuals. The following strives to express my gratitude to everyone who has contributed to the person I am today, but will likely fall short of describing my genuine appreciation for all they have done.

First and foremost I am deeply in debt of gratitude to my mentor Dr. David Butler for his continuous counsel, supervision and most importantly his patience throughout the investigation and writing of this dissertation. I will always be grateful for his enthusiasm, unlimited zeal and high expectations that been major driving forces through my graduate career at the University of Cincinnati.

Thanks and gratitude to Dr. Shawn Hunter for mentoring me when I first joined this department. I will always be grateful for your advice and for your role in my professional growth.

Thanks and gratitude to Dr. Jason Shearn for serving on my dissertation committee and his invaluable contributions to this dissertation.

I also extend my sincere thanks to Dr. Marepalli Rao for being a member of my dissertation committee and for his help with the statistics. Though our personal interaction was limited, I have felt Dr. Rao support.

Special thanks to Cindi Gooch for teaching me cell culture techniques and for her during murine cell harvests.

I would also like to thank Chris Frede, Jane Florer and Dr. Wenstrup for their role in the development of the double transgenic mice used in this dissertation.

I would like to take this opportunity to thank Dr's. Mike Griewe and Keith Kenter for their role in mouse surgeries.

¹ This work was supported by grants from the National Institutes of Health AR 46574-06 from NIAMS, EB002361-02 from NIBIB and BRP from given to the University of Cincinnati.

I am also grateful to my fellow graduate students, faculty and staff at the Department of Biomedical Engineering. Special thanks to Joel Collier, Daria Narmoneva, Jeff Johnson, Mathew Dressler, Natalia Juncosa-Melvin, Victor Nirmalanandhan, Nisha Sipes, Matt Stonecash, Tatiana Mavridas, Abhishek Jain, Nat Dyment, Saurabh Datta, Denise Smith, Vinod Kaimal, Balaji Kalyanaraman, Kirsten Kinneberg, Rachana Jain, Karthik Swaminathan, Daniel Bogusweski. I cannot say thanks enough to Linda Moeller, Mike Sanderson, Lori Beth Derenski, and Denis Bailey. Their help and friendship have made this bearable.

Finally, I wish to thank my parents for always being by my side, even though being physically in India, for their love, sacrifices, and continued encouragement throughout the years. This equally applies to my wife.

This dissertation is dedicated to my parents and my son, Nishyanth.

"Imagination is more important than knowledge. Knowledge is limited. Imagination encircles the world."

Albert Einstein In an Interview by George Sylvester Viereck, for the October 26, 1929 issue of The Saturday Evening Post

ORGANIZATION OF DISSERTATION

This dissertation is organized as a collection of several journal manuscripts, each with the customary sections, including a summary, materials and methods, results and discussion. The presentation of these manuscripts is integrated to produce a coherent dissertation with welldefined objectives and clearly stated conclusions.

Chapter 1 presents the background of tendon injuries and the different approaches to their treatment. This is followed by the specific aims and hypotheses of the current work.

Chapters 2 through 9 present 7 journal manuscripts. The first manuscript presented in Chapter 2, describes Col1 and Col2 gene expression and biomechanics of natural healing of punch defects in the patellar tendons. The second manuscript presented in Chapter 3 examines Col1 and Col2 gene expressions and TGF- β 1 and BMP-2 expressions during normal growth and development of knee structures. Chapter 4 presents the development and validation of a bioreactor capable of delivering precise compressive signals. This chapter is part of a bigger manuscript on bioreactors from our lab. The fourth, fifth and sixth manuscript in Chapters 5, 6 and 7 respectively describe in-vitro studies that investigates the effect of compressive stimulation on chondroyctes, MSCs and fibroblasts respectively. The seventh and the eighth manuscript in Chapters 8 and 9 describe in-vitro studies that investigate the effect of tension on MSCs and chondrocytes respectively. Chapter 9 presents discussion of the studies above and Chapter 10 present recommendations for future research in this field.

vii

Table of Contents

Abstractii
Acknowledgmentsv
Organization of Dissertationviii
List of Tables
List of Figuresxiii
Chapter 1 - Background and Literature Review. 1 1.1. Frequency and Significance of Musculoskeletal Injuries 1 1.2. Tendon Function 1 1.3. Structure and Composition of Tendons. 2 1.4. Biomechanics of Tendons 3 1.5. Tendon Injury. 5 1.6. Tendon Healing. 5 1.7. Conventional Treatment Techniques. 8 1.8. Prosthetic/Biologic Tendon Replacement 10 1.9. Tissue Engineering as a Possible Alternative to Conventional Tendon 11 1.10. Functional Tissue Engineering 12 1.11. Challenges Still Remain in Using FTE to Improve Tendon Repair 13 1.12. Need for New Strategies to Improve Tendon Repair 14 1.13. Use of Fluorescent Proteins as Markers for Collagen I and II Gene Expression 15 1.14. Specific Aims and Hypothesis 18
Chapter 2 - Spatial And Temporal Relationships Of Collagen I And II Gene Expression And Repair Biomechanics In Healing Patellar Tendons Of Double Transgenic Mice
2.1 Introduction 28
2.2 Materials and Methods 30
2.4 Results 35
2.5. Discussion
Chapter 3 - Col1 and Col2 Gene Expression During the Development of Murine Knee
Joints
3.1. Introduction
3.2. Materials and Methods 51
3.3. Results 52
3.4. Discussion53
Chapter 4 - Development And Validation Of A Bioreactor For Delivering CompressiveMechanical Stimulation To Functional Tissue Engineered Cartilage624.1. Introduction634.2. Materials and Methods654.3. Results68

4.4. Discussion	70
Chapter 5 - 3D-In-vitro Effects of Compression and Time in Culture on Aggregate Modulus and on Gene Expression and Protein content of Collagen Type II in Murine Chondrocytes 5.1. Introduction 5.2. Materials and Methods 5.3. Results	76 76 78
5.4. Discussion	88
Chapter 6 - Compressive Loading Applied to Double Transgenic MSC-Agarose Constructs Alters Collagen II Gene Expression, Collagen II Protein Expression and Aggregate Modulus 6.1. Introduction 6.2. Materials and Methods 6.3. Results 6.4. Discussion	97 97 98 99 106
Chapter 7 - Influence Of Compressive Loading On Collagen I Gene Expression And Aggregate Modulus In Tissue Engineered Constructs Containing Double Transgenic Fibroblasts Seeded In Agarose 7.1. Introduction 7.2. Materials and Methods 7.3. Results 7.4. Discussion	110 111 112 113 115
Chapter 8 - Tensile Stimulation Of Murine Stem Cell-Collagen Sponge Constructs Increases Collagen Type I Gene Expression And Linear Stiffness 8.1. Introduction 8.2. Materials and Method 8.3. Results 8.4. Discussion	122 123 124 130 135
Chapter 9 - Tensile Strain Downregulates Col2 Gene Expression In Double Transgenie Murine Chondrocytes Seeded In Collagen Sponges 9.1. Introduction 9.2. Materials and Methods 9.3. Results 9.4. Discussion	c 142 143 145 146 148
Chapter 10 - Discussions and Conclusions	149
Chapter 11 - Recommendations 11.1. Measurement of Fluorescence in Constructs 11.2. Differentiation of Cells in Constructs 11.3. Experimental Conditions	171 172 173 173

Bibliography	178
11.5. In-vitro Testing System of Tensile Constructs	177
11.4. Translational of Results from Murine to Rabbit Models	176
11.5. Injury Model in the Mouse Tendon	175
11.4. Growth and Development of Knee Structures	174

List of Tables

Table 2.1.	Structural and material properties for the patellar tendon healing tissues	43
Table 8.1.	Treatment conditions, response measures, and assignment of constructs	126
Table 8.2.	Sequence of primers used for gene expression analysis and product size in base pairs	129
Table 8.3.	Tensile stimulation did not significantly affect construct dimensions	134

List of Figures

Fig 1.1. The organization of the tendon	3
Fig 1.2. Typical stress strain curve for tendon	4
Fig 1.3. Newborn doubly transgenic mouse foot	19
Fig 2.1. Punch defect injury in the mouse patellar tendon	32
Fig 2.2. Tibia embedded in PMMA and held in place by a pin and cone shaped fixture for securing the patella	34
Fig 2.3. GFP-T and ECFP expression patterns in age-matched controls	36
Fig 2.4. GFP-T and ECFP expression patterns in healing tendons	37
Fig 2.5. Typical GFP-T and ECFP expression in contralateral sham tendons	39
Fig 2.6. Punch defect injuries to the central patellar tendon produced lower force-elongation curves	41
Fig 2.7. Punch defect injuries to the central patellar tendon produced lower stress-strain curves	42
Fig 3.1. Col1 and Col2 gene expression in sagittal sections of murine knees	53
Fig 3.2. Patellar tendon morphology.	54
Fig 3.3. BMP-2 expression in sagittal sections of murine knees	55
Fig 3.4. pSmad1,5,8 expression in sagittal sections of murine knees	56
Fig 3.5. TGF- β 1 expression in sagittal sections of murine knees	57
Fig 3.6. pSmad2,3 expression in sagittal sections of murine knees.	58
Fig 4.1. Compressive stimulus bioreactor	67
Fig 4.2. Media temperature, air temperature, and relative humidity within the bioreactor	71
Fig 4.3. Media temperature and chamber relative humidity are restored to appropriate levels within 20 minutes	71
Fig 4.4. Viability of cells cultured in the bioreactor	72
Fig 4.5. ECFP RFUs increase with mechanical stimulation and culture time	72
Fig 5.1. Rounded ECFP-fluorescing cells in chondrocyte-agarose constructs	84

Fig 5.2. Dynamic stimulation increases ECFP expression and Col2 mRNA	85
Fig 5.3. RFU fluorescence and qRT-PCR positively correlated	86
Fig 5.4. Type II collagen content of chondrocyte-agarose constructs	87
Fig 5.5. Aggregate modulus of chondrocyte-agarose constructs	87
Fig 5.6. Cell viability of chondrocyte-agarose constructs	88
Fig 5.7. Fold increase from day 0 in constructs	92
Fig 6.1. Rounded ECFP cells in MSC-agarose constructs	103
Fig 6.2. Dynamic stimulation increases ECFP expression	104
Fig 6.3. Type II collagen content of MSC-agarose constructs	104
Fig 6.4. Aggregate modulus of MSC-agarose constructs	105
Fig 6.5. Cell viability of MSC-agarose constructs	106
Fig 7.1. Rounded GFP-T expressing fibroblasts in agarose	116
Fig 7.2. Compressive stimulation decreases GFP-T expression	117
Fig 7.3. Compressive stimulation increases linear stiffness	118
Fig 7.4. Cell viability of fibroblast-agarose constructs	119
Fig 8.1. Elongated GFP-T fluorescent cells in MSC-collagen constructs	131
Fig 8.2. Tensile stimulation increased GFP-T expression	132
Fig 8.3. Tensile stimulation increased type I collagen gene expression	133
Fig 8.4. GFP-T RFU and Col1 mRNA positively correlated	134
Fig 8.5. Tensile stimulation increased linear stiffness	135
Fig 9.1. Elongated ECFP expressing chondrocytes in collagen sponges	149
Fig 9.2. Tensile stimulation decreases ECFP	150
Fig 9.3. Tensile stimulation decreases linear stiffness	150
Fig 9.4. Cell viability in chondrocyte-collagen	151

Chapter 1

Background and Literature Review

1.1. Frequency and Significance of Musculoskeletal Injuries

Soft tissue (e.g. tendons, ligaments) injuries represent about half of the 32 million musculoskeletal injuries that take place each year in US [4]. Orthopaedic doctors treat about 14.5 million patients with ligament, tendon sprains and joint dislocations [5]. Statistics also reveal that the incidence of tendon and ligament disorders is high. For example Achilles tendon, rotator cuff, and patellar tendon disorders are reported to be around 100,000 per year, 51,000 per year and 42,000 per year respectively [6-8]. Anterior cruciate ligament injuries are reported to be around 125,000 per year [9]. Such injuries also continue to escalate each year [4]. Such injuries can produce serious disability and discomfort to the patients.

The cost associated with soft tissue injuries are substantial, amounting to over \$30 billion in 1998 [4] which includes both direct (hospitals, physicians etc; 86%) and indirect (lost productivity, morbidity etc; 12%). With the incidence of injuries rising each year the associated cost rises as well. Hence the impact associated with injuries is substantial. Successful repair of such soft tissue injuries should not only return their functionality but also their normal structure and composition. Hence knowledge of these tissues in their normal state is essential.

1.2. Tendon Function

Tendons connect bone to muscle and function to effect the movement of lower limbs, upper extremities and head by transmitting forces from the muscle [4, 10-12]. They carry compressive forces when wrapped around bone like a pulley [13-15]. Tendons also serve as shock absorbers, energy storage sties and serve to maintain posture through their proprioceptive properties [12, 14].

1.3. Structure and Composition of Tendons

Tendons, dense fibrous connective tissues, can be described as biphasic materials composed of both solid and fluid components. Tendons are highly hydrated structures in that 65% of the tissue's wet weight is accounted by water [3, 16-17]. Type I collagen is the primary protein of the tendon accounting for not only 60% to 85% of the dry weight but also for 95% of collagen types present [3, 16-18]. Type II collagen is primarily found in the fibrocartilage regions of the tendon insertions into bone and is also present in tendons that wrap around bone [12-15]. Elastin accounts for about 2% of the dry weight and the rest of the tendon mid-substance is made up of proteoglycans, glycoproteins (such as fibronectin, thrombospondin) and other collagen types such as type, III, V, IX, X, XI and XII [3, 12, 17-18]. These minor collagens in tendons have unique functions. For example type V collagen has been shown to regulate the fibril diameter of type I collagen [18-20]. Type III collagen has been shown to be important during early healing and then is believed to then decrease as new type I collagen is synthesized and organized [21-22]. Collagen types IX, X and XII exist with type II collagen in the insertions [18, 23-25] and thought to minimize the stress concentrations when loads are transmitted from soft tissue into bone [18, 26]. The predominant proteoglycan in tendon is decorin [3, 17]. Decorin along with fibronectin is thought to regulate collagen fibril diameter [18, 27-28].

Collagen in the tendon is arranged in hierarchical levels of increasing complexity beginning with the tropocollagen molecule [3]. Soluble tropocollagen molecules first form crosslinks to create insoluble collagen molecules which then aggregate to form collagen fibrils, then fibers, primary bundles and secondary bundles, and finally the whole tendon (Fig.1.1) [29-31]. Scanning and transmission electron microscopy reveals that the fibrils are 100-500 nm in diameter and consist of microfibrils which are 3.6 nm in diameter [29-31]. A collagen fiber is the smallest tendon unit that can be tested mechanically and is visible under light microscopy [29-

31]. Although collagen fibers are oriented longitudinally, fibers also run transversely forming spirals and pleats[12].



Fig. 1.1. The organization of the tendon structure from collagen fibrils to the entire tendon [3]

Such extracellular matrix of the tendon is synthesized and maintained by the tissue's cells, including tenocytes and tenoblasts (immature tendon cells) [3]. These cells account for about 90-95% of the cellular elements of the tendon [1, 3, 12, 16]. Tenoblasts are spindle-shaped and have numerous cytoplasmic organelles, reflecting their high metabolic activity [12]. As they mature, tenoblasts become elongated and transform into tenocytes with lower nucleus to cytoplasm ratio, less cytoplasmic organelles and less metabolic activity than tenoblasts [3]. The other 5-10% of the cells includes the fibrochondrocytes at the insertion sites, synovial cells on the tendon surface and endothelial cells and smooth muscle cells in the endotenon and epitenon [1, 3, 16]. In pathological conditions, other cells such as inflammatory cells, macrophages, myofibroblasts can also be observed [3].

1.4. Biomechanics of Tendons

Tendons transmit force from muscle to bone and act as a buffer by absorbing external forces to limit muscle damage [32-34]. Tendons exhibit high tensile strength [33-37]. Tendons

are viscoelastic tissues that display 20-40% stress relaxation but rather modest creep [38-39]. The stress-strain mechanical behavior of tendon (Fig. 1.2) depends on the number and types of intramolecular and intermolecular bonds or crosslinks in collagen [33-34, 40]. Unstretched collagen fibers and fibrils in tendons display a crimped configuration [33-34, 41]. When the tendon is strained upto 2%, flattening of the crimp pattern occur leading to the initial concave portion (toe region) [33-34, 42-44] (Fig. 1.2). Beyond this point, the intramolecular sliding of the



Fig.1.2. Typical stress strain curve for tendon [1]

collagen triple helices which causes a more parallel fiber orientation, results in tendon deforming in a linear fashion [33-34, 45-46]. If the strain remains <4-5%, the tendon deforms in an elastic fashion, returning nearly to its original length when un-loaded [33-34, 47]. Between 5%-8% strain, microscopic failures occur, and beyond 8% strain, macroscopic failure occur because slippages of lateral adjoining molecules. As the strain, increases the gap between the molecules increases eventually leading to rupture [33-34, 36, 43, 48-49]. After this, complete failure occurs rapidly, and the fibers recoil into a tangled bud at the ruptured end [32-34]. This structurefunction relationship indicates that for repairs to be successful one most recreate the collagen structure found in normal tissues. The material properties of tendon allow them to tolerate stresses that correspond to forces as high as 15 times body weight. Tendons possess ultimate tensile strength, moduli and strain at failure in the range of 60-100 MPa, 500-800 MPa and 13-22% respectively [50-53]. The above parameters serve as normal benchmarks when evaluating tendon repair.

1.5. Tendon injury

Tendon injuries can be classified as either acute or chronic and either direct or indirect [54-55]. Direct acute tendon injuries occur due to trauma, sports and lacerations by sharp objects. Indirect tendon injuries occur due to acute tensile overload and repetitive microtrauma. These indirect injuries generally occur in the musculotendinous junction (sprains, strains and rupture), or to the osteotendinous junction (avulsion fractures or bone detachment) [55]. These indirect injuries predominate over mid-tendon ruptures because healthy tendon can withstand higher tensile loads than the muscle or most bone junctions [54-59].

1.6. Tendon healing

1.6.1. Tendon healing in animal model

Inducing distinct, but reproducible injuries offers an ideal way to study tendon healing in an animal model. The two most common types of tendon injury models studied in the animal models have been the complete and partial transections [55]. Complete transections with a scalpel better simulate the loss of function seen clinically, despite being more local than the fiber disruption observed at surgery [55, 60]. The advantages of complete transection models are that they are generally reproducible in location, pattern, and extent of damage [55, 60] and create a gap that must be closed between the two free ends generally with the help of sutures. This closure of gap is essential for a variety of reasons: 1) to minimize scar tissue formation, 2) hasten collagen synthesis, and 3) avoid the introduction of non-tendinous tissue between the two ends [55, 61]. One major disadvantage in this model, however, is that the resulting repair biomechanics are strongly influenced by suture type, injury location and knot location [55]. Investigators have also created complete transection models without sutures, allowing tendon healing to commence by gap formation. However, gap dimensions and gap shape affect repair biomechanics. As expected, the rate and extent of natural healing slow as the gap increases in length. The tensile strength of gaps exceeding 3 mm in length does not improve over time [55, 62-63]. These types of gaps are ideal for studying the benefits of cell based tissue engineering but only if they have the initial inherent strength to tolerate the expected in-vivo forces.

Partial transection models can also be created in the middle half or along one side of the tissue. Although they are unlikely to occur clinically, injuries in the central half of the tissue leave the edges of the tissue intact, circumventing the need for suture repair and allowing for immediate postoperative tendon mobilization [55, 64-66]. In addition, partial injuries have the advantage of not requiring sutures to reduce gap formation. The resulting gaps in the central region of the tissue also permit the quality of the newly synthesized tissue to be examined separately from the remaining intact tissue. This is why we in our lab have elected to choose this model to study tendon tissue engineering modalities.

1.6.2. Phases in Tendon Healing

Injured tendons heal by a cascade of events which can be classified into three phases; 1) Inflammation, 2) Proliferation, and 3) Remodeling [1, 16, 33-34, 55]. These phases, which are briefly described below, have characteristic cellular, temporal and biomechanical characteristics. These phases overlap and their duration can vary greatly due to location of injury or disease [1, 33-34, 55, 67-68]. Understanding this biology of tendon repair is essential to devise strategies to effectively repair them.

1.6.2.1. Inflammation

The first step in the healing process is the inflammatory phase in which hematoma occurs in the injured sites resulting in blood clots. This process is then followed by the activation of vasodilators, platelets and the release of pro-inflammatory chemicals from circulating mast cells [1, 33-34, 55, 69-70]. These inflammatory cells are attracted to the injury site in the first 24 hours after injury. These cells help in the breakdown of blood clots and in the digestion of

necrotic tissues [34, 70]. New fibroblasts are then recruited to the injured site. Angiogenesis promoting factors are also released to initiate growth of capillary networks within the wound [34, 55, 70-72]. This phase is characterized by increases in DNA, fibronectin, glycosaminoglycans, water, and collagen type III content [55, 67-68, 73-74].

1.6.2.2. Proliferation

The injury site in this phase is composed of a disorganized matrix of granulation tissue with fibroblasts as the predominant cell type, though a small number of macrophages and mast cells still exist. These fibroblasts are engaged in active matrix synthesis as evidenced by an increase in their endoplasmic reticulum [55, 73]. This phase is characterized by peak increases in Type III collagen and DNA concentrations [34, 70].

1.6.2.3. Remodeling

During this phase, the scar tissue becomes more translucent. Fibroblasts decrease in size and show less collagen and glycosaminoglycan synthesis [34, 55]. Collagen fibers also begin to reorient along the long axis of the tendon [75]. As the scar matures, the ratio of type III to type I collagen decreases and increases occur in collagen crosslinks, glycosaminoglycan, water, and DNA content [55, 76]. However, this process occurs slowly, in that healed tendon can take up to a year to approach the functional strength of uninjured tissue [55, 67, 77-78].

1.6.3. Intrinsic vs. Extrinsic Healing

The tendon healing biology is dominated by two theories, extrinsic and intrinsic healing [33-34, 55, 76, 79-81]. The central tenet in extrinsic healing, is that tendon cannot heal on its own. Tendon repair is accomplished by the formation of adhesions, the infiltration of inflammatory cells and fibroblasts, and an extratendinous blood supply [33-34, 55, 80-85]. In the case of intrinsic healing, it is believed that tendon can heal on its own without the formation of adhesions. This is accomplished by the proliferation of epitenon and endotenon cells within the tendon and with the help of an intratendinous blood supply [71, 79, 81, 86-97]. While both theories seem credible, there is no clear evidence as to which theory is correct. It is plausible

that tendon heals by a combination of both extrinsic and intrinsic factors. Which factor dominates might likely depend on tendon type and location, the magnitude of tendon trauma, availability of synovial fluid and a blood supply and degree of tendon mobilization [79, 86, 98].

1.7. Conventional Treatment Techniques

Conventionally, either conservative or surgical methods or tendon replacements have been used to repair tendon injuries. Conservative management involves rest and pain relief, and can include injection of a variety of drugs, including corticosteroids and physiotherapy [99]. This type of treatment can result in prolonged treatment times, possible weakness in the affected area, recurrent injury, and partial loss of function even after extensive rehabilitation [99]. Conservative methods will not work if the injury is serious.

Surgical methods involve repairing the damaged tissues using sutures and when damage is extensive, replacing the damaged tissue with grafts or replacements. There are a variety of different suture techniques used clinically such as Lin locking, Savage, Kessler, Becker, modified Kessler, Tajima, and epitenon suture method [55-57, 77, 100-101]. However these suture techniques have resulted in mixed results [102-104].

Autografts and allografts have been used as tendon replacements. However preclinical studies in animal models using autografts reveal the potential shortcomings of this treatment approach. While short-term studies show improvements in autograft maximum force, longer-term studies reveal no significant further improvements in strength [105]. Another important drawback in using autografts to replace damaged tendon is the creation of a second morbidity site that heals slowly and incompletely. The best example of this is the use of bone-patellar tendon-bone (BPTB) autografts to reconstruct the anterior cruciate ligament (ACL). While the central-third portion of patellar tendon remains the gold standard to reconstruct the ruptured ACL [106] with a success rate approaching 90% [107], harvesting these graft can lead to patellar tendon rupture [108], transverse patellar fractures and medial or lateral patellar subluxation and dislocation [109] in patients. Animal studies also support these clinical

concerns. These wound sites may never return to normal function, even after prolonged periods [110-111]. For example, even 21 months after removing the goat's central PT, the maximum force and maximum stress of the tendon repair tissue are still only 49% and 35% of normal [111]. After 6 months, the maximum force of the healing canine PT defect were about 60% of normal [112]. Moreover, the modulus of this PT repair site continued to decline from almost 90% of normal initially, to 32% and 16% at three and six months, respectively [112]. Histologically studies have demonstrated that the harvest site is not normal. The harvest site was filled with disorganized collagenous scar tissue composed of small collagen fibrils, is more cellular and vascular than the normal tissue [110-112]. It is obvious from all these studies that even after long periods of time, the repair tissue that develops after harvest of a central bone patellar tendon bone (BPTB) autograft is inferior to normal tissue not only in terms of its structural and material properties, but also in terms of its histological and ultrastructural appearance. It is for these reasons that surgeons not only avoid harvesting the same graft when revising a failed ACL reconstruction, but are looking for new and less invasive alternatives.

While allograft reconstruction avoids the need to create a secondary morbidity site, these grafts also pose a risk of transmitting infectious diseases such as hepatitis and HIV to the recipient. Sterilizing techniques such as ethylene oxide or gamma radiation have been used to minimize the risk of disease transmission [113-118]. Though the freeze-dried ethylene oxide-sterilized allografts have been used in various animal models and in humans [113-114, 118], it has some drawbacks. For example ethylene oxide sterilization has been shown to completely degrade the graft and to cause the formation of femoral cysts in patients [118]. Also the effectiveness of this type of sterilization against HIV is under doubt [118]

Gamma radiation of greater than 3 Mrad are necessary to sterilize grafts [115]. However, these high levels of gamma irradiation also significantly reduce allograft strength [116-117]. While delivering a 2 Mrad dose to the graft did not alter its properties, frozen BPTB grafts which were exposed to 3 Mrad of gamma irradiation showed 27% and 40% reductions in maximum

force and strain energy to maximum force respectively. Should the BPTB allografts require even higher dosages of irradiation to avoid any risk of disease transmission, even more significant, dose-dependent reductions can be expected in both initial structural and material properties [115-116]. These reductions in strength were accompanied by significant reductions in hydroxypyridinium crosslink density [116]. The problems associated with treating these allografts have led researchers to look at synthetic graft materials.

1.8. Prosthetic/Biologic Tendon Replacement

Both prosthetic (polyester [119-120], Gore-Tex prosthesis (polytetrafluoroethylene) [121-123] silicone [124-126] and carbon fibers [127-133]) and biologic materials (collagen [134-135]) have been used as tendon replacements. Prosthetic replacement systems have fallen out of favor in orthopaedic practice because of these following problems. Use of these prosthetics resulted in 1) high incidence of rupture or loosening in patients [136-137], 2) in not fully replacing the function of damaged/torn tissues [120], 3) induced growth [127-131] with the characteristics of a prolonged inflammatory reaction and a disorganized ECM [132, 136], 4) having a finite lifetime as their mechanical properties degraded over time [99], 5) raising questions about their long-term bio-compatibility [124-126], 6) producing particles which accumulated in lymph nodes which in some cases inhibited fibroblast growth [133]. These problems with synthetic grafts have lead researchers to seek our tendon replacements that are biological in nature.

Because of the above-mentioned problems with synthetic scaffolds it would, therefore, be desirable to have a functional tendon replacement that is biologic in origin, which not only integrates into the host but also does not induce inflammatory reaction [138]. Scaffolds made from collagen have been used as biologic tendon replacements [134-135]. Use of collagen scaffolds to treat Achilles tendon gap defects led to tissue ingrowth [135]. The collagen scaffold was completely resorbed by 10 weeks post implantation. The initial strength of these tendon repairs were only 11% of normal, but increased to 80% by 20 weeks [134]. However use of

such biological replacements has some drawbacks. 1) Long term evaluation of repairs show that the strength of the repairs dropped to about 66% of normal [134]. 2) Collagen scaffolds cross-linked with glutaraldehyde resulted in the inducing inflammatory response and resulted in the formation of a fibrous capsule [135]. 3) The strength of the repairs were not significantly different from the repairs achieved by simply reattaching the Achilles tendon [134-135]. 4) Factors such as age, cell proliferation capacity affect the resultant repair from these biologic materials [139-140]. New concepts such as tissue engineering may now offer potential cell-assisted repair approaches to overcome such impairments.

1.9. Tissue Engineering as a Possible Alternative to Conventional Tendon Repair

Given these difficulties with using certain biological and synthetic scaffolds to improve tendon repair, it is not surprising that investigators are using cell-based tissue engineering approaches to treat tendon injuries. Tissue engineering offers an attractive alternative to these clinical problems and has been studied in a variety of animal models. Tissue engineering has been defined as *"a field of science that makes use of biological and/or biocompatible synthetic materials in conjunction with cells to create new tissues or biologic substitutes to serve as functional tissue replacements"* [141].

Investigators have attempted to enhance the healing response of tendons through the use of engineered scaffolds mixed with either differentiated cells such as tenocytes or undifferentiated cells such as mesenchymal stem cells. Several groups have explored tenocyte seeded constructs [142-143]. While these constructs have shown to improve healing[144], still concerns exist. For example factors such as cell seeding density, adhesion, viability, and extracellular matrix production might play a role in the determining the mechanical properties of the resultant tissue engineered constructs [51, 145-146]. There are concerns about cell source because of the creation of a secondary morbidity site to harvest the autologous cells and the time needed for cell expansion in-vitro.

A second approach is the use of mesenchymal stem cells (MSCs) to create tissue engineered constructs [51, 147-156]. The main drawback with the use of MSCs is the question about controlling differentiation. Differentiation towards the fibroblast phenotype can be achieved through the use of growth factors as well as mechanical stimulation [151-155]. Although MSC-seeded scaffolds placed in a tissue defect do improve healing [156] the tissue stiffness and strength do not yet meet in-vivo loading demands [157-159] and questions of safety remain since ideal conditions to create the desired ECM is not fully understood. Similar to the use of tenocytes factors such as scaffold, cell seeding, adhesion, viability, and extracellular matrix production also play a role. Researchers continue to investigate the optimal combination of scaffold, cells and stimulation (either chemical or mechanical or a combination of both) to augment tendon healing. What must also be included, however, is a better understanding of the functional environment into which these constructs are to be placed at surgery.

1.10. Functional Tissue Engineering

Despite the potential advantages of tissue engineering, investigators now recognize the need to adapt the field to include tissue function in the design of tissue engineered repairs. This has resulted in the creation of a new branch of tissue engineering called functional tissue engineering (FTE) [159] whose goal is to enhance tissue stiffness and extracellular matrix synthesis and organization as well as shorten the time needed to fabricate these engineered tissues. "FTE emphasizes the importance of biomechanical considerations in the design and development of cell and matrix-based implants for soft and hard tissue repair" [160]. Functional Tissue Engineering ideally seeks to regenerate or at least functionally repair damaged and diseased tissues, in order to ensure their normal mechanical performance after surgery. Our group first described these criteria for all load-bearing tissues. We emphasized the need to record normal in-vivo tissue forces for different activities of daily living (ADLs) to serve as design parameters for new generations of reparative tissue constructs [156, 160-161]. FTE also emphasizes the need to mechanically precondition the tissue in culture so as to acclimate these

constructs to their future in-vivo loading regimes. Numerous investigators have cited the beneficial effects of applying FTE principles to various types of cell-seeded constructs and tissue models [162-170]. Our group and others have shown the beneficial effects on linear stiffness and collagen expression after applying dynamic, uniaxial tensile as well as compressive strains to constructs seeded with mesenchymal stem cells (MSCs) and chondrocytes, respectively [167, 170-173]. We have also demonstrated that mechanically stimulating MSC-sponge constructs in culture with aspects of the in-vivo strain signal and then implanting these constructs in patellar tendon defects significantly improves the resulting repair biomechanics 12 weeks after surgery [171]. As a result, FTE principles and strategies hold much promise in advancing tendon tissue engineering toward the ideal goal of regenerating normal tissue with normal properties in the functional range of loading.

1.11. Challenges Still Remain in Using FTE to Improve Tendon Repair

At the same time, tissue engineers still face major challenges in creating a truly functional tendon mid-substance and insertion into bone. 1) <u>Restoring the zonal insertion site</u>. Using FTE principles, we have already been able to introduce mechanically-stimulated MSC-collagen sponge constructs to improve repair of rabbit central-third patellar tendon defects. While the 12-week repairs actually match normal PT tangent stiffness up to 150% of peak invivo forces [171], these 12-week repairs still do not recreate the complex fibrocartilaginous zonal insertions of the tendon into bone. Overcoming this challenge requires a new strategy that better controls cell phenotype and matrix expression and assembly. 2) <u>Native tendon biomechanics and biological benchmarks against which we can compare our repairs</u>. 3) <u>Scaffold material</u>. There is no universally agreed upon biomaterial to use as a tendon FTE scaffold. Either current scaffolds need to undergo tests specific to tendon structure-function or investigators need to develop new tendon-specific scaffolds. 4) <u>Cell and culture conditions</u>, Cell type, seeding conditions, and mechanical and chemical culturing conditions need to be

determined in order to optimize the environment for cells in culture. 5) <u>Design parameters to</u> <u>judge successful outcomes.</u> Methods to assess the effectiveness of tissue engineering approaches also need to be developed. We must ascertain whether the tendon has been truly regenerated or at least functionally repaired and at long-enough time periods after surgery.

1.12. Need for New Strategies to Improve Tendon Repair

Although these FTE constructs created in our lab matched or exceeded mechanicallybased tissue engineered design parameters, parameters other than mechanical are also critical to restore. Clearly we need new strategies to create repair tendons with zonal insertions and to speed the FTE process. Below are several proposed strategies.

1) One strategy to create insertions is to introduce primitive cells such as bone marrowderived mesenchymal stem cells or MSCs into biological scaffolds and then precondition these MSC-based constructs to produce tendon like cells in the mid-substance and fibrocartilage-like cells near the attachments. We might accomplish by using either hybrid biomaterials or different stimulation methods (mechanical and or chemical) in the mid-substance vs. ends.

2) Another possible strategy would be to use specialized cells (fibroblasts or chondroctyes) in different regions of the constructs and then apply mechanical strains and other culture conditions to create constructs with tendon-like cells in the mid-substance and fibrocartilage-like cells in the insertions.

3) Still a third strategy for tissue engineers would be to understand tendon midsubstance and insertion biology during normal growth and development and then seek to mimic these events during natural healing or with cell-based therapy. However, our knowledge of these biological processes in different regions of the tissue is still quite limited. Understanding these events could lead to more meaningful therapies that accelerate repair to both the tendon mid-substance and its zonal insertion into bone.

Unknown in this process are the functional loads acting on the tendon during development. The lack of such data hampers our ability to select appropriate mechanical

signals to properly stimulate the cell-based tissue engineered constructs (TECs) in-vitro. Tissue engineers may ultimately need to identify those mechanical signals that cause the TEC to recreate not only regional variations in collagen architecture but also spatial changes in cell phenotype that promote synthesis and maintenance of appropriate extracellular matrix (ECM).

Systematically monitoring relevant signaling remains a very complex and costly tissue engineering process. Currently, this in-vitro stimulation process is slow and serial in nature. This process is also difficult to optimize because various culture conditions like cell and scaffold type as well as environmental stimuli (e.g. mechanical and chemical) must be tested to understand their interactions and how various combinations affect the synthesis of appropriate ECM. These extra steps in the design phase also make tissue engineering quite expensive and time consuming. If these cell-based approaches are to eventually become clinical tools for the orthopaedic surgeon, what is likely required is a novel approach to determine the effects of these environmental stimuli (e.g. specific loads and strains) on cell phenotype in real or at least near-real time.

One example of how this strategy might be applied relates to regional differences in collagen types in tendon. Since Types I and II collagen are major structural proteins found in the tendon mid-substance and insertion site, respectively, we need to understand spatial and temporal expression of these genes in normal, injured and repairing tendons as well as in tendon TECs in culture. Developing a real-time technology to monitor these genes would aid in this effort.

1.13. Use of Fluorescent Proteins as Markers for Collagen I and II Gene Expression

We and others have been linking fluorescent proteins to various gene promoters to visualize expression of genes in real time. Such promoter-fluorescent protein markers have been used to examine embryonic and skeletal development, to distinguish cells in different stages of differentiation, and to conduct cell culture studies [174-178]. We have chosen to

initially use this fluorescent technology in the mouse model to track expression of two important structural genes, Collagen I and II, in real time during development and healing.

The mouse containing the transgene pOBCol3.6GFPtpz was acquired courtesy of David Rowe, University of Connecticut Health Center. This transgene contains a 3.6 Kb fragment of the rat Col1a1 promoter, enhancer sequence and GFP-T. GFP-T expression in these mice is evident in skin, tendon and osseous tissues and indicates upregulation of the Col1 gene [176].

To monitor Col2 expression in the tendon, we have utilized ECFP as a reporter. Plasmid pCol2-ECFP was derived from the expression gene containing the mouse type II collagen promoter and enhancer [179] (provided by W. Horton, Northeastern Ohio College of Medicine). The β -gal gene was replaced with ECFP (Clontech, Palo Alto, CA). Orientation of the ECFP was confirmed by sequencing, and ECFP function was tested in cellulo by transient transfection of chondrocytes in a primary culture from rabbit cartilage. The pCol2-ECFP construct was injected into mice blastocysts. The founder mice identified by PCR showed high levels of expression based on fluorescence in cartilaginous tissue as viewed through a fluorescence stereomicroscope (MZFL3, Leica, Germany)

Mice transgenic for either pOBCol3.6GFPtpz or pCol2-ECFP were then bred to produce double transgenic animals. No apparent phenotypic differences were observed between the doubly and non-transgenic mice. GFP-T fluorescence was apparent in ossifying tissues and ECFP fluorescence was observed in cartilage (Fig. 1.3). This novel reporter mouse has permitted us to visualize, in near real time, either a fibroblast program or a chondrocyte program within a population of cells. Using these mice we have been analyzing Col1 and Col2 gene expression levels during in-vivo development, injury and repair. Our ability to also quantify these gene expression levels in near-real time in culture holds the potential for screening constructs in culture to determine how gene expression patterns change with alterations in cell and scaffold type as well as environmental stimuli). Such technology could ultimately speed development of new repair therapies.



Fig. 1.3. Newborn doubly transgenic mouse foot. GFPT fluorescence is apparent in ossifying tissue (A), while ECFP fluorescence is apparent in cartilage (B).

Using these double transgenic mice, the first part of my dissertation has been focused on observing gene expression during healing by inducing punch biopsy injuries in the patellar tendon of the doubly transgenic mouse and then examining spatial and temporal changes in Col1 and Col2 gene expression in and around the injury site. While the small size of the mouse patellar tendon precludes us from confidently reproducing central one-third defect injuries that occur clinically during harvest of ACL grafts, the punch defect in a superficial PT involves minimal surgery and is thus less invasive than injuries that might be induced in the ACL or in other deeper tendon models. Next I focused on understanding Col1 and Col2 gene expression patterns in normal development and natural healing. I examined these patterns in embryonic to early post natal tendon development.

Using cells from these transgenic mice, the second part of my dissertation research has focused on how mechanical signaling of cell-based TECs in culture affects the plasticity of primitive MSCs and specialized chondrocytes and fibroblasts. I decided to de-couple the complex signals acting on tendon by initially investigating how controlled compressive and tensile displacements influence near real-time Col1 and Col2 gene expression in two cell-based constructs of simple shapes. Although these cylindrical- and rectangular-shaped constructs do not resemble actual tendon geometry of the tendon, these simplifications allow me to carefully apply these controlled displacement patterns to the constructs in a simple fashion that allows their individual effects to be quantified.

1.14. Specific Aims and Hypotheses

When adult tendons are injured, the body repairs the tissue by a complex wound healing cascade that results in the formation of a scar [55, 67-68, 81, 96, 180-182]. The mechanical properties of this scar tissue have been shown to be inferior when compared with normal tissue [55, 181, 183-185]. While natural healing of tendons has been studied [55, 67-68, 81, 96, 180-185], little knowledge currently exists about spatial and temporal patterns of Col1 and Col2 gene expressions during natural healing after injury to tendons. This lack of knowledge is the basis for our first set of research questions. *How does introducing a controlled injury in the mid-substance of the PT affect Col1 and Col2 gene expression over time? How does this injury affect repair biomechanics over time?* We chose to study the effect of injury to the patellar tendon of the doubly transgenic mouse because the patellar tendon is easily accessible unlike other tendons. Also the patellar tendon inserts proximally into the patella and distally into the tibia, which will facilitate mechanical testing of the small mouse tendon after injury. In order to address the research questions above, we developed the following specific aim and hypotheses.

Specific Aim 1: Create a punch defect in the mid-substance of the PT in a 10-week old DT mouse that has nearly achieved skeletal maturity. Evaluate how earlier time intervals after injury (1, 2, and 4 weeks) change Col1 and Col2 gene expression and how later time intervals post surgery (4 and 6 weeks) influence repair biomechanics. Compare these results to gene expression patterns and biomechanics in the contralateral shams and age- matched PTs.

<u>Hypothesis 01:</u> Compared to age-matched PT and sham injuries in the contralateral PT, a punch defect injury increases Col1 gene expression at 1 and 2 weeks post surgery which then declines at 4 weeks post surgery as healing enters the remodeling phase.

<u>Hypothesis 02:</u> A punch defect injury produces no detectable Col2 gene expression in the mid-substance healing wound site.

<u>Hypothesis 03</u>: Repair structural and material properties of the injured PT increase over time but remain significantly lower than structural and material properties for paired sham controls and age-matched controls.

<u>Hypothesis 04</u>: Structural and material properties of shams will not be different from the structural and material properties of age-matched controls.

Results from the natural healing study showed increases in Col1 gene expression in the injured PT mid-substance from 1 to 2 weeks with a subsequent decrease by 4 weeks. This temporal pattern seems to follow the temporal phases of healing. Results also indicate that the guality of repair is inferior. Hence strategies are necessary to improve the guality of repair and one such strategy is tissue engineering. Recently, tissue engineers have turned to developmental biology as a way to try and improve the repair of tendons [186]. The premise for this new approach is that understanding normal growth and development of tendons might help in designing tissue engineering constructs that facilitate later repair. In particular, understanding similarities in expression of extracellular matrix genes and growth factors might allow tissue engineers to develop more effective treatments to augment adult tendon healing. "A bio-mimetic approach to tissue engineering or emulation of some aspects of normal tissue development and remodeling, could be a key to future success in the field of tissue engineering" [186]. This need to understand the expression of extracellular matrix genes and growth factors during normal growth and development led to our next research question. How does embryonic to early post-natal development affect Col1 and Col2 gene expression levels in the murine **patellar tendon?** Two growth factors TGF-β1 and BMP-2, [187-189] and their signaling pathways through PSmads [187, 190] have been shown to be important during normal knee development. Treating tendon fibroblasts with TGF- β increases the phosphorylation of smad2/3 (194), suggesting that TGF- β signaling is mediated through pSmad2,3 [190]. BMP signaling has

been shown to be mediated through pSmad1,5,8 [187]. Details of control of normal tendinogenesis by TGF β 1 and BMP-2 are not known, nor when and where pSmad2,3 and pSmad1,5,8 are expressed. Determining these spatiotemporal activation patterns during tendon development could provide cues to enhance tendon healing using TECs. These led to the next set of research questions. *How does normal knee development affect spatial and temporal expression levels of TGF-\beta and pSmad2,3? <i>How does normal knee development affect spatial and temporal expression levels of TGF-\beta and pSmad2,3? <i>How does normal knee development affect spatial and temporal expression levels of BMP-2 and pSmad1,5,8*? In order to address these questions we developed the following specific aims and hypothesis.

Specific Aim 2: Determine how murine patellar tendon (PT) development from embryonic (E17.5) through early post-natal (P1, P3 and P14) ages influences expression of both the Col1 and Col2 genes, as well as TGF- β 1 and BMP-2, and their upstream signaling molecules, pSmad1,5,8 and pSmad2,3, respectively.

<u>Hypothesis 05:</u> Col1 and Col2 will be expressed at similar times during PT development in utero.

<u>Hypothesis 06:</u> Col1 and Col2 gene expressions increase until P3 with a subsequent decrease by P14.

<u>Hypothesis 07:</u> TGF- β 1 will be expressed in the mid-substance of the PT while BMP-2 will be expressed in the PT insertions, femur and tibia at all ages.

<u>Hypothesis 08:</u> pSmad1,5,8 expression will mirror TGF-β1 expression.

Hypothesis 09: pSmad2,3 expression will mirror BMP-2 expression

Results from growth and development studies of knee structures indicate that there are temporal changes in Col1 and Col2 gene expressions and that there are regions in the PT insertions which have co-expression of Col1 and Col2. These observations lead me to the following research questions. *Can such temporal changes in gene expression be recreated during in-vitro culture using both primitive (MSC) and specialized cells (chondrocytes,*

fibroblasts)? Can such co-expression of Col1 and Col2 also be recreated during in-vitro culture in primitive and specialized cells?

Col1 and Col2 gene expression can be modulated in-vitro by the use of either mechanical or chemical stimulation. Since mechanical stimulation has been the focus of the tissue engineering efforts in our lab, I decided to use tensile and compressive signals to modulate collagen gene expression in cells.

The first set of in-vitro studies were designed to evaluate how compressive stimulation on MSCs, chondrocytes and fibroblasts affect Col2 and Col1 gene expression levels. Researchers have studied the effects of compressive stimulation of, chondrocytes [148, 165, 168-169, 191-197], MSCs [198] and fibroblasts [199]. Dynamic compression of chondrocyteagarose constructs increased type II collagen synthesis, gene expression, synthesis of GAG as well as compressive mechanical properties [148, 165, 168-169, 191-197]. Preliminary studies performed in our lab using chondrocytes from single transgenic ECFP+ mice show that dynamic compression (10% peak strain at 1 Hz for 1 hr followed by a 1 hr rest period repeated for 6 hours/day) induces significant increases in ECFP Relative Fluorescence Units (RFUs) (Col2 gene expression) at both day 7 and 14 compared to non-stimulated controls.

Groups have examined the effects of compressive stimulation on mesenchymal stem cells. Angele et al [198] found that applying 14 and 28 days of cyclic hydrostatic pressure to TECs containing human marrow-derived mesenchymal progenitor cells significantly increased both proteoglycan and collagen contents. Cyclic compressive loads (sinusoidal waveform of 0.33 Hz to a peak stress of 7994 Pa over a period of 4 hours/day) applied to TECs composed of mesenchymal progenitor cells in 3-D scaffolds significantly increased chondrogenic differentiation [200]. Static and dynamic compressive loads applied to TECs containing embryonic limb-bud mesenchymal cells in agarose increased chondrogenic differentiation [201-202]. Cyclic compressive displacements applied to constructs formed using rabbit bone-marrow derived mesenchymal stem cells in agarose induced chondrogenesis in-vitro [203]. Investigators
have also shown that dynamic compressive displacements applied to rabbit MSC-agarose constructs elevated Col2 and aggrecan expression levels. While chondrogenesis does occur in MSCs the amount of the matrix formation and the mechanical properties are lower than that produced by chondrocytes under the same conditions [204]. These studies taken together with studies from the previous paragraph provide evidence that application of compressive stimulation increases Col2 gene expression in chondrocyte and MSC constructs.

Investigators have also examined the effects of compressive stimulation of fibroblast seeded in 3-D constructs. Application of 10% compressive strain to human periodontal ligament fibroblasts in-vitro [205] also decreases Col1A1 RNA expression as well as type I collagen and fibronectin protein expression. Cyclic hydrostatic pressure applied to tendon fibroblasts expanded from rat Achilles tendon has also led to increased Col2 and aggrecan mRNA expression with an associated decrease in Col1 mRNA expression [206]. All of these studies provide evidence that compression decreases Col1 gene expression levels and increase Col2 gene expression levels in fibroblasts.

Given these findings, we then sought to determine if compressive stimulation 1) would increase Col2 gene expression, protein and compressive properties, in doubly transgenic MSCs and chondrocyte constructs and 2) decrease Col1 gene expression and increase Col2 gene expression in fibroblast constructs. These studies motivated several research questions. **Does dynamic compression of murine MSC and chondrocyte constructs increase ECFP** expression? Does this increase in ECFP expression correspond to increases in Col2 mRNA expression in these constructs at the same time interval? Do these expected increases in ECFP and qRT-PCR expression levels lead to an increase in Type II collagen protein content and later increases in aggregate modulus of these constructs? Can dynamic compression of fibroblasts drive these cells toward a chondrogenic phenotype?

In order to address these questions we developed the following specific aims and hypotheses.

Specific Aim 3: Examine how dynamic compression (0% vs. 10% peak strain at 1 Hz for 1 hr followed by a 1 hr rest period repeated for 6 hours/day) of TECs composed of double transgenic chondrocytes cultured in 3-D agarose gels affects ECFP expression, Col2 mRNA expression, type II collagen content and aggregate modulus of the constructs after 0, 7, 14, 21 and 28 days of stimulation.

Compared to non-stimulated controls at equivalent time points in culture, dynamic compressive stimulation of chondrocyte-agarose TECs

<u>Hypothesis 10</u>: Increases ECFP fluorescence.

Hypothesis 11: Increases Col2 mRNA expression.

<u>Hypothesis 12</u>: Increases type II collagen content.

And

Hypothesis 13: Increases aggregate modulus.

Also we propose that,

<u>Hypothesis 14</u>: ECFP fluorescence as quantified by relative fluorescence units (RFUs) and Col2 mRNA will be linearly correlated.

Specific Aim 4: Examine how dynamic compression (10% peak strain at 1 Hz for 1hr followed by a 1hr rest period repeated for 6hours/day) of constructs containing double transgenic MSCs in 3-D agarose gels affects the cell's ECFP expression, type II collagen content, as well as the construct's aggregate modulus after 0, 7, 14, 21 and 28 days of stimulation. Compare Col2 gene expression levels, type II collagen content and aggregate modulus in these MSC-agarose constructs to those observed in chondrocyte-agarose constructs exposed to identical conditions

Compared to non-stimulated controls at equivalent time points in culture, dynamic compressive stimulation of MSC-agarose TECs

<u>Hypothesis 15</u>: Increases ECFP fluorescence.

Hypothesis 16: Increases type II collagen content.

<u>Hypothesis 17</u>: Increases aggregate modulus.

Compared to chondrocyte seeded constructs exposed to similar conditions, MSC seeded constructs will have

<u>Hypothesis 18</u>: Delayed ECFP fluorescence.

<u>Hypothesis 19</u>: Lower ECFP fluorescence.

Hypothesis 20: Lower type II collagen content.

<u>Hypothesis 21</u>: Lower aggregate modulus.

Specific Aim 5: Examine how dynamic compression (10% peak strain at 1 Hz for 1 hr followed by a 1 hr rest period repeated for 6 hours/day) of TECs containing double transgenic murine fibroblasts in agarose gels affects ECFP and GFP-T expression levels and aggregate modulus after 0, 7, 14, 21 and 28 days of stimulation.

Compared to non-stimulated controls at equivalent time points in culture, dynamic compressive stimulation of fibroblast-agarose TECs

<u>Hypothesis 22</u>: Decreases GFP-T fluorescence.

Hypothesis 23: Increases ECFP fluorescence.

<u>Hypothesis 24</u>: Increases aggregate modulus.

The second set of in-vitro studies were designed to evaluate how tensile stimulation of MSCs and fibroblasts affect Col1 and Col2 gene expression levels. Researchers have studied the effect of tensile stimulation on MSCs and chondrocytes. Our lab and others have applied FTE principles to create tendon TECs by mechanically stimulating mesenchymal stem cells (MSC) seeded in collagen constructs in tension. Juncosa-Melvin et al [171] demonstrated that two weeks of mechanical stimulation applied to rabbit MSC-collagen sponge constructs in culture improved their stiffness and also significantly improved PT repair biomechanics at 12 weeks post surgery. Juncosa-Melvin et al [167] also showed that mechanically stimulating these constructs in culture not only improved repair linear stiffness and modulus, but also increased Col1 and Col3 gene expression levels. Altman et al [173] also showed that mechanically stimulating bovine stem cell-collagen gel constructs for 21 days in culture increased Col1 and

Col3 gene expression levels. These studies suggest that applying a tensile stimulus to our murine MSC-seeded collagen constructs might also increase Col1 gene expression levels.

Several modes of mechanical stimulation, including compression, shear, and hydrostatic pressure has been shown to modulate chondrocyte matrix synthesis, but the effects of tensile strains on chondrocytes have not been widely explored. Oscillatory tension (0-10%)is known to decrease Type II collagen synthesis in chondrocyte-seeded constructs [207]. What is not known is how tension applied to murine chondrocytes affects Col2 and Col1 gene expression levels.

Given these findings, we then sought to determine if tensile stimulation would increase Col1 gene expression levels in double transgenic murine MSCs seeded in collagen constructs and decrease Col2 gene expression levels in double transgenic chondrocytes seeded in agarose. This motivated several research questions. *Does tensile stimulation of murine MSCs seeded in collagen sponges increase GFP-T expression? Does tensile stimulation also increase Col1 mRNA expression in these MSC-collagen constructs by quantitative real time RT-PCR, a conventional measure for assessing gene expression? Does such mechanical stimulation increase linear stiffness within these cell-based constructs? How does tensile stimulation of TECs containing chondrocytes seeded in collagen sponge affect both Col1 and Col2 mRNA expression?*

In order to address these questions we proposed the following specific aims and hypotheses.

Specific Aim 6: Examine how applying a tensile stimulus (0% vs. 2.4% peak strain at 1 Hz for 20 seconds followed by a 100 second rest period repeated for 5 hours/day) to a TEC containing double transgenic MSCs cultured in 3-D collagen sponge scaffolds affects GFP-T expression, Col1 mRNA expression and linear stiffness after 0, 7, and 14 days of stimulation.

Compared to non-stimulated controls at equivalent time points in culture, tensile stimulation of MSC-collagen TECs

Hypothesis 25: Increases GFP-T fluorescence.

Hypothesis 26: Increases Col1 mRNA.

And

<u>Hypothesis 27</u>: Increases linear stiffness.

We also propose that

<u>Hypothesis 28</u>: ECFP fluorescence as quantified by relative fluorescence units (RFUs) and Col2 mRNA will be linearly correlated.

Hypothesis 29: Tensile stimulation will not induce ECFP fluorescence.

Specific Aim 7: Examine how a tensile stimulus (2.4% peak strain at 1 Hz for 20 seconds followed by a 100 second rest period repeated for 5 hours/day) applied to a TEC containing double transgenic chondrocytes cultured in collagen sponges affect ECFP and GFP-T expression after 0, 7, 14, 21 and 28 days of stimulation.

Compared to non-stimulated controls at equivalent time points in culture, dynamic compressive stimulation of chondrocyte-collagen TECs

<u>Hypothesis 30</u>: Decreases ECFP T fluorescence.

<u>Hypothesis 31</u>: Increases GFP-fluorescence.

<u>Hypothesis 32</u>: Decreases linear stiffness.

These proposed studies are intended to improve our understanding of the in-vitro behavior of tissue engineered constructs. This knowledge is critical to our ability to control/modulate differentiation of MSCs, chondrocytes and fibroblasts in-vitro into desirable cell phenotypes. This level of control should help us achieve our goals of preconditioning TEC's to rapidly create functional tissue engineered [FTE] constructs for complex structures like tendons that insert into bone.

Chapter 2

Spatial And Temporal Relationships Of Collagen I And II Gene Expression And Repair Biomechanics In Healing Patellar Tendons Of Double Transgenic Mice

The objectives of this study is to examine the spatial and temporal changes in Col1 and Col2 gene expression and repair biomechanics during the healing of murine patellar tendons. 10 week mice carrying fluorescent transgenes for Col1 and Col2 gene were used. A 0.75mm full thickness punch defect was created in the mid-substance of the patellar tendon of 27 mice. The contralateral limb served as sham controls. Another 27 animals served as non-operative controls. Col1 and Col2 gene expression were examined at 1, 2 and 4 weeks post surgery. The patellar tendons were tested biomechanically at 4 and 6 weeks post surgery. Our results indicate that swelling and disorganization decreased with healing time. Col1 gene expression after injury increased by 2 weeks and then decreased in intensity by 4 weeks. No Col2 expression was seen in the mid-substance of healing tendons. Force-elongation and stress-strain curves demonstrate that healing tendons. In the future, this repair model and models like it should allow researchers to develop a roadmap for tissue engineering and cell therapy that more closely simulates normal development, potentially leading to regeneration rather than repair of tendon injuries.

2.1. Introduction

Injuries to tendon account for nearly half of the 33 million musculoskeletal injuries seen by physicians in the United States alone each year [4]. Epidemiologists estimate at least 100,000 Achilles tendon (AT) injuries [5] and 75,000 to 125,000 anterior cruciate ligament (ACL) injuries [208] are diagnosed and treated annually in the US alone. Patellar tendon injuries constitute a significant problem in a wide variety of sports [209-212]. While rupture of patellar

tendons is relatively infrequent, the incidence of such ruptures increases in older, active patients [213-218]. The cost of such musculoskeletal injuries account for over \$30 billion of health care dollars spent each year [1]. Hence the impact associated with these injuries is substantial.

Successful orthopaedic surgical procedures should result in the return of normal tissue properties. Unfortunately, surgical repairs often exhibit inferior biomechanical properties compared to the native structures [138, 219-238]. Though knee cruciate ligament reconstructions using tendon grafts are 90% successful [219], studies suggest that these grafts achieve only a small fraction of their original strength and remain susceptible to re-injury [138, 220-222]. Moreover, the defect created in the patellar tendon as a result of the graft harvest fills slowly and can maintain a gap for up to 27 months [223]. This healing defect in the tendon exhibits inferior maximum stress at 3 months post surgery (36% of normal) which appears to decline by 6 months to 14% [224]. Animal studies have demonstrated that biomechanical and biochemical properties of the harvest site may never return to normal. even after prolonged healing time [111, 224, 239-240]. Healing tendons in the sheep recover only about 80% of their original strength in about a year [225, 227, 241]. This tendon repair site remains vulnerable to re-injury [228] and is still abnormal by ultrasound even 10 years after surgery [229]. Over 50,000 rotator cuff repairs are performed each year [230-231], but failure rates are high [242], reaching 40% due to muscle contraction, decreased range of joint motion, neurovascular damage, or postoperative alteration of normal shoulder mechanics [232-233]. Even fetal tissues which have show scarless repair after tendon healing after injury, exhibit inferior biomechanics [243]. Results from all these studies clearly indicate that more effective repair strategies are needed.

Tissue engineering [244] a cell-based therapy is one such strategy. Tissue engineering approaches have recently been used to improve repair of tendon defects [171, 245-248]. However, despite these developments in tissue engineering, we know less about the molecular biologic response in healing tendons. This is especially true for Collagen I and II gene expression, two critically important structural genes upregulated in musculoskeletal tissues like

tendon and its insertion into bone, respectively. As will be shown, we have bred double transgenic mice with fluorescent promoters for Collagen I and II gene expressions in order to track cellular response during the early phase of tendon healing.

In this study, we compared spatial and temporal patterns of Col1 and Col2 gene expression in normal patellar tendon vs. patellar tendons subjected to full thickness, mid-substance punch defect injuries. We hypothesized that relative to normal tendon expression patterns, mid-substance injury would: 1. significantly increase collagen I gene expression in the wound site by one week after injury. 2. Then decrease collagen I expression by 4 weeks post injury. 3. Produce no detectable collagen 2 gene expression in the mid-substance healing wound site. 4. Result in inferior repair stiffness at both 4 weeks and 6 weeks post injury compared to contralateral sham tendon stiffness and external normal tendon stiffness. 5. Higher nuclear aspect ratio at 1 and 2 weeks post injury compared to contralateral sham and agematched controls.

2.2. Materials And Methods

Experimental Design

This study was approved by the University of Cincinnati IACUC. Fifty four (54) double transgenic (DT), 10-week old male mice were assigned to either a surgery group (n=27) or a non-operative age-matched control group (n=27) [249]. In each surgery animal, we created a full-thickness, half-width (0.75 mm) punch defect injury in the left patellar tendon mid-substance. These defects were wide enough to create reproducible injuries allowing our healing results to be related to those from a previous study [249]. A sham punch defect injury was performed in the contralateral right PT mid-substance. Nine (9) of the 30 surgery animals were assigned for Col1 and Col2 gene expression at 1, 2 and 4 weeks post surgery (n=3 for each time point) The remaining 18 animals were assigned for biomechanics at 4 and 6 weeks post surgery (n=9 for each time point). Of the 27 age-matched controls, three (3) animals each were assigned to

examine Col1 and Col2 expression at 11, 12 and 14 weeks, and nine (9) animals each were assigned to measure biomechanics at 14 and 16 weeks of age.

Surgical and Post-Operative Activity

Each animal was initially anaesthetized with 5% isoflourane and maintained under a 3% flow during surgery. Both hindlimbs were aseptically prepared prior to the surgical procedure. Using loupes at 2.5X magnification, an anterior skin incision was made over each patellar tendon and the edges of the patellar tendon were identified. Medial and lateral borders of the tendon were then longitudinally incised. Jeweler's forceps were initially used to create a pocket beneath the patellar tendon. A plastic-backed, modified #15 knife blade was then slipped beneath the full substance of the tendon (Fig. 2.1.A) [249]. Using a 0.75mm punch (Shoney Scientific, Waukesha, WI) a full-thickness defect was created in the mid-substance of the left tendon, using the knife blade as a support, so as not to injure tissues beneath the tendon (Figs. 2.1. B&C) [249]. The modified knife blade was also slipped behind the sham patellar tendon but no punch defect was created (Fig. 2.1.A). Both the injury and sham sides were closed using a 5-0 nylon suture. The subjects were monitored as they recovered from anesthesia and then permitted ad-lib activity in a cage.

Sacrifice

The mice were euthanized using CO_2 following protocols established by IUCAC. Hindlimbs were harvested and then prepared for either histology or biomechanics.

Histology

Surgery, sham and non-operative tissues were dissected to isolate the tendon midsubstance and insertions into bone (BTB). Tissues were then fixed in 4% paraformaldehyde overnight. The bone ends then fixed in 0.5M EDTA/PBS decalcifying solution for 7 days. The specimens were then soaked for 15 minutes in OCT and transferred to 30% sucrose where they were soaked for 1 additional day. All fixation steps were performed at 4^oC.



Fig 2.1. A punch defect injury was created in the mouse patellar tendon. A) A modified blade was first inserted beneath the tendon to provide support; B) a mid-substance injury was then made using a 0.75mm biopsy punch; C) revealing the mid-substance tendon injury. Only step A was performed in the contralateral sham tendon.

Specimens were then immersed in disposable base molds filled with embedding medium frozen on dry ice and stored at -80^oC. The bone-tendon-bone units were then sectioned in the sagittal plane (25 µm) and examined under an inverted fluorescent microscope (Axiovert 25, Carl Zeiss Inc., Göttingen, Germany). Filters for GFP-T (XF104-2, Omega) and ECFP (XF114-2, Omega) were selected to identify regions of Collagen I and Collagen II gene expression, respectively. Sections were also visualized using a rhodamine filter set (11002VZ, Chroma) to rule out auto-fluorescence. Sections were stained with H&E and nuclear aspect ratio was quantified using stereological sampling technology as described previously [250-251].

Biomechanical Analysis

Each injured, sham and unoperated control patellar tendon assigned for biomechanics was dissected and cleaned, leaving only the patella, patellar tendon, and tibia as one unit. The struts on either side of the patellar tendon were removed in order to leave a repair width of approximately 0.75mm. Tendon width and thickness were quantified from digital images in order to compute the cross-sectional areas of the normal and repair tissues.

The tibia was then embedded in polymethylmethacrylate in a custom-designed gripping fixture and secured in place with a metal pin (Fig. 2.2, [249]). The patella was then secured in a custom-designed, cone-shaped gripping fixture (Fig. 2.2, [249]). Each specimen was immersed in a Plexiglas chamber containing PBS (37^oC, 7.4).

The grips were then secured to a load cell above and an actuator below in a materials testing system (100R; Testresources, Shakopee, Minnesota). After ensuring vertical alignment, each specimen was preloaded to 0.02 N and then preconditioned for 50 cycles using a sinusoidal waveform (1% peak-to-peak strain at 1Hz). Each test specimen was then elongated to failure at a constant strain rate of 0.1%/s [249]. Force and grip-to-grip elongation were continuously recorded during the failure test. Linear stiffness was computed from the linear region of the force–elongation curve. Modulus was computed from the linear region of the



Fig 2.2. Tibia embedded in PMMA and held in place by a pin and cone shaped fixture for securing the patella. Our thanks to Dr. Lou Soslowsky for his laboratory's assistance with designing these grips.

stress-strain curve. Maximum force and elongation to maximum force were also recorded as well as the strain energy to maximum force (the area beneath the force-elongation curve to maximum force). Maximum stress, strain to maximum stress, and strain energy density to maximum stress was also calculated.

Statistical Analysis

Eight biomechanical parameters (linear stiffness, maximum force, elongation to maximum force, strain energy, modulus, maximum stress, strain to maximum stress, and strain energy density), nuclear aspect ratio and GFP-T average intensity were compared among treatment groups at each time interval and across time intervals using a three-way ANOVA. Significance was set at α = 0.05.

2.3. Results

Gene Expression in Unoperated Patellar Tendon and Knee

Normal fluorescence patterns in the PT and knee were similar for animals of the same age. A typical example of fluorescence pattern for 12 week group is shown (Fig. 2.3C). These patterns were also similar across the three age groups (11, 12 and 14 weeks) and there were no difference in intensity measured quantitatively (Table 2.1). Mild GFP-T fluorescence was evident along the entire PT soft tissue substance for all three age groups (Figs. 2.4A-C). Areas of GFP-T fluorescence were relatively uniform in intensity in the tissue but not observed at all tendon locations in the section. GFP-T fluorescence was slightly more intense in the cortical bone regions of the femur, tibia and patella. This fluorescence was even more intense in the trabecular regions in all bones, especially in the 14 week-old animals. ECFP fluorescence was restricted to the femoral, tibial and patellar cartilage surfaces as well as to the proximal femoral and distal tibial growth plates (e.g. Fig. 2.4B for 12 week-old mouse). No ECFP fluorescence was present elsewhere in the cortical or trabecular bone of the patella, tibia and femur, however.

Bilateral Expression Patterns 1 Week after Injury (11 week-old mouse)

Fluorescence patterns in the PT and knee were similar for different 11 week-old animals. One week after injury, Col1 gene expression appeared to be lower than normal expression in the punch defect wound site as well as along the entire tendon mid-substance (Fig. 2.5A).

However, creating the punch defect resulted in swelling of the entire tendon and not just in the defect region at 1 week (Fig. 2.5A). This swelling was greater than that observed in either the contralateral sham (Fig. 2.6A) or its age-matched control (Fig. 2.4A). The healing tissue also appeared quite disorganized (Fig. 2.5A). Although ECFP expression was not observed within the tendon body, faint ECFP expression was seen in the insertional regions (Fig. 2.5A). Compared to injured tissues, sham controls at one week post surgery showed mildly increased



Fig 2.3. GFP-T fluorescence patterns are similar in different animals in A) 2 week post surgery B) corresponding shams C) 12 week age-matched controls

Col1 expression in the insertion sites and on the posterior surface (Fig. 2.6A). However, these shams showed no swelling and a more parallel fiber orientation compared to the injured tissue (Fig. 2.6A). Average GFP-T intensity was significantly lower in injured tendons than sham and age matched controls (Table 2.1, p<0.05).



Fig 2.4. Typical GFP-T and ECFP expression patterns in: A) 11, B) 12, and C) 14 week-old, age-matched controls. GFP-T expression in evident in the mid-substance of the tendon and ECFP expression is evident in the articular cartilage of the condyle and in the growth plate of the tibia at all ages. F-Femur, T-Tibia, PT- Patellar Tendon.



Fig 2.5. Typical GFP-T and ECFP expression patterns in healing tendons at A) 1, B) 2, and C) 4 weeks of natural healing (corresponding to 11, 12 and 14 week-old animals). Each arrow indicates the region where the punch defect was originally created. GFP-T expression is highest at 12 weeks. Edema of the tendon is especially evident at both 1 and 2 weeks of natural healing. F-Femur, T-Tibia, PT- Patellar Tendon.



Fig. 2.6. Typical GFP-T and ECFP expression in contralateral sham tendons at A) 1, B) 2, and C) 4 weeks after surgery. Note the local increases in GFP-T expression levels on the posterior surfaces of the tendons. However, no swelling is obvious as a result of the procedure. F-Femur, T-Tibia, PT- Patellar Tendon.

Bilateral Expression Patterns 2 Weeks Post Injury (12 week-old mouse)

Fluorescence patterns in the PT and knee were similar for different 12 week old animals (Figs. 2.3A&B). Compared to results at 1 week, Col1 gene expression in the 2-week injured tendon was typically elevated all along the tendon mid-substance (e.g. Fig. 2.5B). These increases were particularly notable just proximal and distal to the cut edges of the punch defect site (Fig. 2.5B). The injured tendon showed a much more well-aligned fiber orientation with much less swelling compared to one week (Fig. 2.5B). Sham tendons (Fig. 2.6B) showed similar fiber orientation to age-matched controls (Fig. 2.4B) but with slightly elevated Col1 gene expression (Fig. 2.6B). Faint ECFP expression was noted in the insertional region of the injured and sham tendons but no ECFP expression was observed in either tendon mid-substance (Figs. 2.5B&2.6B). Average GFP-T intensity was significantly higher in injured tendons than sham and age matched controls (Table 2.1, p<0.05).

Bilateral Expression Patterns 4 Weeks post Injury (14 week-old mouse)

Once again fluorescence patterns in PT and knee were similar. By 4 weeks post injury, typical GFP-T expression had diminished all along the tendon (Fig. 2.5C), although it was still elevated compared to 1-week injured tendons (Fig. 2.4A). Collagen fibers were also more aligned compared to earlier time intervals. The injured tendon showed only modest thickening and/or swelling, and these changes were more restricted to the vicinity of the wound site (Fig. 2.5C). Sham PTs were relatively normal in appearance, but showed slight thickening (Fig. 2.6C). ECFP expression was only mildly elevated in the insertion regions of both the injured and sham tendons. Average GFP-T intensity was not different between injured tendons than sham and age matched controls (Table 2.1, p>0.05).

Nuclear aspect ratio

Nuclear aspect ratio in age-matched controls at 11, 12 and 14 weeks were similar (Table 2.2, p> 0.05). Similarly the nuclear aspect ratio in shams at 11, 12 and 14 weeks were similar (Table 2.2, p>0.05). No difference was seen between age-matched and shams at

corresponding time points (Table 2.2, p> 0.05). 1 week injured tendons had significantly higher aspect ratio than both 2 week and 4 week injured tendons and corresponding sham and age matched controls Table 2.2, p<0.05). By 4 weeks the injured tendons had ratios comparable to that of controls.

Healing Biomechanics 4 and 6 Weeks after Injury

Force-elongation and stress-strain curves for injured, contralateral sham, and external age-matched control PTs are shown in the 14- and 16-week old animals at 4 (Figs. 2.6A & 2.7A) and 6 weeks post injury (Figs. 2.6B&2.7B), respectively. Note the similar curve shapes and amplitudes to failure between the sham and control tendons for each age group. Also note the

	1 week	2 weeks	4 weeks
Injury	22.3±4.6^	184.1±5.2*^	41.6±10.2**
Sham	45.63±4.6	43.6±7.6	48.7±9.4
Age-matched	48.8±2.3	50.4±7.2	45.3±9.8

Table 2.1. Average GFP-T intensity was highest at 2 weeks post injury. * Significant difference in GFP-T intensity in 2 week injured PT from both 1 and 4 weeks. ** Significant difference in GFP-T intensity compared to both 1 week and 2 weeks ^ Significant difference in GFP-T intensity between injury and sham at corresponding time points.

	1 week	2 weeks	4 weeks
Injury	0.567±0.3*^	0.41±0.3**	0.36.±0.25
Sham	0.32±0.2	0.27±0.3	0.31±0.19
Age-matched	0.29±0.12	0.31±0.22	0.29±0.16

Table 2.2. Average nuclear aspect ratio was highest at 1 weeks post injury. * Significantdifference in aspect ratio in 1 week injured PT from both 2 and 4 weeks. ** Significant differencein aspect ratio in 2 week injured PT from 4 weeks. ^ Significant difference in aspect ratiobetween injury and sham at corresponding time points.

similar appearance of these sham and control curves for the 14 vs. 16 week animals. No significant differences in structural and material properties were found among the sham and control tendons at the two time intervals (Table 2.3; p>0.05). The force-elongation and stress-strain curves for the injured tendons from each of the two time intervals were also lower than the corresponding age-matched controls and contralateral shams. The structural parameters measured from these curves revealed significantly lower stiffness, maximum force, strain

energy, and elongations to maximum force for the injured vs. age-matched shams and external controls (Table 2.3, p<0.05). The material properties measured from these curves revealed significantly lower modulus, maximum stress, strain energy density and maximum stress for the injured vs age-matched shams and external controls (Table 2.3, p<0.05). Time after injury (4 vs. 6 weeks) did not significantly affect any of the structural properties (stiffness, maximum force, strain energy to maximum force, elongation to maximum force) or computed material properties (modulus, maximum stress, strain energy density to maximum stress, or strain to maximum stress) (p>0.05). Repair stiffness and maximum force were 54.5% and 42% of sham control values at 4 weeks post injury (p<0.05) and 57.6% and 52% of sham values at 6 weeks post injury, respectively (p<0.05; Table 2.3). Repair modulus and maximum stress were 46.1% and 34% of sham control values at 4 weeks post injury (p<0.05; Table 2.3).

2.4. Discussion

The fields of musculoskeletal tissue engineering and developmental biology can benefit from murine models in which cells fluoresce either GFP-T or ECFP in response to changes in Col1 and Col2 gene expression, respectively. While others are using immunohistochemistry and in situ hybridization to track changes in gene expression, this method offers an advantage in that local intracellular changes in these patterns can be observed during healing as well as during normal growth and development. In fact, our group has already observed these patterns in the DT mouse during embryonic to early post-natal development (Chapter 3).

With this information, we can now contrast how these developmental patterns compare with current methods for healing soft tissues and their insertions into bone. Substantial differences between healing and development can also motivate promising new and novel therapies where, e.g. cell-based approaches might better mimic sequences of developmental gene expression patterns.



Fig. 2.7. Compared to age-matched controls and contralateral sham injuries, punch defect injuries to the central patellar tendon produced lower force-elongation curves at A) 4 weeks and B) 6 weeks post injury.





Such models are particularly important given the frequency of tendon injuries, the inconsistent results observed following classical repair of these injuries, and the lack of biological and biomechanical knowledge about the local cellular milieu in the healing wound site over time. Hence knowing both spatial and temporal expression of important structural genes

like Collagen I and Collagen II after tendon injury and healing will be critical to designing more effective engineered constructs that accelerate and enhance the repair process. Our model also takes advantage of key transgenic technology for tracking both Col1 and Col2 expression in response to controlled mid-substance tendon injuries.

Variable	Age-Matched		Sham		Repair	
	14 weeks	16 weeks	14 weeks	16 weeks	14 weeks	16 weeks
Structural Properties						
Max Force(N)	4.03±0.53	3.53±0.62	4.13±0.35	3.89±0.43	1.74±0.43*	2.02±0.42*
Stiffness(N/mm)	8.79±1.34	9.3±1.1	7.94±0.96	9.58±0.11	4.33±1.1*	5.52±0.86*
Strain Energy (N/mm)	1.67±0.37	1.47±0.33	1.78±0.26	1.50±0.23	0.47±0.16*	0.49±0.16*
Max Elongation (mm)	0.77±0.13	0.51±0.11	0.82±0.08	0.59±0.07	0.54±0.05*	0.41±0.05*
Material Properties						
Max Stress(MPa)	36.7±4.9	32.1±5.6	37.6±3.2	32.46±3.6	12.8±3.3*	16.8±3.52*
Modulus (MPa)	239.8±36.7	234.7±27.7	216.6±26.4	239.6±17.8	99.9±23.4*	93.9±22.6*
Strain Energy Density (N/mm) per mm ³	5.11±1.1	4.57±1.0	5.40±0.78	4.67±0.69	1.21±0.41*	1.49±0.49*
Max Strain (%)	25.7±4.4	19.7±4.4	27.5±2.8	19.6±2.6	25.1±1.9*	16.2±2.3*

Table 2.3. Structural and material properties (mean±SEM) for the patellar tendon healingtissues compared to values for contralateral sham and age-matched controls. * Statisticallydifferent from shams (p<0.05).</td>

GFP-T expression in the tendon mid-substance of the normal 11, 12 and 14 week-old DT mice indicates upregulation of Col1 gene expression by tenocytes along the entire tissue structure. The fact that these patterns are similar for both the external controls and shams across time suggests that gene expression has stabilized over this range of ages. The fact that some regions in the tendon did not show GFP-T expression may mean that cells were either not fluorescing at the time point of evaluation (note that our GFP-T reporter has a 24-hour half life) or just beginning to show expression which could not be detected.

Observed GFP-T fluorescence intensity is likely more an indication of which cells are upregulating Col1 expression and not necessarily the level of expression by any one cell. In situ hybridization studies are still needed to determine the actual number of cells showing increased expression and their degree of upregulation. The minimal detectable ECFP fluorescence observed in the insertion sites of the normal PT likely indicate that cells that maintain this fibrocartilage zone are not upregulating this gene. This is in contrast to what is known to occur by in-situ hybridization during growth and development [252].

Our current study documents the temporal changes in Col1 and Col2 gene expression levels following punch defect and sham injuries. We have shown that a baseline level of collagen I gene expression (Col1) is present one week after injury, suggesting that the response to this injury is not as immediate as one might expect. The increases in swelling, disorganization and Col1 gene expression by two weeks are dramatic and surprising, however. The fact that these changes in structure and Col1 expression are not localized to the injury site is also surprising and emphasizes how an entire tissue is affected by a local punch injury, possibly as a result of the disruption of vascular supply and biomechanical tension to the tendon. The subsequent reduction in swelling and gene expression by four weeks post injury suggests that the tissue is already undergoing an early repair process.

While early healing is associated with inferior biomechanical properties, we did observe a partial return in biomechanical stiffness and maximum force compared to external controls and sham values. This improvement in structural biomechanical properties between 4 and 6 weeks is quite encouraging. Unfortunately, we cannot provide an underlying reason for this improvement since our experimental design did not include an assessment of either gene expression or tissue structure at 6 weeks post injury. However, based on these changes between 2 and 4 weeks combined with our 6-week biomechanical results, we can speculate that gene expression levels likely undergo further decreases toward control levels with corresponding reductions in tissue thickness and more parallel alignment of collagen fibers.

Despite these patterns of change, tendon properties are still not normal even after 6 weeks of healing.

Our study supports findings that type I collagen protein is essential in the tendon repair process as are type III collagen and many glycosaminoglycans and growth factors (e.g. FGF and TGF-β) [253]. Once collagen fibers are injured, repair mechanisms occur where macrophages and inflammatory cells degrade collagen after which collagen I fibrils and fibers reconstitute the defect site [67, 81, 96, 110, 112, 180, 182, 254]. The temporal pattern of collagen repair and remodeling has been also been documented[254]. This study concludes that collagen protein is synthesized and already beginning to assemble by four weeks post injury [254]. To meet this timeline, recent studies by Berglund et al [255] conclude that Col1 mRNA is expressed in the healing tendon after only 12 to 24 days of healing. Galatz et al [253] found even earlier changes during tendon repair with increases in Col1 mRNA at the insertion sites by 10 days post injury. Lin et.al found by immunohistochemistry that by 2 weeks post injury, Col1 protein expression is greatest in wild type mice but dramatically reduced in IL6 or IL4 deficient mice [249]. This result suggests that IL6 and/or IL4 are necessary to upregulate Col1 gene expression during healing in tendons. Our results showing increased GFP-T expression (reflective of Col1 gene expression) by 14 days post defect injury also agree with findings by Loiselle et.al who found that Col1 mRNA peaked at 14 days during the healing of murine intrasynovial tendon adhesions [256].

While we did not specifically examine healing biomechanics at longer periods, it is not surprising that even after 63 days of healing, murine intrasynovial tendon adhesions still produce healing tissue with inferior maximum strength [256]. In fact, our study also supports the healing of rabbit PT window defects at 4 weeks, where inferior maximum force, maximum stress, stiffness, modulus, strain energy and strain energy density were seen in healing rabbit PT window defects [257]. Assembling the results of all of these studies with our own emphasizes the fact that healing produces inferior tissues. In fact, these inferior healing

biomechanical values combined with our structural and genetic findings strongly suggest the value of novel techniques like cell-based therapies to accelerate and improve this repair process.

Our study has some limitations. 1. We did not quantify Col1 and Col2 mRNA levels by correlating qRT-PCR to our fluorescence levels in our histological images. We have previously shown strong positive correlations between fluorescence and PCR in tissue engineered constructs containing murine cells that fluorescence Col1 and 2 genes [162-163]. 2. We did not study the expression of markers important for tendon healing such as TGF-β, BMP, etc. Future studies are now underway now to study these upstream markers. 3. The biomechanical results for the healing regions may have included some adjacent strut tissue. The inclusion of adjacent tissue was unavoidable due to the extremely small size of the repair area. Thus, our biomechanical results may have overestimated the true properties of the repair tissue alone. Such a small size increases the difficulty of inducing repeatable injuries in the murine model. Future studies, particularly those involving tissue engineering, may require using larger animals if genetic markers can be incorporated. 4. Sectioning effects could have disrupted our histological sections and could have influenced the intensity of the GFP-T and ECFP fluorescence.

In this study, we were able to document temporal and spatial patterns of collagen I and collagen II gene expression during the healing of a central third mid-substance patellar tendon defect in a novel double transgenic murine model. To our knowledge, this is the first time that local expression of both genes has been documented in a tendon at the cellular level. We further document the modest genetic and biomechanical healing that occurs after punch defect injury. In the future, this repair model and models like it should allow researchers to develop a roadmap for tissue engineering and cell therapy that more closely simulates normal development, potentially leading to regeneration rather than repair of tendon injuries.

Chapter 3

Col1 and Col2 Gene Expression During the Development of Murine Knee Joints

Recently, tissue engineers have turned to developmental biology to try and improve the repair of tendons and ligaments. The premise for this new approach is that describing normal growth and development of tendons and ligaments might help in designing tissue engineering constructs and in facilitating later repair. In particular, understanding temporal changes in extracellular matrix gene expression and growth factor expression might allow tissue engineers to develop more effective treatments to augment adult tendon and ligament healing. Unfortunately, investigators have identified very few relevant markers and growth factors expressed during normal tendon and ligament development. What are needed are near realtime markers that can monitor both spatial and temporal changes in gene and/or protein expression. To that end, we have created double transgenic mice with two distinct fluorescent markers, Green Fluorescent Protein-Topaz (GFP-T) and Enhanced Cyan Fluorescent Protein (ECFP), to track changes in Col1 and Col2 gene expression, respectively. We have monitored these fluorescent changes in knee tendons and ligaments from embryonic (E17.5) through early post-natal (P1, P3 and P14) development. We found that these structures form early, that Col1 and Col2 are expressed in utero and that both become localized, organized and enhanced shortly after birth. Our GFP-T and ECFP fluorescence expression indicates that Col2 expression precedes Col1 expression and rapid changes occur in spatial expression in the tissue midsubstance and insertion sites. These developmental changes in Col1 and Col2 expression also induce temporal and spatial changes in the expression of two related growth factors, TGF-B1 and BMP-2, respectively and their upstream markers, PSmad1,5,8 and pSmad2,3. Replicating spatial and temporal expression patterns of these important markers could represent one important way to speed and improve tissue engineered repairs after tendon injury.

3.1. Introduction

Soft tissue (tendon, ligaments etc) injuries represent almost 50% of all musculoskeletal injuries that occur in the US each year [5, 258]. These include frequent injuries to the patellar tendon (42,000), Achilles tendon (100,000) and rotator cuff (51,000) as well as ruptures to the anterior cruciate ligament (125,000) [6-9]. Inadequate healing of these injuries places patients at risk to dysfunction and disability.

Surgeons have traditionally tendon and ligament disorders by reducing inflammation, restoring flexibility, and performing surgical repair [6, 259-260]. Many repairs/reconstructions have been attempted (e.g. sutures, resorbable biomaterials, autografts, and allografts) with varying success [9, 105-107, 110, 112, 261-264]. Such problems with conventional techniques have led to new strategies such as tissue engineering and functional tissue engineering.

Our group helped to create a new branch of tissue engineering called Functional Tissue Engineering (FTE) [265]. Those working in FTE seek to determine forces and deformations during activities of daily living (ADL) in order to establish design goals for increasing extracellular matrix synthesis, improving tendon organization, shortening remodeling time, and enhancing biomechanical stiffness. These investigators also use bioreactors to mechanically stimulate tissue engineered constructs or TECs in-vitro with aspects of in-vivo tissue force or deformation [167, 171, 265]. Based on these results, FTE principles and strategies now hold much promise for advancing tendon tissue engineering toward the ideal goal of regenerating normal tissue properties in the functional range of loading. For example, when our group has used FTE principles to repair central-third rabbit PT defects using MSC-collagen constructs, we have been able to generate repairs that match normal tendon stiffness up to 150% of peak invivo forces after 12 weeks of healing [171]. Despite these successes in restoring repair stiffness and strength, these 12-weeks repairs still did not regenerate normal zonal insertions into bone. Furthermore, our group has still not identified the mechanical and chemical stimulation patterns in culture that optimize cell-based tendon repair. Thus, to improve both the mechanical function

and the structure of these repairs, we need new strategies that more rapidly and fully recreate all aspects of normal tissue biology, structure and function.

FTE could benefit from a more thorough understanding of how tendons normally develop and grow. Currently, our knowledge of normal tendons growth and development is quite limited as well as how these events compare with natural tendon healing and repair in the adult [252]. Tendon structures, which are known to undergo early regulatory changes around birth, are likely controlled by TGF-β and BMP which in turn are under the control of the downstream pSmad signaling pathways. Such knowledge would be invaluable and could lead to repair strategies that might someday regenerate the tendon mid-substance and insertion site after injury.

Developmental cues could help tissue engineers spatially arrange and mechanically or chemically stimulate cells in the TEC to more efficiently synthesize and maintain an appropriate extracellular matrix (ECM).Unfortunately, this process can be slow, incremental and difficult to optimize. Investigators must test various culture conditions (cell and scaffold type, mechanical and chemical stimuli) to determine if appropriate ECM has been synthesized. This process can also become quite expensive and time-consuming as well. If these cell-based approaches are to eventually become clinical tools for the orthopaedic surgeon, we need a more novel, real-time approach that would permit us to rapidly determine how these environmental stimuli (e.g. specific loads and strains) affect cell phenotype at a reasonable cost.

Types I and II collagen are major structural proteins in the tendon mid-substance and insertion site, respectively. We need to understand how genes that synthesize these proteins are spatially and temporally expressed in normal, injured, and repairing tendons, as well as tendon cell-based constructs. We also need a more real-time technology to monitor expression of these genes. In this regard we have created double transgenic mice in which two critically important structural genes, Col1 and Col2, are linked to intracellular Green Fluorescent Protein–Topaz (GFP-T) and Enhanced Cyan Fluorescence Protein (ECFP), respectively. Cells in these

mice fluoresce green and/or cyan depending on which collagen genes are activated, permitting us to monitor expression patterns for all of these tendon states.

Thus our objectives in this initial study were to track changes in Col1 and Col2 gene expression, respectively, in the knee patellar tendon mid-substance and insertions from embryonic (E17.5) through early post-natal (P1, P3 and P14) development. We hypothesized that: 1. Col1 and Col2 would be expressed at similar times in utero. 2. Expression of both genes would be localized, organized and enhanced shortly after birth. 3. Post-natal development would further induce temporal and spatial changes in both of these structural genes. 4. Such spatial and temporal changes in Col1 and Col2 expression would mirror that of two growth factors, TGF- β 1 and BMP-2, respectively. 5. Development would also induce temporal and spatial changes in the expression of corresponding upstream signaling markers, pSmad1,5,8 and pSmad2,3 respectively.

3.2. Materials and Methods

Experimental Design

We assigned four animals each from different litters of double transgenic mice to one embryonic (E17.5) and three early post-natal time points (P1, P3, P14). The time points selected for studying changes in Col1 and Col2 expression were based on a prior study [252] where investigators monitored expression of extracellular matrix and growth factor genes during development of the supraspinatus tendon-to-bone insertion from E13.5 to 14 days. Knees were harvested and sectioned from each of the 4 animals at each time point. ECFP and GFP-T fluorescence were imaged in the same sections to indicate those cells that showed Col2 and Col1 expression, respectively. ECFP and GFP-T fluorescence were imaged in the same sections to indicate those cells that showed Col2 and Col1 expression, respectively. Adjacent sections were then stained with antibodies for TGF- β 1 and BMP-2. Different adjacent sections

were also stained with antibodies for pSmad1,5,8 and pSmad2,3. Different adjacent sections were also stained with H&E.

Harvest, Sectioning and Preparation of Hind Limbs

Animals were sacrificed by CO₂ asphyxiation in accordance with IACUC protocol. The entire hindlimb was removed from each hip. The skin was then removed to expose the underlying soft tissues. Care was taken while dissecting the muscle tissue to avoid damaging the patellar tendon and the knee ligaments. Each limb was fixed for 2 days in freshly prepared 4% paraformaldehyde which was dissolved in PBS and adjusted to pH 7.4 with NaOH. Fixed limbs were then placed in disposable base molds filled with frozen embedding medium devoid of any air bubbles. Molds were wrapped in aluminum foil and stored at -80°C. Hindlimbs were then cut in sagittal section (20µm thick) in a microtome. Sections were then stored at -80°C until they were ready for staining and imaging.

Immunohistochemistry and H&E staining

Non-specific binding sites were first blocked using 4% BSA in 0.1%Triton for 1 hour. Histological sections were incubated overnight at 4°C with 10mg/ml of one of the following: a) primary monoclonal anti-TGF-β1 and 2 (sc-146, Santa Cruz Biotechnologies, Santa Cruz, CA); b) anti-BMP-2 and 4 (sc-6267, Santa Cruz Biotechnologies, Santa Cruz, CA); c) anti-pSmad 1, 5, and 8 (9511, Cell Signaling, Beverly, MA); or d) anti-pSmad 2,and 3 (sc-11769-R, Santa Cruz Biotechnologies, Santa Cruz, CA). Sections were then stained with secondary antibody conjugated with CY5 (Jackson Immuno Research Laboratories, West Grove, PA) at a concentration of (1:200) for 1 hr.

For H&E sections were allowed to air dry to remove all moisture, washed in PBS and stained in immersed in hematoxylin and eosin for atleast 2 mins. The sections were finally cleared in xylene and coverslips were mounted.

Imaging the Knees

H&E sections were imaged in a light microscope (Zeiss, San Diego, CA). Fluorescence were examined in the confocal microscope(Zeiss, San Diego, CA) with filter cubes for GFP-T, ECFP and CY5 (Omega). Images were taken at 10X magnifications.

Quantification of Fluorescence

Custom Matlab code and a stereological sampling strategy[250-251] were used to quantify both the ECFP and GFP-T fluorescence in the images. One way ANOVA with Tukey's HSD was performed to determine if there were significant differences in either ECFP or GFP-T fluorescence among ages. Significance was set at p<0.05.

3.3. Results

GFP-T (Col1) and ECFP (Col2) fluorescence patterns in the patella-patellar tendon-tibia complex were similar among animals of the same age but from different litters. At all four ages, intracellular ECFP expression was robust in the subchondral bone regions and GFP-T expression was seen in the patellar mid-substance (Fig. 3.1A-D). ECFP expression was most intense in the bone ends at E17.5 and P3, decreasing both at P1 and at P14 (Fig. 3.1 & Table 3.1). GFP-T expression was quite evident in the patellar tendon mid-substance at P3 and most intense at P14 (Fig. 3.1 & Table 3.1). Cells expressing both ECFP and GFP-T fluorescence were seen in the insertions at both P3 (not shown) and P14 (Fig 3.2A-D).

	E17.5	P1	P3	P14
ECFP	94.6±16.5	46.1±5.2*	74.6±10.23	51.3±8.7*
GFP-T	12.3±4.6	15.4±7.2	38.7±9.8^	155.4±12.45^ #

Table 3.1. Average EFCP/GFP-T intensity at all four ages. * Significant difference in ECFPintensity from both E17.5 and P3. ^ Significant difference in GFP-T intensity from E17.5 and P1.# Significant difference in GFP-T intensity from P3.

Tissues showed both spatial and temporal differences in histological appearance. Between E17.5 and P3, cell number increased dramatically (Fig 3.3). Collagen fibrillogenesis also increased from P3 to P14, resulting in dramatic increases in tissue thickness (Fig 3.3). Collagen fiber alignment also increases with age as evidenced by the changing birefringence in the micrograph in Figure 3.3. The insertion region of the tendon showed some of the most



Fig. 3.1. Typical GFP-T and ECFP fluorescence expression in sagittal sections of murine knees at A) E17.5, B) P1, C) P3 and D) P14. GFP-T expression appears most intense at P14 while ECFP expression is most intense at E17.5 and P3. Magnification is 10x for all sections. F-Femur, P-Patella, PT-Patellar Tendon.

striking differences. Initially, the fibers at the distal end of the patellar tendon appeared to be adjacent to the developing tibial cartilage ("I" in E17.5, Fig 3.3). However, by P1 (day of birth), this region had incorporated into the proximal tibia and by P14, the birefringent insertional fibers extended throughout the insertion site (Fig 3.2).

Immunohistochemistry revealed differences in expression of growth factors and downstream signaling molecules across age groups. BMP-2 was evident in the patellar tendon mid-substance at E17.5 and P1 (Fig 3.4). BMP-2 was also quite pronounced in the proximal tibia by P14 (Fig 3.4). However, no BMP-2 could be found in any location at P3. Similarly, pSmad1,5,8, downstream marker of BMP signaling, was seen in the patellar tendons at all ages (Fig 3.5). TGF-β1 (Fig. 3.6) and pSmad2,3 (Fig. 3.7) (downstream of TGF-β1 signaling) were also seen in the patellar tendons at all ages.

3.4. Discussion

Our double transgenic reporter mouse model is a novel tool to study Col1/Col2 gene expression during the development of knee structures. We can visualize cells engaged in



Fig. 3.2. A) ECFP and GFP-T expression at P14, B) Magnified view of the insertions showing both ECFP and GFP-T expression , C) Only ECFP expression, D) Only GFP-T expression. Note the presence of cells expressing both ECFP and GFP-T as indicated by white arrows. F-Femur, P-Patella, PT-Patellar Tendon.

chondrogenesis, osteogenesis and tendinogenesis in near real time. In particular, our ability to analyze gene expression in near real time in culture has the potential to speed the development process by permitting the investigator to easily screen constructs for appropriate gene expression as culture conditions (cell and scaffold type and environmental stimuli) are changed. The 24hr half-life of the fluorescent proteins also permits us to monitor changes in expression over somewhat longer time periods without the fear of "missing" very rapid changes in gene expression.



Fig. 3.3. Shown are birefringent images of patellar tendon morphology at the tendon mid-substance (T) and tibial insertion (I) at four time intervals of development (E17.5, P1, P3 and P14). Note the crimped appearance and integration into the tibia at P14.

GFP-T and ECFP expression show well-defined spatial and temporal patterns during embryonic and early post-natal development. Our ECFP/GFP-T fluorescence results show that developing knee structures (including the patellar tendon and insertions) undergo rather repeatable and predictable spatial and temporal changes in Col1/Col2 gene expression. The greater intensities in ECFP (Col2) expression at E17.5 and P3 compared to P1 and P14



Fig. 3.4. BMP-2 expression in sagittal sections of murine knees at A) E17.5, B) P1, C) P3 and D) P14. BMP-2 is evident in the PT at E17.5 and P1 and prominent in the tibia at P14. T-Tibia, PT-Patellar Tendon. Red represents staining for BMP-2, Cyan represents Col2 expressing cells, Yellow represents Col1 expressing cells.



Fig. 3.5. pSmad1,5,8 expression, downstream of BMP-2 expression in sagittal sections of murine knees at A) E17.5, B) P1, C) P3 and D) P14. Unlike BMP-2, pSmad1,5,8 expression is seen in patellar tendons at all ages. Red represents staining for BMP-2, Cyan represents Col2 expressing cells, Yellow represents Col1 expressing cells.

suggest that tendon and ligament structures undergo early regulatory changes around birth which could be controlled by upstream signaling events. The appearance of GFP-T (Col1) expression between E17.5 and P1 and its further enhancement by P3 raises questions as to whether the same cells switch their mode of gene expression either by internal programming or through mechanical and/or chemical factors.

The increasing intensity and improving alignment of GFP-T expressing cells in the patellar tendon during development could offer strategies for improving the integrity of tissue engineered constructs in culture. In sagittal sections, we observed that GFP-T expressing cells increased fluorescence levels and were already well-aligned with the tendon axis by P3.
Evidence in the literature suggests changes in Col1 gene expression during early development but these changes have not been localized to the cellular level before. Regarding alignment, the fact that rows of tendon cells typically orient before collagen fibrillogenesis [14, 266] suggests that cell orientation strongly influences the organization of the ECM. Such knowledge about



Fig. 3.5. TGF-β1 expression in sagittal sections of murine knees at A) E17.5, B) P1, C) P3 and D) P14. TGF-β1 is seen in patellar tendons at all ages although it is more pronounced at E17.5 and P3. T-Tibia, PT-Patellar Tendon. Red represents staining for BMP-2, Cyan represents Col2 expressing cells, Yellow represents Col1 expressing cells.



Fig. 3.6. pSmad 2,3 expression downstream of TGF- β 1 in sagittal sections of murine knees at A) E17.5, B) P1, C) P3 and D) P14. Similar to TGF- β 1. pSmad 2,3 is seen in patellar tendons at all ages. T-Tibia, PT-Patellar Tendon. Red represents staining for BMP-2, Cyan represents Col2 expressing cells, Yellow represents Col1 expressing cells.

organization and gene expression of cells in the developing tendon could be quite useful if strategies could be devised for mimicking these temporal and spatial events by applying tensile stimuli to cells within tissue engineered scaffolds in culture and then determining their effects on repair after surgery.

While creating a tendon or ligament mid-substance is now quite conceivable, recreating a complex tendon or ligament insertion into bone will require more sophisticated approaches. Our group recently found that mechanically stimulating rabbit MSC-collagen scaffold constructs in culture resulted in well-organized and biomechanically competent tendon ECM 12 weeks after surgical implantation [171]. However, our group [171] and others [62, 267-269] have not been successful in regenerating the complex tendon or ligament zonal insertion site into bone after injury. The fact that Col2 gene appears to be expressed before Col1 gene during joint development suggests the need to investigate whether triggering cells to also express Col2 early after injury might increase our chances of recapitulating a zonal insertion. The fact that Col1 and Col2 genes are also co-expressed by cells in the insertional region during development emphasizes the value of closely monitoring these spatial and temporal changes during development and then driving expression of both structural genes in the adult healing attachment site. Given this early post-natal co-expression, animal models like our double transgenic mouse or use of in situ hybridization [252] could be quite valuable ways to monitor the developmental process so as to learn how to better repair and maybe someday regenerate soft tissue insertions into bone.

Previous studies in the literature indicate that regions of immature and even adult tendons and ligaments can convert to fibrocartilage at their sites of attachment to bones, similar to our insertional findings during embryogenesis. For example, a tendon or ligament can synthesize cartilage matrix proteins where it passes over bony surfaces [26, 270-271], presumably due to the high compressive and shear loads. Such adaptation reflects the common ancestry of connective tissue progenitors and cartilage-forming cells. Our data shows similar findings in that by P1, the distal end of the patellar tendon incorporates into the proximal tibia through differentiation from the tendon side and/or growth of cartilage up to the tendon boundary. These results suggest that to recreate insertion sites during tendon repair, investigators should monitor and attempt to enhance co-expression of Col1 and Col2 genes which is likely lacking in the current set of FTE tendon constructs.

These changes in GFP-T/ ECFP fluorescence around birth suggest that knee structures undergo regulatory changes which could be controlled by either biologic or physical factors. A recent study of rat supraspinatus entheses indicates that physical factors do not play a

significant role in enthesis maturation in the first 14 days postnatally, implying that biologic factors may drive early postnatal development [272]. Thus our Col1/Col2 gene expression patterns could be a result of differential expression of biological factors, especially BMP-2 and TGF-β1.

Immunohistochemistry did reveal differences in growth factor patterns across age groups. For example, BMP-2 was only evident in the PT mid-substance until P1 and in the tibia at P14. BMP signaling has been suggested to control regional differentiation in tendon. Generally BMP signaling is inhibited in the tendon mid-substance, allowing tendon fiber formation, but is activated at the insertion site, allowing fibrocartilage formation[273-274]. The appearance of BMP-2 in the tendon mid-substance is contrary to what we expected. It has been suggested that BMP-2 might act as a regulator of TGF- β 1 signaling which is important in the mid-substance [274]. The absence of BMP-2 in the tendon mid-substance beyond P1 is also significant, suggesting that BMP-2 is likely important only during early development of the insertions and that other members of the BMP family could play important roles during later development.

pSmad1,5,8 expression did not mirror that of BMP-2 expression. Since only the BMP family acts through the pSmad1,5,8 pathway, other members of the BMP family are also active during development. Future studies are necessary to determine which BMPs are expressed at these developmental time points. Once these BMPs have been identified, these should be used to stimulate TECs in culture to see if such stimulation improves not only the in-vitro mechanical properties of TECs but also the resulting in-vivo repair mechanical properties.

Unlike BMP-2, TGF- β 1 was seen in the patellar tendons at all ages. TGF- β 1 has been shown to have a central role during morphogenesis and tissue formation by influencing the differentiation of connective tissue precursors into tendon forming cells [275]. Thus this expression of TGF- β 1 can help to explain the increases in Col1 gene expression we observed. Also the appearance of TGF- β 1 at all time points suggests that this growth factor is likely

important not only during early development, but also during later development of the tendon mid-substance. This suggested role of TGF- β 1 during development taken together with results in the literature showing improvement in collagen synthesis [276-277] and improvements in healing of tendon mechanics [278] makes this a key factor to be investigated during the development of tissue engineered constructs for tendon repair.

pSmad2,3 expression mirrors that of the TGF- β 1 expression. This suggests that the expressed TGF- β 1 acts on the cells in the tendon mid-substance and further emphasizes the importance of TGF- β 1 in the formation of tendon mid-substance and possibly delivery of this growth factor locally to TECs.

Recreating tendon zonal attachment sites after injury is another important problem facing tendon tissue engineers [62, 171, 267-269]. Our growth and development results motivate new strategies to solve this problem. Our results showing: 1) appearance of Col2 before Col1 during development, 2) early co-expression of Col1 and Col2 in insertion regions, and 3) differences in expression of BMP-2 vs.TGF- β 1, suggest that to recreate insertions during tendon repair, investigators should monitor and attempt to enhance the co-expression of Col1 and Col2 in the end regions of tendon constructs. Researchers should also strive to regionalize the expression of BMP-2/TGF- β 1, if these are lacking in the current set of FTE tendon constructs.

Our study is not without limitations. We only monitored GFP-T and ECFP fluorescence as early indicators for Col1 and Col2 gene expression in sections of the murine knees. While these fluorescent markers may appear to be subjective measures, we have previously shown positive significant correlations between relative fluorescence units or RFAs and qRT-PCR for Col1 and Col2, respectively [2, 163, 279]. We did not measure actual Type I or Type II collagen protein expression from these sections. However, we have shown in culture that Type II collagen protein is present within 7 days after upregulation of Col2 gene expression [2, 163, 279]. Intracellular fluorescence is more an average intensity rather than a quantity precisely

describing local cellular expression. Thus, our results indicate more about how many cells are actually upregulating either Col1 or Col2 rather than the degree to which expression is changing in each cell. This limitation is also in part a result of the 24 hour half life we chose for each fluorescent reporter. We are currently planning in situ hybridization studies to determine individual cell expression levels. Although our double transgenic model should allow us to track changes in Col1 and Col2 expression in near real time, currently we must sacrifice animals and section knees to image the fluorescence. Once we can identify powerful in vivo optical technologies that can visualize at the cellular level, we should be able to image the entire knee with the required magnification and conduct cost- and time-efficient longitudinal studies that eliminate assignment of animals for gene expression patterns at different ages and healing times.

In summary, we have used a murine model to study the development of collagenous soft tissue knee structures. We have shown that during normal development of tendons there are temporal and spatial changes in the expression of Col2 and Col1, BMP-2 and TGF-β1, and pSmad1,5,8 and pSmad2,3. Understanding differences in structural gene expression among developing structures as well as normal and healing adult tissues should allow us to formulate treatments for augmenting and accelerating tendon and ligament healing. Replicating rapid temporal changes in ligament and tendon gene expression and growth factors may also be important when mimicking key events in culture so as to create functional tissue engineered structures. Future studies will look at the expressions of upstream markers for Col1 (e.g., scleraxis, tenomodulin and thrompospondin) as well as for Col2 (e.g., Sox 9) in hopes of recapitulating more developmental events and dramatically improving the rate of soft tissue and insertional healing after injury and disease.

Chapter 4

Development And Validation Of A Bioreactor For Delivering Compressive Mechanical Stimulation To Functional Tissue Engineered Cartilage

External mechanical stimuli play a critical role in the natural development of musculoskeletal soft tissues, suggesting that delivering mechanical signals to constructs in culture that mimic aspects of in-vivo signals might create more "functional" tissues before surgery. Unfortunately, it has been difficult to identify or build bioreactors that can apply the desired mechanical stimulation to engineered load-bearing structures like articular cartilage and meniscus. To that end, we have designed and fabricated a one-step process bioreactor combining the culture and stimulation phases in order to create more functional tissue engineered cartilage. Our culture chamber, housed within an electromagnetic testing system (Model ELF 3200, Bose Corp, Eden Prairie, MN), sustains a long-term cell culture environment that is comparable to conditions created in standard incubators. Cells cultured for up to two weeks in this one-step bioreactor show comparable viability to cells cultured in standard incubators. A custom mechanical fixture within the chamber can stimulate up to twelve constructs and record tissue reaction loads in half of these, providing real-time calculation of construct stiffness for optimizing our stimulation conditions. The testing system with the custom fixture delivers repeatable and simultaneous dynamic compressive signals. Preliminary results show that compressive stimulation of chondrocyte-agarose constructs in our reactor for 14 days significantly increases collagen II gene expression. Using this bioreactor, we hope to speed the creation of more functional tissue engineered cartilage and menisci for future pre-clinical and clinical use.

4.1. Introduction

Cell-based tissue engineering strategies [280] have gained in popularity as an alternative approach to repair and replacement of injured and diseased soft tissues and organs, especially with the problems faced with conventional repair techniques[281-289]. Initially, tissue engineers simply mixed cells and scaffolds to create tissue engineered constructs for implantation in-vivo [51, 148, 150, 246-248, 290]. Recently, however, the field of tissue engineering has evolved as researchers began to recognize the importance of restoring mechanical function in-vivo. In 2000, a new branch of tissue engineering called functional tissue engineering (FTE) was advanced to account for this need for restoring function [265, 291-293]. The central tenet of FTE is the need to measure the loads and strains acting on tissues in-vivo during normal activities and to then use these as design parameters and requirements for the tissue engineered constructs. FTE also suggests that these mechanical stimuli could be used to "precondition" constructs during the in-vitro culture phase in order to acclimate them to their future in-vivo loading regimes.

With the emergence and growth of FTE, researchers have been striving to develop bioreactors which can stimulate cell/ cell-constructs with complex in-vivo profiles. Early investigators used spinning flasks and rotating vessels to apply hydrodynamic pressure and rotational forces [294-295]. Others used flexible membranes deformed by air pressure to stretch cells in monolayer and in 3-D culture[296-298] or to compress tissue explants[299]. Still, the paucity of commercially-available bioreactors has led many investigators to develop specialized systems that are typically housed within standard cell culture incubators [300-302]. Despite these efforts, these systems often lack the ability to apply complex dynamic in-vivo stress/strains and/or loads/displacements required to effectively stimulate FTE constructs. In

the case of cartilage, it has been shown that it experiences peak stresses of 2-8MPa,[303-304] and strains of about 3% to 20% [305-306] during various activities.

To achieve these FTE demands, we contend that a bioreactor should satisfy four design principles [2]. Bioreactors should: 1) Sustain a cell-culture like environment during periods of mechanical stimulation to ensure adequate cell viability; 2) Have the ability to stimulate multiple constructs with either identical or individual waveforms to reduce development times; 3) Should deliver precise displacements to compliant TECs, including signals that mimic in-vivo activities of daily living; and 4) Should be capable of monitoring reaction loads and adjusting displacement patterns based on these loads.

The pneumatic bioreactor used in our lab for over 10 years satisfies just two of the above mentioned design principles. The pneumatic system housed in a standard incubator sustains a TEC in culture with adequate cell viability. The pneumatic system also stretches five silicone dishes with individual waveform patterns, each dish having wells to accommodate four TECs. However, our pneumatic system is also regulated by pressure differential and displacement stops to limit peak strains. These characteristics mean that the signal we deliver to the TEC is not very precise and cannot mimic complex in-vivo deformation patterns. In addition, this system lacks the instrumentation required to monitor construct force in individual TECs.

To satisfy more of our bioreactor design requirements, we identified an electromagnetic testing system (ElectroForce 3200; BOSE Corp., Eden Prairie, MN) that could not only deliver a wide variety of dynamic, controlled-displacement waveforms with micron precision, but could also import future in-vivo displacement data to drive the actuator as we move toward our ultimate goal of using physiologic signals to stimulate engineered tissue in-vitro. We also sought a multi-station system that could support multiple load cells to allow for real-time monitoring of the forces applied to our engineered tissues. However, since it was not feasible to

place this system within a standard incubator, we instead chose to create and sustain a cell culture environment within the testing system.

Therefore, our objective in this study was to develop a bioreactor within an electromagnetic testing system that 1) maintains cell culture conditions and 2) could apply precise and repeatable dynamic displacements to multiple tissue engineered constructs and provide load feedback in real time. The following work describes our validation of the culture environment and applied displacements within our system as well as preliminary data on the performance of this system for loading cells in 3-D constructs.

4.2. Materials and Methods

Environmental chamber

A chamber made of plexiglass was adapted to fit within the electromagnetic system. The chamber (ported chamber and saline accessory bath) provided by BOSE (BOSE Corp., Eden Prairie, MN) served as the environmental chamber of the bioreactor. The bottom of the chamber can be filled with a 2L water bath. To maintain the temperature and the humidity within this chamber we heated this water bath using a feedback system I (Fig. 4.1A) [2].

The CO₂ levels within the chamber are controlled at 5.0% by using an external CO₂ controller (Model 3057; ThermoForma, Marietta, OH). This controller gets feedback from a sensor placed in the environmental chamber. CO₂ from the tank reaches the environmental chamber through a series of steps. First CO₂ flows from the tank into this external controller. From there CO₂ flows into an antechamber containing a blower. This antechamber is attached to the environmental chamber (Fig. 4.1B)[2]. The blower then forces the CO₂ into the environmental chamber. A CO₂ sensor mounted within the environmental chamber constantly measures CO₂ levels. When the measured level falls below the specified limit (5.0%), the controller injects CO₂ until the specified limit is achieved.

Apparatus for applying compressive loads

A fixture consisting of two parallel anodized aluminum plates are used to apply precise compressive waveforms from a single actuator shaft of this electromagnetic dynamic testing





system (Fig. 4.1A). The top plate is attached to the actuator shaft while the bottom plate is attached to the vertically-adjustable arm mounted to the bottom of the chamber. These two plates fit within the environmental chamber and moves on bearings along stainless steel rods that maintain proper alignment of these plates (Fig. 4.1B)[2]. The bottom plate can securely house up to twelve 35 mm dia. tissue culture dishes. Six thin aluminum rods with adjustable screws and containing submersible load cells (Model 31, 25 lb capacity; Honeywell Sensotec, Columbus, OH) extend downward from the top plate and are positioned above the dish locations.

A quick-connect coupler system was designed to facilitate rapid loading and removal of the constructs (MC1002, Colder Products, St. Paul, MN). The female component of this coupler is attached to the free ends of rods that extend downward from the top plate. Male component of this coupler is attached to a delrin disk (8 mm thick x 15 mm dia) through a 4" x 4" piece of 100 Å thick Teflon FEP film (DuPont, Circleville, OH). This film is secured on top of the 35mm culture dishes via a custom delrin ring. This ring also has small threaded hole for media exchange (Fig. 4.1C)[2]. This film prevents media evaporation, reduces the risk of contamination and does not impede actuator motion. This film is also permeable to O₂ and CO₂. This coupler system allows easy rotation of specimen groups to increase the number of constructs that can be stimulated daily. The couple system also allows each dish to be removed easily from the system for observation. Once this coupler is assembled, constructs can then be subjected to compressive stimuli by activating the system actuator, which in turn can move the top plate and press the delrin disk onto the specimen within the dish.

Cell sources

Two types of murine cells were used in different studies to validate our bioreactor. For the cell viability study, mesenchymal stem cells (MSCs) of 6-8 week old mice wild type mice were used. For the compressive stimulation study, rib chondrocytes from six double transgenic newborn mice created in our laboratory were used. These cells have two colorimetric

promoters, enhanced cyan fluorescent protein (ECFP) and Green Fluorescent Protein (GFP-T), which are activated when the type II collagen gene and type I collagen genes are expressed, respectively. The creation of the double transgenic mice is described elsewhere[163].

Characterization of the bioreactor environment

A hygrometer (Model 11-661-18; Fisher Scientific) was used to measure both air humidity and temperature within the environmental chamber. CO₂ levels were measured using both a feedback sensor on the controller and independent Fyrite gas analyzer (Model 11-7029; Bacharach). A traceable digital thermometer (Model 11-661-9; Fisher Scientific) was used to measure the temperature of 5 ml of culture media within a tissue culture dish placed in one of the stations on the bottom plate. Time needed for the environmental conditions to reach steady cell-culture conditions was first monitored. The chamber door was then opened for five minutes to simulate dish exchanges, and conditions were again monitored to determine the time required to re-establish the steady state.

Measurement of axial displacements among bioreactor stations

A sonomicrometry system (Sonometrics Corp., Ontario, Canada)[307] was used to measure the relative displacements among stations located at all corners of the top plate (i.e. adjacent to the stainless steel rods). One piezoelectric peg crystal was mounted at the center of the delrin disk attached to the coupler end, and another was mounted in the center of a tissue culture dish. The disk and dish were immersed in water to allow for communication between the ultrasound crystals. The top plate was moved vertically using input sinusoidal waveforms of 0.1 and 1.0 mm amplitudes at 1 Hz for ten cycles. The chosen peak amplitudes define the range over which we expect to stimulate tissue engineered constructs. Relative crystal distances were measured and compared to the actuator displacement recorded by the internal LVDT of the electromagnetic system (Model ElectroForce 3200 system, Bose Corp.).

Determining cell viability/proliferation

The MSC isolation and expansion protocol up to P2 are described elsewhere [162]. Marrow-derived murine MSCs at P2 were plated onto 35-mm tissue culture dishes at a concentration of 100,000 cells per dish. Each dish was then covered with Teflon film and secured via the delrin ring to maintain sterility. Controls were placed in a standard incubator (Model 370; Thermo Electron Corp., Marietta, OH). Media was replaced every two days.

Cell proliferation and viability were examined after 3, 7 and 14 days of culture. At each time point, three dishes were removed from both the bioreactor and standard incubator and cell proliferation was determined using an MTT assay (Vybrant MTT Cell Profileration Assay Kit, Invitrogen, Carlsbad, CA). We performed the assay in accordance with manufacturer's protocol[162]. The absorbance of formazan was measured at 570 nm in a microplate spectrophotometer (Spectra Max M2, Molecular Devices), and absolute cell numbers were obtained by plotting the obtained absorbance values on a known standard curve.

Mechanical stimulation of chondrocytes cultured in 3-D

Double transgenic newborn mice were euthanized by CO₂ following Institutional Animal Care and Use Committee (IACUC) protocols. Chondrocytes were first harvested from the ribs of six mice using an established procedure[308]. The cells were cultured in monolayer until passage 3, suspended in media (BGJb + 10% fetal calf serum), and mixed with equal volumes of 4% (w/v) agarose (Sigma) to obtain 2% agarose with 10x10⁶ cells/ml. This mixture was cast between sterile glass plates from which 8 mm dia x 3 mm thick cylindrical constructs were created using a sterile biopsy punch. For each animal, three constructs were created for each treatment group. Cells were subjected to cyclic sinusoidal compression at 10% peak strain at 1 Hz for 1 hour followed by a 1 hour rest period for 6 hrs/day in the bioreactor r[302]. Non-stimulated constructs were cultured in the same bioreactor without stimulation. Media was changed daily. At 0 and 14 days in culture, constructs were visualized in a fluorescence

microscope (Zeiss Axiovert) with a filter set (XF114-2, Omega) for cyan fluorescence (ECFP), indicative of Col2 gene expression. Other filter sets for green and red fluorescence (XF104-2, Omega and 11002VZ, Chroma) were also used to rule out autofluorescence. Constructs were washed thoroughly in PBS and then mechanically homogenized using a glass mortar and Teflon pestle, and ECFP was quantified in Relative Fluorescence Units (RFUs) using a spectrophotometer (Spectra Max M2, Molecular Devices) with excitation/emission/cutoff wavelengths of 436 nm/486 nm/475 nm, respectively. RFU values were normalized to day 0 values for all samples. Differences in RFUs were examined using a mixed effects model (SAS proc mixed) with time in culture and compression as fixed factors and animal as a random factor with α =0.05.

4.3. Results

Characterization of the environment

Increasing the temperature of the 2L water bath in the bottom of the environmental chamber to 39°C increased the media temperature to 37°C and the chamber air temperature to

32°C. This increase in media temperature occurred within 100 min of heating the water bath (Fig. 4.2). Relative humidity (RH) within the chamber reached 75% within 80 min, which was comparable to levels measured in the standard incubator in our lab using the same hygrometer (70-76%). CO_2 levels reached a 5%



Fig 4.2. Media temperature, air temperature, and relative humidity within the bioreactor reach levels appropriate for cell culture within 100 minutes (n=3, mean \pm std dev).

steady state within 14 minutes as measured by both sensors. Opening the chamber door for five minutes minimally affected the environment (Fig. 4.3), as media temperature was only

reduced by 1°C and relative humidity was restored to approximately 70% within 10 minutes of closing the chamber door.



Fig 4.3. Media temperature and chamber relative humidity are restored to appropriate levels within 20 minutes after opening of chamber door for five minutes (n=3, mean \pm std dev).

Characterization of axial displacements

The relative crystal motions were nearly identical for both input waveforms. Differences

in peak-to-peak amplitudes were ≤ 0.014 mm between all stations for both waveform amplitudes. The crystal motions were in phase with the input motions throughout the motion path.

Cell proliferation/viability

The number of viable, proliferative cells within the bioreactor was not significantly different from that of cells cultured in



Fig 4.4. Viability of cells cultured in the bioreactor is comparable to that of cells cultured in a standard incubator over 14 days (n=3, mean \pm std dev).

the standard incubator at 3, 7, and 14 days of culture (p>0.25; Fig. 4.4). Compared to 3 day cultures, cell numbers increased by 60.8% and 63.8% after 7 days of culture in the incubator and bioreactor, respectively. Compared to 7 day cultures, cell numbers decreased slightly by 10.3% and 5.6% in both the incubator and bioreactor, respectively.

Mechanical stimulation of chondrocytes cultured in 3-D

ECFP fluorescence was observed both in monolayer and in 3-D culture at day 0 and was maintained in all constructs at 14 days. Fluorescing cells in both stimulated and non-stimulated groups showed а rounded morphology. Quantitative RFU data (Fig. 4.5) showed that ECFP expression significantly increased with both



Fig 4.5. ECFP RFUs increase with mechanical stimulation and culture time (n=18, mean ± std dev).

stimulation and culture time. Stimulated constructs had 14% higher RFU levels than their nonstimulated counterparts (p=0.002). RFUs in non-stimulated constructs significantly increased by approximately 127% after 14 days compared to day 0 controls, with an even greater 158% increase in the stimulated group (p<0.001).

4.4. Discussion

The benefits of mechanical stimulation have been both recognized and observed in a variety of tissue models [162-170]. Hence researchers have been designing bioreactors [170, 300-301, 309] to stimulate constructs using variety of stimulation profiles. We in our lab have created and validated a novel bioreactor for applying compressive stimulation to constructs. This type of bioreactor can be applied to cartilage, meniscus, disc and interface tissue engineering. In this bioreactor we have shown that we can 1) maintain and sustain cell-culture conditions. 2) apply precise displacements 3) stimulate 12 constructs at a time 4) improve Col2

gene expression in murine chondrocyte-constructs. With the instrumented load cells we can also measure the reaction forces. Thus our bioreactor satisfies all the four design criteria set forth for FTE bioreactors.

Our bioreactors have certain advantages when compared with the ones currently being used [170, 300-301, 309] .The first is the ability to calculate stiffness in real time by recording load vs displacement data during culture. The stiffness value can then be compared against normal tissues thus providing biomechanical feedback for adjusting the stimulation to better "challenge" the constructs and to accelerate their maturation in culture for more rapid delivery at The second advantage is the flexibility of our control system allows for easy surgery. manipulation of the various stimulation parameters, including waveform shape, frequency, and amplitude, and it can also receive imported in-vivo displacement profiles when these are available. Other systems may be limited in this capacity. There is also much potential for continued improvement, including the integration of systems for controlling nutrient supply and continuous media exchange and the addition of actuators and/or alteration of indenter shapes to create shear deformation/loading conditions. Ultimately, the concepts generated and employed in the construction of this bioreactor can be applied to the development of a more complex system that can reproduce most or all of the 3-D motions needed to simulate more physiologic joint motions.

Using our novel bioreactor, we plan to embark upon cartilage and fibrocartilage FTE in a similar fashion by optimizing the in-vitro stimulation profiles to create a more mechanically viable engineered tissue before surgery.

Chapter 5

3D In-vitro Effects Of Compression And Time In Culture On Aggregate Modulus And On Gene Expression And Protein Content Of Collagen Type II In Murine Chondrocytes

The objectives of this study were to determine how culture time and dynamic compression, applied to murine chondrocyte-agarose constructs; influences construct stiffness, expression of Col2 and type II collagen. Chondrocytes were harvested from the ribs of six newborn double transgenic mice carrying transgenes that use enhanced cyan fluorescent protein (ECFP) and green fluorescent protein (GFP-T) as reporters for expression from the Col2a1 and Col1a1 promoters, respectively. Sixty three constructs (8mm dia x 3mm thick) per animal were created by seeding chondrocytes (10x10⁶ per ml) in agarose gel (2% w/v). Twenty eight constructs from each animal were stimulated for 7, 14, 21 or 28 days in a custom bioreactor housed in an electromagnetic system. Twenty eight constructs exposed to identical culture conditions but without mechanical stimulation served as non-stimulated controls for 7, 14, 21 and 28 days. The remaining seven constructs served as Day 0 controls. Fluorescing cells with rounded morphology were present in all constructs at all five time points. Compared to time-matched, non-stimulated controls, 7, 14, 21 and 28 days of stimulation significantly increased Col2 expression by ECFP fluorescence and mRNA expression by quantitative RT-PCR. Col2 gene expression in both stimulated and non-stimulated constructs showed initial increases up to day 14 and then showed decreases by day 28. Compared to time-matched controls, stimulation significantly increased type II collagen content at 21 and 28 days. Compared to time-matched controls, stimulation significantly increased aggregate modulus at only 28 days. There was a significant increase in aggregate modulus in stimulated constructs between day 0 and 7, and between day 21 and day 28. This study reveals that compressive mechanical stimulation is a potent stimulator of Col2 gene expression that leads to measurable

but delayed increases in protein (type II collagen) and then biomechanical stiffness. Future studies will look at the effects of components of the mechanical signal in culture and address the question of whether such in-vitro improvements in the tissue engineered constructs actually enhance repair outcome after surgery.

5.1. Introduction

Articular cartilage injuries are quite prevalent and if left untreated, can lead to long-term osteoarthritis [310-313]. In the US alone, nearly 21% of the adult population suffers from osteoarthritis [314] resulting in over \$128 billion in direct and indirect costs[315]. Tissue engineering [244] is an appealing conceptual alternative when conventional repair techniques (e.g. arthroscopic techniques, periosteal/perichondral grafts, autograft/allograft osteochondral transplantation and prostheses) [316-322] prove unsatisfactory.

Tissue engineering approaches have evolved as investigators recognize the importance of mechanical function. Researchers mix cells with scaffolds to create tissue engineered "constructs" or TECs that can be stimulated in culture before surgical implantation (e.g. to fill tissue defect sites) [246-248]. However, these TECs often lack the inherent mechanical stiffness to tolerate large in-vivo forces such as those acting on articular cartilage. To address this problem, investigators have been applying principles of Functional Tissue Engineering (FTE) [159-160, 323] to precondition constructs using aspects of in-vivo tissue forces and deformations [157-158] to create new generations of reparative tissues [167, 170-171, 324-325]. Such preconditioning improves: 1) construct material properties by increasing synthesis of extracellular matrix proteins like collagens I and II [167, 170, 324-327]; and 2) repair biomechanics after implantation into defect sites [171, 324].

Unfortunately, tissue engineered cartilage constructs often require months of intermittent mechanical conditioning to achieve mechanical properties and protein composition even remotely approaching those of native cartilage [327]. For example, Mauck et.al [326] determined that tissue engineered cartilage constructs required a full 8 weeks of dynamic

loading in culture in order to achieve 75% and 25% of a normal tissue's Young's modulus and unconfined dynamic modulus, respectively. Even with these mechanical improvements, total collagen content was still only 12% of native tissue values [326]. While such long development cycles may be satisfactory in a research setting, these intervals are not practical for patients in a clinical setting.

To shorten the development cycle and reduce production costs, new technologies are needed to rapidly and repeatedly assess the evolving TEC in culture. During the early development period, tissue engineers should be able to non-destructively assess the TEC in culture to avoid creating inappropriate collagen types (e.g. Type I vs. Type II collagen in a TEC that could adversely affect the repair of articular cartilage). For example, if investigators could monitor how mechanical signaling influences Col2 gene expression in near real time, they could dramatically improve their ability to synthesize Type II collagen protein. Such technologies could also allow the tissue engineer to determine which mechanical signals most increase Col2 gene expression and compressive stiffness.

To address these needs, we have developed two innovative technologies. 1) We have bred specialized double transgenic (DT) mice that carry transgenes that use enhanced cyan fluorescent protein (ECFP) and green fluorescent protein (GFP-T) as reporters for Col2a1 and Col1a1 expression, respectively. ECFP and GFP-T each have a half life of 24 hours and serve as visual reporters for cells that express the genes for Types II and I collagen, respectively. ECFP linked to Col2a1 offers the potential to ultimately track in near real time how various stimuli (mechanical and/or chemical) influence Col2 gene activity in culture. 2) We have also designed and fabricated a bioreactor to deliver compressive displacement profiles with micron precision to constructs in an incubator setting over four-week intervals. This bioreactor is described elsewhere [2].

Using these two technologies, our objectives in this study were to test four hypotheses related to compressive stimulation of double transgenic chondrocyte-agarose constructs. We

hypothesized that compared to controls, compressive stimulation significantly increases: 1) ECFP fluorescence and Col2 mRNA expression; 2) Type II collagen production; and 3) TEC aggregate modulus up to 28 days in culture. We also hypothesized that 4) ECFP fluorescence and Col2 mRNA expression are positively correlated.

5.2. Materials and Methods

Experimental Design

Rib chondrocytes were obtained from six newborn DT mice (one cell line per mouse). Sixty-three cell-agarose constructs (8 mm dia. x 3 mm thick) were created per animal using one cell density (10 x 10⁶ cells/ml) [169] in a 2% w/v ratio to maintain chondrocyte phenotype [328] and to achieve a power of 85% to detect treatment-related differences if they existed. We compared the responses of twenty eight (28) stimulated (S) constructs and twenty eight (28) non-stimulated (NS) controls at 7, 14, 21 and 28 days in culture (seven (7) per time period). These results were also compared to findings from seven (7) additional NS controls per cell line at day 0. S constructs all received the same compressive stimulation profile (sinusoidal displacement pattern @ 1Hz to 10% peak strain for one hour followed by 1 hour of rest, repeated three times a day) adapted from Mauck et al [169]. The seven (7) S and seven (7) NS constructs at each time point were assigned as follows: one for ECFP fluorescence in a spectrophotometer (measured in Relative Fluorescence Units (RFU))[329], one to evaluate changes in mRNA expression for Col2 and glyceraldehyde-3-phosphate dehydrogenase (GAPDH) using real-time quantitative reverse transcriptase polymerase chain reaction (qRT-PCR) [167], one for confined compression to determine aggregate modulus [330], one for cell viability using an MTT assay [331], and three constructs for Type II collagen content using ELISA [332]

Chondrocyte isolation and expansion

Creation of the murine DT model is described elsewhere [163]. Newborn mice were euthanized by CO₂ using protocols approved by the Institutional Animal Care and Use

Committee (IACUC). Chondrocytes were harvested from the rib cages using previouslypublished procedures [333]. Briefly rib cages from each mouse were repeatedly washed in PBS and digested in 2.5 ml of Protease (Sigma P-8811; 2mg/ml) for 30 minutes in an incubator (Steri-Cult Model 3033, Forma Scientific). After incubating overnight in Collagenase B solution (Sigma; 1mg/ml), the digest was washed with media (phenol red free DMEM, 10% FBS, pen/strep), filtered (70µm) to remove remaining bone pieces, sorted by flow cytometry to exclude non-ECFP cells and plated in a 100mm tissue culture dish (Falcon) and incubated for one day at passage 0 (P0). Intracellular cyan fluorescence was evident in these chondrocytes when visualized in a fluorescence microscope (Axiovert 25, Carl Zeiss Inc., Göttingen, Germany) equipped with ECFP and GFP-T (Omega) filter sets. The monolayer chondrocyte culture was then trypsinized using 0.25%/EDTA and replated at 1x10⁶ cells per 100mm plate at P1 and cultured for 7 to 10 days. Chondrocytes were passaged until P3. No intracellular GFP-T expression was seen in cultured chondrocytes from P0 through P3.

Construct preparation

Chondrocytes at P3 were suspended at 20x10⁶ cells/ml in media (BGjB, Invitrogen-Gibco BRL/Life Technologies Inc., Gaithersburg, MD) containing 1% antibiotic/antimycotic and 10% fetal bovine serum (FBS; Atlanta Biologicals, Lawrenceville,GA). Cell suspensions were mixed in equal volumes with 4% (w/v) agarose (TYPE VII, low gelling temperature, Fisher Scientific) and PBS to produce a cell-gel solution of 10 x 10⁶ cells/ml in 2% (w/v) agarose. This solution was then cast for 20 minutes between sterile glass plates separated by 3-mm spacers. Disks (8mm diameter) were cored with a sterile dermal punch and each cultured in 5 ml of media in a multi-well plate for 2 days (henceforth denoted as "day 0") in a standard incubator (Steri-Cult Model 3033, Forma Scientific) to allow cells to adjust to their new culture environment. Media was replaced daily.

Construct preparation for mechanical stimulation

All constructs were transferred from multi-well plates to individual 35 mm glass bottom tissue culture dishes (P35G-0-10-C, MatTek Corp, Ashland, MA) which were covered with Teflon FEP film (100Å; DuPont, Circleville, OH) by means of a custom delrin ring. The film, permeable to O_2 and CO_2 but not to water vapor, ensured sterility and prevented evaporation. The dish design is provided elsewhere (see Fig. 5.3c; [2]).

Dynamic compressive stimulation

Constructs were subjected to dynamic compression in a custom-built bioreactor housed in an electromagnetic testing system (ElectroForce 3200; BOSE Corp., Eden Prairie, MN) [2]. Briefly, this bioreactor consisted of twelve testing stations and a chamber to sustain a cell culture environment ($37^{\circ}C$, 5% CO₂, \geq 75% relative humidity). Six of the twelve stations were equipped with loading rods to stimulate the constructs and the remaining six were devoid of loading rods and housed the NS controls.

Before starting dynamic compression, the environmental chamber of the bioreactor was aseptically cleaned and the system's culture environment was allowed to achieve steady state culture conditions. Culture dishes were then placed into each holding station. Consistent with previous patterns[169], constructs were subjected to the compressive strain profile described above. Feeding media was replaced daily. At specified time points, dishes containing treated and control constructs were removed from the bioreactor.

Fluorescence Microscopy

At each time point after the end of stimulation, the constructs subjected to RFU analysis were washed in PBS for 1 hour with gentle shaking to remove media and then visualized for ECFP and GFP-T fluorescence in a fluorescence microscope (Axiovert 25, Carl Zeiss Inc., Göttingen, Germany). To rule out cell auto-fluorescence, constructs were also visualized for rhodamine using an 11002VZ (Chroma) filter set.

Spectrophotometric Analysis

After imaging, constructs assigned for gene expression measurements were mechanically homogenized using a glass mortar and teflon pestle. The homogenate was suspended in 1ml of PBS and pipetted into a black bottom microplate (200µl per well in five wells). ECFP fluorescence in these digests was quantified in Relative Fluorescence Units (RFUs)[329] reading the microplate in a spectrophotometer (Spectra Max M2, Molecular Devices) using excitation/emission wavelengths of 436 nm/486 nm, respectively, with a cut off filter of 475 nm. ECFP RFU was normalized to day 0 values for all samples.

Real-time quantitative RT-PCR

RNA extraction, conventional gene expression analysis and real-time quantitative PCR (qRT-PCR) were performed according to published protocols [167]. Mouse specific primers were used for Type II collagen and glyceraldehyde-3-phosphate dehydrogenase (GAPDH) gene expression. Before use in the experiment, all primers were extensively tested under conventional and real-time quantitative RT-PCR conditions to ensure: 1) they were specific for each gene, and 2) that we obtained a single clean band when running the electrophoresis gel (with the associated base pair size for each gene). The absolute amount of the corresponding gene mRNA in each construct was obtained from the corresponding gene standard curve. Expression was normalized by calculating the ratio between type II collagen and GAPDH genes for each sample. GAPDH was used as the housekeeping gene, because GAPDH did not change with treatment conditions when we mechanically stimulated cell-scaffold constructs in tension containing mesenchymal stem cells from 10 week old double transgenic mice and from skeletally mature NZ white rabbits [163, 167].

Determination of Cell Viability

At each time point, cell viability in constructs was determined using an MTT assay (Vybrant MTT Cell Profileration Assay Kit, Invitrogen, Carlsbad, CA), in which live cells reduce MTT to a strongly pigmented formazan product to provide colorimetric indicators of cell viability,

following the manufacturer's instructions. Briefly the cell-agarose constructs were homogenized using a mortar and pestle. The homogenate was incubated for 4 hours in an incubator in 500µl of 12mM MTT solution. This was followed incubation for 18 hours in 5ml SDS-HCL to completely dissolve the formazan. The absorbance of formazan was measured at 570 nm in a microplate spectrophotometer (Spectra Max M2, Molecular Devices, Sunnyvale, CA) [331]. The viability results were normalized to day 0 constructs.

Determination of Type II Collagen Content

A native Type II collagen detection kit (Chondrex Inc, Catalog #6009, Redmond, WA) was used to quantify Type II collagen. The three constructs from each treatment group and animal were pooled. Each construct was washed in PBS for 1 hour with gentle shaking to remove media. The constructs were then lyophilized, their dry weight measured, and incubated in cold water at 4°C. Constructs were subjected to pepsin digestion and elastase digestion following manufacturer's protocols. ELISA was measured according to these protocols [332].

Biomechanical evaluation of the agarose-cell constructs

Aggregate modulus of the cell-agarose constructs was determined using confined compression testing. Constructs were placed in cryovials and stored at - 80°C. Prior to testing, each tissue specimen was thawed in room temperature phosphate buffered saline (PBS) for 10 minutes. A 6-mm diameter core was created from the center of each construct and its height measured using a light-force micrometer (Model ID-C1012CE, Mitutoyo Corp., Japan). Each sample was placed into a 6-mm diameter x 4.3-mm deep well and subjected to a constant contact stress of 0.066 MPa until equilibrium displacement was achieved. During testing, specimen height and applied load were continuously measured by a fine resolution LVDT (\pm 0.25", Sensotec Corp., Columbus, OH, U.S.A.) and load cell (5lb Sensotec Corp., Columbus, OH, U.S.A.), respectively. Tissue aggregate modulus (H_A) was calculated using the formula H_A= (Fo/ πa^2) / (u/h) where Fo is applied compressive force, a is specimen radius, h is specimen thickness and u is equilibrium displacement [330].

Statistical Analysis

We determined the presence of treatment-related differences among values for all response measures (ECFP RFUs, Col2 mRNA, aggregate modulus, cell viability and Type II collagen content) using a two-way ANOVA with Tukey's HSD post hoc testing (SAS software). We chose animal as a random factor, culture time and compressive stimulation as fixed factors, and response measures as dependent variables [334]. Significance was set at p < 0.05.

5.3. Results

Qualitative inspection of constructs showed cells with rounded morphology, distributed homogenously in the constructs. Fluorescence microscopy revealed ECFP fluorescence at day 0 in all 3-D constructs which was maintained in 7, 14, 21 and 28 day constructs (Fig. 5.1). No qualitative differences were apparent in the amount or intensity of ECFP fluorescence among the images for non-stimulated and stimulated constructs at the four time points (Fig. 5.1 A-G). Only cyan fluorescing cells were observed in the sections examined. No cells in any of the treatment groups expressed GFP-T and hence we did not measure RFUs or perform qRT-PCR for GFP-T and Col1a1 expression, respectively. No autofluorescence was detected in samples using rhodamine filter sets.

Quantitative results showed that applying a dynamic compressive displacement significantly increased ECFP expression in the 3-D constructs. Compared to time-matched, NS controls, compressive stimulation significantly increased RFU values by 1.34 fold,1.8 fold,1.27 fold and 1.24 fold at 7,14, 21 and 28 days, respectively (p<0.05; Fig. 5.2A). There was a significant increase in RFUs in S constructs between day 7 and 14, and a significant decrease between day 21 and day 28 (p<0.05; Fig. 5.2A). There was a significant decrease in RFUs in S between day 21 and day 28 (p<0.05; Fig. 5.2A).

NON-STIMULATED

STIMULATED



Fig. 5.1. Rounded ECFP-fluorescing cells in non-stimulated (A, C, E & G) and stimulated (B, D, F & H) constructs at day 7 (A & B), day 14 (C & D), day 21 (E & F) and day 28 (G & H).

Dynamic compressive displacement also significantly increased Col2 mRNA by qRT-

PCR. Compared to NS controls and relative to GAPDH, compressive stimulation significantly

increased mRNA expression by 1.5 fold, 1.32 fold,1.36 fold and 1.45 fold at 7, 14, 21 and 28 days, respectively (p<0.05; Fig. 5.2B). There was a significant increase in Col2 mRNA in S constructs between day 7 and 14, and a significant decrease between day 21 and day 28 (p<0.05; Fig. 5.2B). There was a significant increase in Col2 mRNA in NS constructs between day 7 and day 14 and significant decrease between day 21 and day 28 (p<0.05; Fig. 5.2B). There was a between day 21 and day 28 (p<0.05; Fig. 5.2B). There was a significant decrease between day 21 and day 28 (p<0.05; Fig. 5.2B). There was a significant decrease between day 21 and day 28 (p<0.05; Fig. 5.2B). There was a significant decrease between day 21 and day 28 (p<0.05; Fig. 5.2B). There was a significant decrease between day 21 and day 28 (p<0.05; Fig. 5.2B).



Fig. 5.2. Dynamic stimulation increases (A) ECFP expression by Relative Fluorescence Units (RFU) and (B) Col2 mRNA expression by qRT-PCR at all four time periods of observation. * Significant increase compared to non-stimulated control at same time point (p< 0.05). ** Significant difference when compared to same treatment at previous time point. Data represented as mean + one standard deviation. N=6 for all groups.

The results from qRT-PCR and fluorescence analysis were also positively correlated (Fig. 5.3, $r^2 = 0.92$ with a slope for the linear regression curve of 2.09).



Fig. 5.3. RFU fluorescence of ECFP expression and qRT-PCR are positively correlated over the four time periods of evaluation (days 7, 14, 21 & 28). (Slope = 2.09; r²=0.92). Circle denotes Day 0 constructs, triangles indicate non-stimulated constructs, and squares denote stimulated constructs.

Compressive stimulation also increased Type II collagen content of S constructs, but only after 21 and 28 days of stimulation. Compared to NS controls, S constructs showed 1.40 fold and 1.24 fold increases in Type II collagen content at 21 and 28 days, respectively (p<0.05; Fig. 5.4). There was a significant increase in type II collagen content in S constructs between day 14 and 21, and between day 21 and day 28 (p<0.05; Fig. 5.4).

Compressive stimulation increased the aggregate modulus of the chondrocyte-agarose constructs, but only after 28 days of stimulation. After this period of treatment, S constructs showed a 1.67 fold increase in aggregate modulus compared to NS controls (p<0.05; Fig. 5.5). There was a significant increase in aggregate modulus in S constructs between day 0 and 7, and between day 21 and day 28 (p<0.05; Fig. 5.5).



Fig. 5.4. Type II collagen content of constructs increased significantly after 21 and 28 days of dynamic compressive stimulation. * Significant increases compared to non-stimulated controls at same time point (p< 0.05). ** Significant increase when compared to same treatment at previous time point (p< 0.05). Data represented as mean + one standard deviation. N=6 for all groups. w.w denotes wet weight.



Fig. 5.5. Aggregate modulus of constructs increased significantly after 28 days of dynamic stimulation. * Significant increase compared to non-stimulated control at same time point (p< 0.05). ** Significant increase when compared to same treatment at previous time point (p< 0.05). Data represented as mean + one standard deviation. N=6 for all groups.

No significant changes occurred in cell viability due to mechanical stimulation or for increasing time in culture (p>0.05; Fig. 5.6). Day 28 constructs showed 8% lower cell viability than day 0 constructs.



Fig. 5.6. No significant differences were measured in cell viability between stimulated and non-stimulated constructs after 0, 7, 14, 21 and 28 days (p>0.05). Cell viability exceeded 90% at all time periods in culture. Data represented as mean + one standard deviation. N=6 for all groups.

5.4. Discussion

Tissue engineers could benefit from new methodologies that rapidly and nondestructively assess the effects of various stimuli on expression of important structural genes. We chose to track changes in Col2 gene expression since Type II collagen, the primary structural protein found in articular cartilage, effectively resists ion-induced internal pressures as well as lateral strains created by compressive forces in cartilage [335]. We achieved this goal in the murine model by linking the Col2 gene to a fluorescent protein with a 24-hour half life that would allow us to detect near real-time changes in expression. While a protein with a shorter half life might have provided a "transducer-like" biological response to changes in mechanical signal, we also wanted to ensure that we did not miss observing these changes when they occurred.

Effects of Compressive Stimulation and Time on Gene Expression

Our current study using these transgenic chondrocytes was, in part, designed to examine the individual and combined effects of dynamic compression and time in culture on ECFP and Col2 gene expression. Both spectrophotometric analysis to measure ECFP RFUs and qRT-PCR to monitor Col2 mRNA expression showed similar patterns over time in culture (Figs. 5.2A and 5.2B). While NS constructs displayed modest temporal changes in RFUs (relative to Day 0 values), mechanical stimulation significantly enhanced fluorescence both compared to controls and between Day 7 and Day 14 (Figs. 5.2A, 5.7A and 5.7B). Very similar temporal and stimulation patterns were also observed by qRT-PCR (Fig. 5.2B), supporting our first hypothesis that compressive stimulation significantly increases ECFP fluorescence and Col2 mRNA expression.

Our results mirror the findings of two shorter-term studies but directly contradict other published reports showing no changes in collagen gene expression after mechanical stimulation. De Croos et.al [336] found significant increases in Col2 mRNA expression 12 hours after exposing bovine chondrocyte-calcium phosphate constructs to 1 kPa compressive pressure at 1 Hz for 30 minutes. In a similar way, Xie J et.al [337] found that 24 hours of continuous dynamic compression applied to rabbit chondrocytes seeded in microporous elastomeric scaffolds of poly(L-lactide-co-epsilon-caprolactone) significantly increased Col2 gene expression. By contrast, Hunter and coworkers found no change in Col2 gene expression when constructs containing bovine chondrocytes seeded in collagen gels were subjected to dynamic compression (± 4% strain @ 1 Hz for 24 hrs) [338]. Demarteau et al also showed no alterations in Col2 mRNA levels when human articular chondrocytes were seeded in PEGT/PBT (polyethylene glycol terephthalate/polybutylene terephthalate) foams and exposed to dynamic compression (6 cycles of sinusoidal deformation to 5 % peak strain at 1Hz followed by a 10 h rest period for 3 days) [339]. Mauck et.al [192] actually observed decreases in type II collagen promoter activity when bovine chondrocytes (transfected with pCol2-LUC promoter reporter

plasmid) were seeded in 2% w/v agarose. Mauck subjected these TECs to 10% cyclic compressive deformation at 1 Hz for either 60 or 180 mins, after which they observed decreases in Type II collagen activity 24 and 72 hours post stimulation. The differences seen among all of these studies could be attributed to different experimental conditions such as the choice of biomaterial, cell source, stimulation profile, and time of evaluation. For example, it is worth noting that our results are longer-term, showing stimulation-induced increases in Col2 mRNA expression and RFUs that persist after 28 days of stimulation. Future studies could benefit from tracking changes over longer time periods as well as systematically varying and controlling other factors to determine their importance in the tissue engineering fabrication process.

Numerous studies have sought to explain how dynamic compression increases Col2 gene expression. One study found that such increases occur through transcriptional activation, possibly through the Sp1 binding sites residing in the proximal region of the Col2a1 gene promoter [340]. Others suggest that certain phenomena like cell/tissue strain, fluid pressurization and flow, electrokinetic phenomena, convective transport, and release of cytokines/growth factors can trigger the complex chain of events that modulate the production of ECM proteins like Type II collagen [341-343]. Although not a specific objective of the current study, we do plan to study several of these mechanisms and how they might increase Col2 gene expression and ultimately Type II collagen production.

Our results also support our fourth hypothesis that ECFP fluorescence and Col2 mRNA expression are positively correlated (r²=0.92; Fig. 5.3). These correlations are similar to GFP-T vs. Col1 mRNA results from a previous study in our lab where MSCs from double transgenic mice were seeded in collagen sponge scaffolds and exposed to tensile stimulation[163]. Both sets of findings are encouraging for tissue engineering applications and suggest that tracking fluorescence changes are a reasonable surrogate for monitoring changes in mRNA expression by gRT-PCR.

Effects of Stimulation and Time on Type II Collagen Content and Aggregate Modulus

Our results support our second and third hypotheses that compressive stimulation significantly increases Type II collagen production and TEC aggregate modulus. Although Type Il collagen content relative to mRNA expression is delayed in both construct types, mechanical stimulation significantly enhances and that the amount of measured protein. While Col2 gene expression peaked between 7 and 14 days and then declined (Figs. 5.2B, 5.7A and 5.7B), Type Il collagen content showed significant increases at 21 days with further enhancement at 28 days. Both the S and NS constructs showed these significant effects (Figs. 5.4, 5.7A and 5.7B) with S constructs showing further increases in protein content at these later time periods. The fact that changes in Col2 expression occur prior to increased Type II collagen content is not unexpected. However, the fact that this delay may be as long as 14 days in culture and that mechanical stimulation can enhance protein production by as much as 25-33% are rather surprising findings and likely specific to the cells, scaffold and media conditions chosen for these experiments. It is also worth noting that increased Type II collagen content at 21 and 28 days may have: 1) hampered diffusion of nutrients, inducing small decreases in cell viability (Fig. 5.6); and 2) switched cells to more of a "maintenance mode," thereby decreasing Col2 gene expression (Fig. 5.2B). These conclusions are only speculations at this time and will be the subject of further work by our group.

Our study also demonstrated further delays in biomechanical effects of mechanical stimulation. While Type II collagen content was elevated by twenty one (21) days, an additional seven (7) days of dynamic compression (28 days total) was required to significantly improve aggregate modulus (Figs. 5.5, 5.7A and 5.7B). Again, these effects are likely dependent on the cell, scaffold, and mechanical signals imposed. One might also question whether factors like freezing the constructs before testing could have affected our biomechanical results. Freezing was required because multiple constructs had to be tested each day, each test required five hours to complete, and we had only two confined compression testers available. To directly





Fig. 5.7. Fold increase from day 0 in (A) non-stimulated constructs, (B) stimulated constructs. Note the earlier increase in Col2 expression, followed by delayed increase in type II collagen content, then followed by an increase in construct modulus. Data represented as means \pm one standard deviation. N=6 for all groups.
examine potential freezing effects, we did culture two chondrocyte-agarose constructs from each of five cell lines (from 5 mice) for seven (7) days without mechanical stimulation and then froze one of each pair. Unconfined compression testing revealed no significant effect of freezing on aggregate modulus between groups (p>0.05). Our lab has also previously shown that freezing does not affect mechanical properties of MSC-collagen constructs [344].

Our improvements in construct aggregate modulus are similar to those of Mauck et al [169, 326], who stimulated bovine chondrocyte-agarose TECs to 10% cyclic compressive deformation. This stimulus profile significantly increased equilibrium aggregate modulus compared to time-matched, non-stimulated constructs. However, these studies also showed that aggregate modulus of non-stimulated constructs actually decreased between day 28 and day 55. These observations taken together with our results suggest that time in culture and dynamic stimulation can each significantly increase the matrix structure and biomechanics of chondrocyte-agarose constructs but that new strategies may be required to sustain these improvements in longer-term culture.

Overarching conclusions from the current study

Among the most important findings in our study are the temporal and stimulationinduced relationships among gene expression, protein expression and biomechanics. Most studies in the literature have examined the effects of stimulation on only one or two of these response measures. Instead, we chose to simultaneously monitor treatment-induced effects on all three measures. The increases we observed in Col2 gene expression at 7, 14, 21 and 28 days only culminated in increased Type II collagen content after 21 and 28 days and increased aggregate modulus after 28 days. Gene expression must obviously precede protein deposition, but any biomechanical benefits arise only after the protein is assembled into a more functional matrix. Additional studies are needed to corroborate these temporal findings in the murine model and in other model systems and to determine if gene and protein expression patterns might serve as predictors of biomechanical response, both in-vitro and ultimately in-vivo.

Using a murine chondrocyte source to develop tissue engineered constructs offers certain advantages and disadvantages. Although repairing cartilage defects in the mouse model using FTE principles remains a distinct challenge, murine cells offer molecular tools that are not currently available in higher models such as rabbits, goats and sheep. Such biological tools also exist in the human model, but human chondrocytes are not readily available. Performing in-vitro experiments in the mouse that can be successfully correlated to results in higher models may someday permit translation to humans. Finally, the murine model is an ideal candidate to evaluate signaling pathways involved in mechanotransduction. For example, our group is already examining the effects of mechanical stimulation on BMP/FGF signaling pathways with and without inhibitors.

Our study is not without limitations. 1) We did not track changes in expression of ECFP with passage number. Reduction in ECFP levels might indicate that these chondrocytes are dedifferentiating, hence prolonging any future increases in gene and protein expression and subsequent construct biomechanics. It has been shown previously that chondrocytes can rapidly dedifferentiate after even one passage and could exhibit more of a fibroblastic phenotype with increased Col1 expression [345-346]. However, none of the cells in the current study showed any GFP-T fluorescence up to P3, a marker that would indicate Col1a1 expression. In future studies, we plan to use our double transgenic cells to monitor the effects of cell passage number and loading type on both Col2a1 and Col1a1 gene expression, Type II collagen content and construct biomechanics. 2) We did not track real-time changes in Col2a1 expression in the intact construct. These structures are currently too thick and opaque to perform such non-destructive imaging. We continue to seek novel imaging technologies to observe such changes in cellular fluorescence that would permit us to perform longitudinal studies on the same constructs over time. 3) It was necessary to pool fluorescence from all cells extracted from constructs undergoing the same experimental condition. Thus, we can report only average rather than specific or local effects of compression on individual cells. 4) We

used juvenile chondrocytes because they could be easily extracted from the ribs of the newborn mice and because of our interest in examining how developmental biology might be applied to adult tissue healing. We also plan to study how adult chondrocytes respond to these stimuli. 5) Autofluorescence of agarose gel fragments could increase overall RFU values. In this study, we assumed that these changes were constant across groups. 6) While Type II collagen content and biomechanics significantly increased after 21 to 28 days in culture, these response measures do not directly indicate the degree of collagen assembly or alignment. 7) We did not specifically track changes in the production of proteoglycans like aggrecan. Increases in aggrecan content could reflect improvements in a construct's aggregate modulus. In the future we intend to track the production of proteoglycans. 8) We performed only confined compression testing on the constructs. Future studies should apply other testing modes like unconfined compression or indentation. 9) We applied only one compressive stimulus pattern to the constructs. We still need to systematically vary components of the mechanical signal so as to optimize gene and protein expression as well as construct biomechanics in the shortest possible time interval. 10) Other forms of stimulation (e.g. chemical stimulation with TGF- β) may also induce increases in these response measures. In the future, we intend to contrast results using these methods with those from our current study.

In conclusion, our results reveal that both compressive stimulation and time in culture increase Col2 gene expression, Type II collagen content and aggregate modulus in constructs formed from murine chondrocytes seeded in agarose gels. The transgenic and bioreactor technologies that we report hold much promise in optimizing tissue engineering methodologies for replacement of damaged and diseased tissues like articular cartilage and fibrocartilage. In particular, we hope to soon identify which profiles of environmental stimulation (mechanical, chemical, etc.) result in the most rapid and greatest increases in Col2 gene expression levels. Such innovative technologies will be needed to discover promising tissue engineering treatments to speed the repair of damaged and diseased tissues.

Chapter 6

Compressive Loading Applied To Double Transgenic MSC-Agarose Constructs Alters Collagen II Gene Expression, Collagen II Protein Expression And Aggregate Modulus

The objectives of this study were to determine how long-term dynamic compression, applied to murine MSC-agarose constructs, influences construct stiffness, collagen 2 gene expression and synthesized type II collagen protein. MSCs were harvested from the long bones of six ten-week old double transgenic mice carrying transgenes that display enhanced cyan fluorescent protein (ECFP) and green fluorescent protein (GFP-T) as reporters for expression from the Col2a1 and Col1a1 promoters, respectively. Fifty-four constructs (8mm dia x 3mm thick) per animal were created by seeding MSCs (10x10⁶ per ml) in agarose gel (2% w/v). Twenty-four constructs from each animal were stimulated for 7, 14, 21 or 28 days in a custom bioreactor housed in an electromagnetic system. Another twenty-four constructs were exposed to identical culture conditions but without mechanical stimulation and thus served as nonstimulated controls at 7, 14, 21 and 28 days. The remaining six constructs served as Day 0 controls. Fluorescing cells with rounded morphology were present in only the day 14, 21 and 28 constructs. Compared to time-matched, non-stimulated controls, 21 and 28 days of stimulation significantly increased Col2 expression as measured by ECFP fluorescence, type II collagen content, and aggregate modulus. There were also significant increases in Col2 expression and type II collagen content in stimulated constructs between day 21 and day 28. Future studies will examine how components of the in-vitro mechanical signal affect these same response measures and address the question as to whether such in-vitro improvements in the tissue engineered construct enhance repair outcome after surgery.

6.1. Introduction

Articular cartilage injuries heal poorly because of the lack of blood supply and present a significant clinical problem [347-349]. These problems have motivated some surgeons to implant osteochondral allografts which can fill a cartilage defect with mature tissue, but which also raise concerns about disease transmission, immune response and proper height and integration into the defect [350-362]. Others have used cell-based therapies such as autologous chondrocyte implantation for the repair of cartilage defects [363-364]. This approach has provided patients with some pain relief but concerns remain about the structural integrity and integration of the cells and periosteal flap initially and the synthesized matrix in the longer term [348, 365]. Consequently, researchers are now investigating various chondrocyte-seeded scaffold constructs [366-369] in the laboratory. However, a major limiting factor using this approach is whether sufficient autologous chondrocytes can be harvested. This problem is especially evident in older and osteoarthritic patients who exhibit decreased chondrogenic capacity [370-371], proliferation rates, collagen synthesis and sensitivity to cytokines and growth factors [372-374]. Studies have also indicated that chondrocytes from these OA patients show degenerative alterations and DNA damage [375]. Using autologous chondroctyes also poses the risk of donor-site morbidity [376].

Mesenchymal stem cell (MSC)-based therapies have the potential to overcome some of the disadvantages of chondrocyte-based therapies. MSCs are a promising alternative cell source for cartilage repair due to their ease of isolation and expansion and their chondrogenic differentiation potential [151, 376-380]. For example, MSCs undergo chondrogenic differentiation in the presence of growth factors [381-386]. Dynamic compressive loading and hydrostatic loading also enhance MSC differentiation along a chondrogenic lineage [198, 387-393]. In the absence of growth factors, dynamic compression of constructs containing bone marrow stromal cells stimulates Collagen II gene expression [390]. Dynamic hydrostatic loading in the physiological range also upregulates chondrogenic gene expression [392-393] and

protein synthesis [198, 392]. Taken together, these studies support the advantages of using dynamic loading to stimulate MSC constructs for cartilage tissue engineering.

Our laboratory has developed novel transgenic mouse and bioreactor technologies to study loading-induced effects on TECs containing cells like MSCs. We have created double transgenic mice in which the Col1 and Col2 genes are linked to intracellular Green Fluorescent Protein- Topaz (GFP-T) and Enhanced Cyan Fluorescent Protein (ECFP), respectively [162-163]. Cells from these mice fluoresce green when the Col1 gene is activated and cyan when the Col2 gene is activated. Using MSCs from these mice we have shown that tension upregulates Col1 expression [163] while compression of chondrocytes upregulates Col2 expression [162]. We have also demonstrated that gene expression by fluorescence and qRT-PCR are positively correlated [162-163]. Finally, we have created and validated bioreactors capable of providing micron-level precision waveforms to stimulate constructs in compression and tension [2].

The objective of the current study is to utilize these technologies to examine the effects of dynamic compression on 3-D tissue engineered constructs (TECs) containing murine double transgenic MSCs in agarose scaffolds. By depositing these cells in scaffolds and placing the constructs in our bioreactor, we sought to test the following hypotheses: that compared to the results for non-stimulated TEC controls, compressive stimulation up to 28 days in culture increases; 1) the amount of ECFP fluorescence (an indicator of Col II gene expression); 2) the amount of Type II collagen content; 3) TEC aggregate modulus.

6.2. Materials and Methods

Experimental Design

MSCs were obtained from the bone marrow of six (6) ten-week old double transgenic mice as previously described [163, 394-395]. Fifty four cell-agarose constructs (8 mm dia. x 3 mm thick) were created per animal using one cell density of 10 x 10^6 cells/ml in a 2% w/v agarose scaffold [162]. Twenty-four (24) TECs were stimulated (S) and the other twenty-four (24) TECs were non-stimulated (NS). Six (6) S and six (6) NS TECS were assigned to each of

four time periods of assessment (7, 14, 21 and 28 days in culture). The remaining six (6) constructs served as day 0 NS controls. S constructs all received the same compressive stimulation profile (sinusoidal displacement pattern @ 1Hz to 10% peak strain for one hour followed by one hour of rest, repeated three times a day) [162, 366]. The six (6) S and six (6) NS constructs at each time point were assigned as follows: one for ECFP fluorescence in a spectrophotometer (measured in Relative Fluorescence Units (RFU)) [162, 396], one for confined compression to determine aggregate modulus [162, 330], one for cell viability using an MTT assay [162, 397], and three constructs for Type II collagen content using ELISA [162, 398].

Mesenchymal stem cells

In accordance with protocols established by University of Cincinnati's Institutional Animal Care and Use Committee (IUCAC), mice were euthanized by CO₂. Femur and tibia were isolated and were scraped to remove all the soft tissues. Bones were then transected under sterile conditions and placed in modified centrifuge tubes. 400g of centrifugation for one minute was used to extract the bone marrow [163, 394-395]. These bone marrow cells were initially plated at concentration of 7.5 x 10⁶ cells/100 mm dishes and fed media with supplements (MesenCult[™] Basal Medium for Mouse Mesenchymal Stem Cells and Mouse Mesenchymal Stem Cell Stimulatory Supplements, Stem Cell Technologies, Vancouver, BC) [163, 394-395]. The se dishes were washed with phosphate buffered saline (PBS, Gibco BRL/Life Technologies Inc., Gaithersburg, MD) after two days to remove non-adherent cells, and fed fresh media [163, 394-395]. After seven to ten days in culture, cells at P0 were trypsinized and replated at 1 x 106 cells/100mm dishes and cultured for one additional week [163, 394-395]. This process was repeated until cells reached P4 [163, 394-395].

Construct Creation

To produce a cell-gel solution of 10 x 106 cells/ml in 2% (w/v) agarose, MSCs were suspended at 20x10⁶ cells/ml in media (BGjB, Invitrogen-Gibco BRL/Life Technologies Inc., Gaithersburg, MD) containing 1% antibiotic/antimycotic and 10% fetal bovine serum (FBS;

Atlanta Biologicals, Lawrenceville,GA) and mixed in equal volumes with 4% (w/v) agarose (TYPE VII, low gelling temperature, Fisher Scientific) and PBS. Sterile glass plates with a 3mm spacer were used to cast this cell-gel solution. The solution was allowed to gel for 20 minutes and eight (8)-mm diameter samples were then cored. Cells were then cultured in these agarose constructs under free swelling conditions in a standard incubator (Steri-Cult Model 3033, Forma Scientific) for two days (denoted henceforth as Day 0). Media was replaced daily.

Compressive Stimulation

To maintain sterility and to prevent media evaporation, dishes containing the constructs were covered with a Teflon FEP film (100Å; DuPont, Circleville, OH) held in place by a delrin ring [2, 162]. These dishes were then placed into each well of the custom-built bioreactor [2]. Constructs were then subjected to dynamic compression profile as described in Experimental Design above [2, 162]. Feeding media was replaced daily.

Fluorescence Analysis

Cyan Fluorescence in the constructs was analyzed qualitatively and quantitatively. Each stimulated and non-stimulated construct was visualized in a fluorescence microscope (Zeiss Axiovert) with a filter set (XF1142, Omega) for cyan fluorescence [2, 162]. Constructs were further visualized using filter sets for green and red fluorescence (XF104-2, Omega and 11002VZ, Chroma) to rule out autofluorescence [2, 162]. TECs were then mechanically homogenized to quantify ECFP fluorescence. Constructs were homogenized using a glass mortar and Teflon pestle. The minced constructs were then placed in a spectrophotometer (Spectra Max M2, Molecular Devices) where ECFP RFUs were quantified using excitation/emission/cutoff wavelengths of 436nm/486nm/475nm, respectively [162].

Determination of Cell Viability

At each time point, cell viability in constructs was determined using an MTT assay (Vybrant MTT Cell Profileration Assay Kit, Invitrogen, Carlsbad, CA). The cell-agarose constructs were first homogenized using a mortar and pestle and incubated in 500µl of 12mM MTT solution for 4 hours. This was followed by additional incubation in 5ml SDS-HCL for 18 hours to completely dissolve the formazan. Formazan absorbance was measured at 570 nm in a microplate spectrophotometer (Spectra Max M2, Molecular Devices, Sunnyvale, CA) [162, 397]. Viability results were normalized to Day 0 constructs.

Determination of Type II Collagen Content

Type II collagen content was quantified using native Type II collagen detection kit (Chondrex Inc, Catalog #6009, Redmond, WA) [398]. Three constructs from each treatment group and animal were then pooled for such quantification. The details of the assay procedure are described elsewhere [162]. Briefly constructs were lyophilized, their dry weight measured, and subjected to pepsin digestion and elastase digestion following manufacturer's protocols. ELISA was measured according to these protocols[398].

Biomechanical evaluation of the agarose-cell constructs

Constructs for biomechanical evaluation were placed in cryovials and stored at -80°C Prior to testing each tissue specimen was thawed at room temperature in phosphate buffered saline. Aggregate modulus of the cell-agarose constructs was determined using confined compression testing as previously described [162]. Briefly a 6-mm diameter core was created from the center of each construct and subjected to a constant contact stress of 0.066 MPa until equilibrium displacement was achieved. During testing, specimen height and applied load were continuously measured. Tissue aggregate modulus (HA) was calculated using the formula HA= $(Fo/\pi a2) / (u/h)$ where Fo is applied compressive force, a is specimen radius, h is specimen thickness and u is equilibrium displacement[162].

Statistical Analysis

SPSS software was utilized to perform two way ANOVA with Tukey's HSD post hoc test. to determined the presence of treatment-related differences among values for all response measures. Significance was set at p < 0.05.

6.3. Results

Fluorescence Microscopy

Inspection of the TECs showed the presence of homogeneously distributed rounded cells. Cells did not fluoresce ECFP in any of the constructs at day 0 or at day 7. ECFP fluorescent cells were found in all 14, 21 and 28 day constructs (Fig. 6.1). However, no apparent differences in ECFP fluorescence were observed among the S and NS constructs at corresponding time points.

Spectroscopic Analysis

Quantitative results showed that applying a dynamic compressive displacement significantly increased ECFP expression in the 3-D constructs at only later time points. Compared to time-matched, NS controls, compressive stimulation significantly increased RFU values by 1.44 fold and 1.63 fold at 21 and 28 days, respectively (p<0.05; Fig. 6.2). There was also a significant increase in RFUs in the S constructs between day 21 and 28, (p<0.05; Fig. 6.2).

Type II Collagen Content

Similar to the RFU results by spectroscopic analysis, compressive stimulation increased Type II collagen content in S constructs, but only after 21 and 28 days of stimulation. Type II collagen content was not detectable in day 0 and day 7 samples. Compared to NS controls, S constructs showed 1.72 fold and 1.67 fold increases in Type II collagen content at 21 and 28 days, respectively (p<0.05; Fig. 6.3). There was also a significant increase in type II collagen content in S constructs between day 21 and day 28 (p<0.05; Fig. 6.3).



Fig. 6.1. Rounded ECFP-fluorescing cells in non-stimulated (A, C & E) and stimulated (B, D & F) constructs at day 14 (A & B), day 21 (C & D) and day 28 (E & F).



Fig. 6.2. Dynamic stimulation increases ECFP expression by Relative Fluorescence Units (RFU) at day 21 and day 28. * Significant increase compared to nonstimulated control at same time point (p< 0.05). ** Significant difference when compared to same treatment at previous time point. Data represented as mean + one standard deviation. N=6 for all groups.



Fig. 6.3. Type II collagen content of constructs increased significantly after 21 and 28 days of dynamic compressive stimulation. * Significant increases compared to non-stimulated controls at same time point (p< 0.05). ** Significant increase when compared to same treatment at previous time point (p< 0.05). Data represented as mean + one standard deviation. N=6 for all groups. ww denotes wet weight.

Aggregate Modulus

Compressive stimulation significantly increased the aggregate modulus of the chondrocyte-agarose constructs, but only after 21 and 28 days of stimulation. S constructs showed 1.37 fold and 1.55 fold increases in aggregate modulus compared to NS controls at day 21 and day 28, respectively (p<0.05; Fig. 6.4).

Cell Viability

No significant changes occurred in cell viability due to mechanical stimulation or for increasing time in culture (p>0.05; Fig. 6.5). Day 28 constructs showed 7% lower cell viability than day 0 constructs.



Fig. 6.4. Aggregate modulus of constructs increased significantly only at 21 and 28 days of dynamic stimulation. * Significant increase compared to non-stimulated control at same time point (p< 0.05). Data represented as mean + one standard deviation. N=6 for all groups.



Fig. 6.5. No significant differences were measured in cell viability between stimulated and non-stimulated constructs after 0, 7, 14, 21 and 28 days (p>0.05). Cell viability exceeded 95% at all time periods in culture. Data represented as mean + one standard deviation. N=6 for all groups.

6.4. Discussion

Understanding the response of MSCs to physical forces is of critical importance when developing cell-based therapies for cartilage repair. Developing methods that permit rapid visualization of desirable gene expression profiles saves time in identifying the most promising treatment strategies. We chose to develop novel murine transgenic technologies and bioreactors in order to study the effects of compression on gene and protein expression as well as biomechanics for these MSC-agarose scaffolds. Based on our prior work stimulating transgenic chondrocytes in culture, we naturally hypothesized was that dynamic compression applied daily to MSCs in agarose culture would enhances Col2 gene expression, type II collagen content and aggregate modulus of constructs, thus resulting in the formation of stiffer and more functional cartilaginous tissue matrices.

Our quantitative fluorescence results support our first hypothesis that compressive stimulation significantly increases ECFP fluorescence and hence Col2 mRNA expression. The compressive stimulus did increase both measures but unlike our prior study using chondrocytes,

only after longer time points of stimulation. In fact, the cells in our current study only expressed ECFP fluorescence after 14 days of stimulation. These results agree with those found by others examining Col2 gene expression by MSCs [390, 399]. In these studies, MSCs in agarose culture also enhanced Col2 gene expression, but only after 2 weeks in culture. Our results involving dynamic compression are also consistent with prior reports which showed increased Col2 gene expression by stimulating MSCs, but only at time points in culture [203, 389]. It is worth noting, however, that the changes we observed in Col2 gene expression by MSCs in the current study are nearly as rapid as the changes in expression by chondrocytes which show increased Col2 gene expression as early as 2 days in culture [162, 336-339]. The fact that MSCs are slower to respond to biophysical stimulation than fully differentiated articular chondrocytes is not altogether surprising. It has suggested, for example, that the developing pericellular and extracellular matrix may be regulating mechano-stimulation of MSCs undergoing chondrogenesis [400]. Furthermore, cartilage matrix synthesis by articular chondrocytes in dynamically- compressed agarose gels is known to be greater at later time points than earlier time points [162, 401], implying that a well-developed matrix may be required for the transduction of mechanical compression to elicit a cellular biosynthetic response. This is further supported by the fact that the amount of type II collagen in our constructs is greater at later culture periods.

Consistent with our findings for gene expression, our other results support our second and third hypotheses that compressive stimulation significantly increases type II collagen content and aggregate modulus. While both hypotheses were supported, we again found that compressive stimulation only affected protein expression and biomechanics at later time intervals. Similar to the findings for Col2 gene expression, 21 and 28 days of compressive strain were required to increase Type II collagen content (Fig. 6.2). This temporal increase in type II collagen content at later points is similar to the results reported for MSCs [389, 400] and chondrocytes [162]. In a similar fashion, 21 and 28 days of stimulation were required to

positively influence the aggregate modulus of the TECs. Such improvements in aggregate modulus relative to time-matched controls may seem delayed in time, but are still faster than the 42 days required to alter the aggregate modulus of porcine chondrocyte-agarose TECs when subjected to 10% cyclic compressive deformation [400]. When these observations are coupled with our results, it is clear that time in culture and dynamic stimulation can each significantly increase the matrix synthesis and biomechanics of MSC-agarose constructs. However, knowing whether these positive effects can be sustained will require longer-term culture experiments (which are now underway in our lab) and possibly new strategies should gene expression increases decline as we previously found after stimulating chondrocytes [162].

The fact that temporal and stimulation-induced relationships among gene expression, protein expression and biomechanics differ between chondrocytes and MSCs is worth further study. For example, in the chondrocyte-agarose model, the increases we observed in Col2 gene expression at 7, 14, 21 and 28 days only culminated in increased Type II collagen content after 21 and 28 days and increased aggregate modulus only after 28 days [162]. For the MSC-agarose model system, however, the increases we observed in Col2 gene expression were synchronized with the increases in Type II collagen content and aggregate modulus. This suggests differential maturation and structure–function relationships between MSC- and chondrocyte-seeded agarose TECs. Others have also found similar differences between MSCs and chondrocytes [402]. Our system permits these questions to be answered and will be the subject of our ongoing research.

Our study is not without limitations. 1) Autofluorescence of agarose gel fragments may have led to increases in measured RFU values. In this study, we assumed that these changes were constant across groups. 2) Quantification of RFU is done by pooling of all cells and hence we know only the average effect of dynamic compression and not the specific effects of compression on individual cells. Ultimately, more local recordings may be required since a prior study found that gene expression varies in the core vs. the peripheral region of the cell-seeded

hydrogel [389]. 3) Our type II collagen content quantification does not indicate the degree of collagen assembly or alignment. 4) We did not track changes in the production of proteoglycans like aggrecan. Increases in aggrecan content could likely reflect improvements in a construct's aggregate modulus although type II collagen is essential for confining the expansion of GAGs on the aggrecan molecule. In the future we intend to track the production of proteoglycans like aggrecan to more fully understand TEC composition and architecture. 5) These in-vitro studies in culture may not reflect the functionality of TECs once they are implanted at surgery. The murine model represents a continual challenge in this regard due to its small size. Future transgenic studies will need to be performed in larger model systems where surgery can be controlled.

Our results have shown that compressive stimulation and time increase type II collagen gene expression, protein content and aggregate modulus in murine MSCs seeded in agarose gel. We expect that such increases in gene expression will result in rapid synthesis and accumulation of collagen. Future studies will seek to optimize which aspects of the mechanical signal produce the greatest increases in Col2 gene and protein expression as well as aggregate modulus. Such improvements can then be translated to larger animal and tissue repair models where surgery can be even better controlled.

Chapter 7

Influence Of Compressive Loading On Collagen I Gene Expression And Aggregate Modulus In Tissue Engineered Constructs Containing Double Transgenic Fibroblasts Seeded In Agarose

The objectives of this study were to determine how up to 28 days of in-vitro dynamic compression, applied to murine fibroblast-agarose constructs, influences Col1 and Col2 gene expression levels as well as aggregate modulus of the tissue engineered constructs (TEC). Fibroblasts were harvested from the tail tendons of six ten-week old double transgenic mice. These mice carry transgenes that use enhanced cyan fluorescent protein (ECFP) and green fluorescent protein (GFP-T) as reporters for expression from the Col2a1 and Col1a1 promoters, respectively. Twenty-seven constructs (8mm dia x 3mm thick) per animal were created by seeding MSCs (10x10⁶ per ml) in agarose gel (2% w/v). Twelve constructs from each animal were stimulated (S constructs) for 7, 14, 21 or 28 days (n=3 each) in a custom bioreactor housed in an electromagnetic system. Another twelve constructs were not stimulated (NS controls) but were exposed to identical culture conditions for the same time periods. The remaining three constructs served as day 0 controls. GFP-T fluorescing cells were present in all constructs. No ECFP expressing cells were seen in constructs. Compared to time-matched, NS controls, 28 days of stimulation significantly decreased GFP-T fluorescence. Compared to timematched controls, stimulation significantly increased the linear stiffness of constructs at 28 days. This study reveals that compressive stimulation downregulates Col1 gene expression, does not induce Col2 gene expression, but does increase aggregate modulus in compression. Future studies will examine the effects of components of the mechanical signal in culture and address the question of whether long term stimulation would further decrease Col1 gene expression and possibly even lead to detectable Col2 gene expression in culture.

7.1. Introduction

Tendons that wrap around bone have adapted by modulating their structural and biochemical characteristics. Tissues like the rabbit flexor digitorum profundus (FDP) tendon create a fibrocartilaginous (FC) region after birth [13-15]. These contact regions show a basket-weave collagen arrangement [403] and their proteoglycan and GAG contents are two to ten times higher than values in the adjacent normal tendon [404] .Cells in this fibrocartilage rich region also display a more rounded morphology than tenocytes in nearby tendon tissue [405-406].

Many studies suggest that this fibrocartilage (FC) region develops after birth in response to local compressive forces from the opposing bone surface. At birth, the tendon near the bone shows no evidence of fibrocartilage[407].Instead the FC zone forms during early postnatal development, presumably in response to changing physical activity levels [206, 266]. When investigators exposed tendon explants to continuous in-vitro cyclic compression, sulfate incorporation into biglycan and proteoglycan increased up to 1.5-fold and 3- fold, respectively [408]. Similar in-vitro studies demonstrated how compressive loading increased both aggrecan and proteoglycan expression [409-411]. Moreover, surgically translating the tendon to eliminate compressive stresses on the FC-rich contact zones rapidly disrupts the basket-like weave and depletes GAGs [13] and depletion of GAGS [412]. These studies mentioned above demonstrate that the compressive stimuli play a central role in the fibrocartilage formation.

While in-vivo [13, 412] and in-vitro[408-411] studies have been performed on tendons in compression, only a few studies have examined the effects of dynamic compression on fibroblast-seeded scaffolds[206, 413]. Shim et.al applied cyclic hydrostatic pressure to Achilles fibroblasts and observed decreases in Col1 gene expression and upregulation of Col2 and aggrecan gene expression [206]. Elder et.al also applied cyclic hydrostatic pressure to murine embryonic fibroblasts and observed increases in sGAGs [413]. Dynamic compression applied to chondrocyte-seeded scaffolds increases Col2 gene expression [162, 336-339] and applied to

MSC-seeded scaffolds enhances the chondrocytic nature of MSCs [387-391]. However, studies that apply dynamic compression to fibroblasts are needed. Thus the goal of this study was to evaluate the effects of dynamic compression on fibroblast-seeded scaffolds.

Fluorescent transgenic mouse technologies and bioreactors have been created in our lab that permits us to study the effects of compression on constructs containing fibroblasts. In these mice the Col1 gene is linked to Green Fluorescent Protein-Topaz (GFP-T) and the Col2 gene is linked to Enhanced Cyan Fluorescent Protein (ECFP), respectively [162-163]. Thus the cell fluoresce either green or cyan if Col1 or Col2 gene is expressed respectively. Studies from our lab show that MSCs from these mice upregulate Col1 in tension [163] and chondrocytes from these mice upregulate Col2 in compression [162]. When gene expression by fluorescence is quantified using a spectrophotometer, the relative fluorescence units (RFUs) and PCR are positively correlated [162-163]. We have also created and validated a bioreactor capable of providing micron precise waveforms to stimulate constructs [2].

Hence the objective of this study was to utilize these technologies to examine the effect of dynamic compression on 3-D murine fibroblast-constructs. By seeding these fibroblasts in scaffolds and placing the constructs in our bioreactor, we hypothesized that compared to nonstimulated constructs at equivalent time points, compressive stimulation would: 1) decrease GFP-T fluorescence, a marker for Col1 gene expression; 2) increase ECFP fluorescence, an indicator of Col2 gene expression; and 3) increase the construct's aggregate modulus in compression.

7.2. Materials and Methods

Experimental Design

Tail tendon fibroblasts were obtained from six (6) ten-week old double transgenic mice. 2% w/v agarose with fibroblasts at a density of 10×10^6 cells/ml were created. A total of twenty-seven (27) cell-agarose constructs (8 mm dia. x 3 mm thick) were created per animal. [162]. Of these twelve (12) of the constructs were stimulated (S) in compression and another twelve (12)

served as non-stimulated (NS) controls at days 7, 14, 21 and 28 (n=3 each). The remaining three (3) constructs served as day 0 NS controls. S constructs all received the same compressive stimulation profile (sinusoidal displacement pattern @ 1Hz to 10% peak strain for one hour followed by 1 hour of rest, repeated three times a day) [162, 366]. The three (3) S and three (3) NS constructs at each time point were assigned as follows: one for ECFP fluorescence in a spectrophotometer, [162-163, 396]one for confined compression to determine aggregate modulus [162, 330], and one for cell viability using an MTT assay [162, 397].

Fibroblast isolation and cultivation

Mice were euthanized by CO_2 according to approved Institutional Animal Care and Use Committee (IACUC) protocols. Tails from these mice were harvested aseptically and the skin removed to expose the parallel tail tendons. Each tendon was dissected free from bone and placed in sterile 100mm culture dishes. Tendons were minced into small pieces and a few ml of FBS was added. After 2 days, 10 ml of media (Advanced DMEM, 10% FBS, pen/strep) was added to each plate. Adherent cells were allowed to grow for 10 to 14 days before first passage. Cells were then trypsinized and replated at 1 x 10⁶ cells/100mm dishes and cultured for one additional week. Cells were then repeatedly sub-cultured at a density of 1 x 10⁶ cells/100 mm dish until passage four (P4).

Construct Creation

A 4% (w/v) agarose solution was created under sterile conditions and maintained at over 400C in a water bath to prevent gelatin. Fibroblasts at P4 were trypsinized and resuspended in media Advanced DMEM, Invitrogen-Gibco BRL/Life Technologies Inc., Gaithersburg, MD) containing 1% antibiotic/antimycotic and 10% fetal bovine serum (FBS; Atlanta Biologicals, Lawrenceville,GA) at a concentration of 20x106 cells/ml. Equal volumes of 4% agarose was the mixed with equal volumes of fibroblast suspension carefully in order to prevent bubbles. This created a final cell-gel solution of 10 x 106 cells/ml in 2% (w/v) agarose. Sterile glass plates with a 3mm spacer were used to cast this cell-gel solution. After 20 minutes of gellation, 8-mm

diameter specimens were cored and transferred to multi-well plates. 5ml of media was added to each well. This fibroblast-fibroblast-agarose tissue engineered constructs (TECs) were then cultured for an additional two days under free swelling conditions in a standard incubator (Steri-Cult Model 3033, Forma Scientific) with media replaced daily. The resulting TECs were henceforth classified as Day 0 constructs.

Compressive Stimulation

Before stimulation, the S constructs were first transferred from the multi-well plates to 35mm diameter glass bottom dishes (Matek Corp) and fed with 2ml of media. The dishes were covered with Teflon FEP film (100Å; DuPont, Circleville, OH). The film is held in place by a custom delrin ring [2, 162]. This film maintained the sterility of constructs and also prevented media evaporation. These dishes were then loaded into our custom bioreactor and subjected to compressive stimulation whose profile is described in the experimental design section [2, 162]. The NS constructs were also cultivated in the same bioreactor but without the compressive stimulation. Feeding media to these constructs was replaced daily.

Fluorescence Analysis

GFP-T and ECFP fluorescence were analyzed qualitatively and quantitatively in the TECs. Each S and NS constructs was washed in PBS to remove the media and visualized in filter sets for GFP-T(XF104-2, Omega) and ECFP(XF1142, Omega) in a fluorescence microscope (Zeiss Axiovert) with a filter set for GFP-T fluorescence and a filter set for ECFP fluorescence [2, 163]. Each TEC was further visualized using a filter set for red fluorescence (11002VZ, Chroma) to rule out autofluorescence. Constructs were homogenized as previously described to extract cells [163]. These homogenates were then pipetted into a black bottom microplate (100µl per well in three wells). RFUs in these were quantified in a spectrophotometer as previously described [2, 163].

Determination of Cell Viability

MTT assay (Vybrant MTT Cell Profileration Assay Kit, Invitrogen, Carlsbad, CA), was performed to determine cell viability. The fibroblast-agarose TECs were homogenized using a mortar and pestle. [162]. The homogenate was incubated for 4 hours in an incubator in 500µl of 12mM MTT solution and then incubated for 18 hours in 5ml SDS-HCL to completely dissolve the formazan. The absorbance of formazan was measured at 570 nm in a microplate spectrophotometer (Spectra Max M2, Molecular Devices, Sunnyvale, CA) [162, 397]. The viability results were normalized to results for day 0 constructs.

Biomechanical evaluation of fibroblast-agarose TECs

Aggregate modulus of the TECs was determined using confined compression testing. At each time point the constructs were placed in cryovials and stored at - 80°C. On the day of testing tissue specimen was thawed in room temperature phosphate buffered saline (PBS) for 10 minutes. The details of the testing are described elsewhere [162]. Briefly a 6-mm diameter core was created from the center of each construct and subjected to a constant contact stress until equilibrium displacement was achieved. During testing, specimen height and applied load were continuously measured. Tissue aggregate modulus (HA) was calculated using the formula HA= (Fo/ π a2) / (u/h) where Fo is applied compressive force, a is specimen radius, h is specimen thickness and u is equilibrium displacement[162].

Statistical Analysis

Two way ANOVA with Tukey's HSD post hoc test were performed to determine differences among values for all response measures with significance set at p <0.05. SPSS software was used to perform these tests.

7.3. Results

Qualitatively, we observed homogeneously distributed and rounded cells in all constructs. GFP-T fluorescing cells were present in all constructs (Fig. 7.1) but no GFP-T



Fig. 7.1. Rounded GFP-T expressing fibroblasts in agarose.

fluorescence was detected using any of the other filter sets. However, no apparent differences could be seen in ECFP fluorescence between the S and S constructs at each of the time points. None of the TECs showed cells displaying ECFP expression.

Quantitative results showed that dynamic compressive strain significantly decreased GFP-T expression in the TECs at 28 days. Compared to time-matched NS controls, the treatment significantly decreased RFU values by 0.56 fold at 28 days (p<0.05; Fig. 7.2).



Fig. 7.2. Compressive stimulation decreases GFP-T expression by Relative Fluorescence Units (RFU) day 28. * Significant increase compared to non-stimulated control at same time point (p< 0.05). Data represented as mean + one standard deviation. N=6 for all groups. NS- Non-Stimulated, S-Stimulated.

Compressive stimulation increased aggregate modulus of the fibroblast-agarose constructs, but only after 28 days of stimulation. S constructs showed a 1.25 fold increase in aggregate modulus compared to NS controls at day 28 (p<0.05; Fig. 7.3).



Significant increase compared to non-stimulated control at same time point (p< 0.05). Data represented as mean + one standard deviation. N=6 for all groups.

No significant changes occurred in cell viability due to mechanical stimulation or for increasing time in culture (p>0.05; Fig. 7.4). Day 28 constructs showed 8% lower cell viability than day 0 constructs.

7.4. Discussion

The objective of this study was to examine the effects of compression on fibroblast gene expression and resulting biomechanics. In this study, murine tail tendon fibroblasts were cultured, expanded and seeded in 3-D agarose to determine the effect of dynamic compressive strains on Col1/Col2 gene expression and aggregate modulus. Agarose was used as the biomaterial because of its ability to be compressed. The specific loading profile for stimulation was selected because it was previously shown to improve Col2 gene expression, protein, sGAG and mechanical properties of constructs [162, 336-339].



(p>0.05). Cell viability exceeded 92% at all time periods in culture. Data represented as mean + one standard deviation. N=6 for all groups.

Our quantitative fluorescence results support our first hypothesis that compressive stimulation significantly decreases GFP-T fluorescence and hence Col1 mRNA expression but only at later time points. In the current study, tail tendon fibroblasts exposed to dynamic compression downregulated Col1 gene expression. This decrease in Col1 gene expression is consistent with the idea that cyclic tensile strain is the primary stimulus for Collagen type I synthesis [163, 167, 414-421], and with reports that compression downregulates Col1 gene mRNA expression in chondrocytes [422]. Our results are also similar to another study in which fibroblasts from human periodontal ligament were cyclically contracted on a flexible membrane [423]. The cells in that study synthesized less Collagen type I and fibronectin protein and downregulated Col1A1 mRNA compared to non-stretched controls. Similar results were seen when rat Achilles fibroblasts were seeded in pellets and then subjected to cyclic hydrostatic

pressure. The cells in that study showed significantly less Col1 mRNA compared to nonstretched controls[206].

The absence of ECFP expression in any of the TECs indicates that cells were not upregulating the Col2 gene, thus negating our second hypothesis. While a compressive stimulus has been shown to upregulate Col2 gene expression in other cells types [162, 203, 206, 336-339, 389-390, 399, 413, 424-425], in our current experiments none of the constructs showed ECFP-positive cells. Our results directly contradict other reports in the literature, suggesting differences among applied treatments or the cell-scaffolds receiving these compressive signals. For example, when one study subjected rat Achilles fibroblasts in pellets to cyclic hydrostatic pressure, investigators showed increased the expression of Col2 mRNA [206]. It is noteworthy, however, that the mechanical profile we delivered in the current study is identical to the one that resulted in upregulation of Col2 gene expression when applied to murine chondrocyte-agarose scaffolds[162]. Such differences suggest that either fibroblasts are either terminally differentiated and incapable of responding to compressive stimulation or that the signal must be applied for longer periods to elicit the response. It is also plausible that fibroblasts might require a different compressive signal than chondrocytes to upregulate Col2 gene expression. Another reason could be that our fluorescence method is not sensitive enough to detect small changes in Col2 expression and thus might require the use of PCR.

At the same time, our biomechanical results support our third hypothesis that compressive stimulation significantly increases aggregate modulus of constructs. Aggregate modulus of stimulated constructs was significantly higher than non-stimulated controls at day 28. While compression decreased the expression of Col1 gene expression and did not induce Col2 gene expression it is plausible that they also increased the synthesis of proteoglycans and GAGS in our system. Compression has been shown to increases the synthesis of proteoglycans and GAGS[426-427]. Such accumulation of GAGs could have lead to increases in aggregate modulus we observed at day 28.

Our study is not without limitations. 1) We homogenized agarose constructs to record fluorescence. Agarose chunks in the homogenates could have increased the observed RFU Values. We assumed that these increases were constant across groups. 2) The experimental design did not allow us to track the changes in proteoglycans and collagen in these constructs, which could explain the increases in aggregate modulus. In the future we intend to track the production of proteoglycans and collagen. 3) We did not observe any ECFP expression. It is plausible that the cells were exhibiting Col2 expression, but in such low amounts that it could not be detected by our fluorescence methods. In the future we plan to use qRT-PCR techniques along with our fluorescence methods.

We conclude that although compressive stimulation of fibroblast-agarose constructs decreases Col1 gene expression, this stimulus was not sufficient enough or long enough to induce Col2 gene expression. However, the compressive stimulus was substantial enough to increase the aggregate modulus of constructs. In the future we plan to track protein and proteoglycan synthesis in fibroblast-agarose constructs as well as extend the experiment beyond 28 days to determine whether and when Col2 gene expression might be present. We also plan to examine different components of the compressive stimulus to evaluate their effects on Col1/Col2 gene expression in these cells.

Chapter 8

Tensile Stimulation Of Murine Stem Cell-Collagen Sponge Constructs Increases Collagen Type I Gene Expression And Linear Stiffness.

The objectives of this study were to determine how tensile stimulation delivered up to 14 days in culture influenced type I collagen gene expression in stem cells cultured in collagen sponges and to establish if gene expression, measured using a fluorescence method, correlates with an established method, real time quantitative RT-PCR (gRT-PCR). Using a novel model system, mesenchymal stem cells (MSCs) were harvested from six double transgenic mice in which the type I and type II collagen promoters were linked to green fluorescent protein-topaz (GFP-T) and enhanced cyan fluorescent protein (ECFP), respectively. Tissue engineered constructs were created by seeding 0.5 x 106 MSCs onto type I collagen sponge scaffolds in a silicone dish. Constructs were then transferred to a custom pneumatic mechanical stimulation system housed in a standard incubator and stimulated for 5 hours/day in tension for either 7 or 14 days using a repeated profile (2.4% peak strain for 20 seconds at 1 Hz followed by a rest period at 0% strain for 100 seconds). Control specimens were exposed to identical culture conditions but without mechanical stimulation. At three time points (0, 7 and 14 days) constructs were then prepared for evaluation of gene expression using fluorescence analysis and qRT-PCR, and the remaining constructs were failed in tension. Both analytical methods showed that constructs stimulated for 7 and 14 days showed significantly higher collagen type I gene expression compared to non-stimulated controls at the same time interval. Gene expression measured using qRT-PCR and fluorescence analysis were positively correlated (r = 0.9). Linear stiffness of stimulated constructs was significantly higher at both 7 and 14 days compared to non-stimulated controls at the same time intervals. Linear stiffness of the stimulated constructs at day 14 were significantly different from that of day 7. Future studies will vary the mechanical

signal to optimize type I collagen gene expression to improve construct biomechanics and invivo tendon repair.

8.1. Introduction

Injuries to soft tissues (tendon, ligament and meniscus) represent almost half of the 33 million musculoskeletal injuries occurring in the US each year [428-433] and often lead to surgery [434-436]. Inadequate healing of these injuries places patients at risk to dysfunction and disability. Tissue engineering [244] is an appealing conceptual alternative when conventional repair techniques (autografts, allografts, xenografts, and prostheses) [177, 281-289]prove unsatisfactory.

Tissue engineered constructs made by seeding mesenchymal stem cells (MSCs) in scaffolds are being used to repair soft tissue defects [51, 148, 150, 248, 290, 434], but these are usually vulnerable early after surgery because they lack the stiffness and strength of native tissue structures [248, 434]. To address this problem, investigators have been applying principles of Functional Tissue Engineering (FTE) [159-160] to use recorded in-vivo tissue forces [157-158] as design parameters for new generations of reparative tissue constructs. Some in FTE [159-160] have also delivered aspects of these in-vivo mechanical signals to precondition constructs while still in culture. Such preconditioning improves the material properties of constructs for soft tissue engineering by increasing the synthesis of extracellular matrix proteins such as collagen [172, 324, 417, 437].

Unfortunately, any mechanical and biological benefits of mechanical stimulation are usually not assessed until the end of mechanical stimulation in culture [161, 248, 438]. Typically, four or more weeks may be required before a destructive test is performed to judge whether a stimulus up-regulates gene expression or increases protein accumulation. This delay hampers the ability of the investigator to quickly optimize the stimulus in culture. A strategy that permits investigators to monitor near real-time gene expression throughout the tissue engineering process could allow them to either modify the stimulus or terminate an experiment leading to an

undesirable outcome. Unfortunately, no such method currently exists for tissue engineers to rapidly assess how mechanical (chemical, etc.) stimuli affect near real-time gene expression.

To address this need, we bred double transgenic mice having type I and type II collagen promoters linked to Green Fluorescent Protein-Topaz (GFP-T) and Enhanced Cyan Fluorescent Proteins (ECFP), respectively. These intra-cellular proteins are expressed when the type I and type II collagen genes are activated, respectively. Such fluorescent proteins have recently served as visual reporters for transgene activity and can be viewed in real time in living tissues [174-178]Investigators have used these promoter-fluorescent protein reporters: 1) to examine early embryonic development, 2) to conduct cell culture studies, and 3) to identify cells within a defined lineage in primary cell culture [174-178] Another value of this technology, when applied to tissue engineering, could be the ability to track how mechanical stimulation affects near realtime collagen gene expression during maturation of constructs in culture.

The first objective of this study was to determine how a controlled mechanical stimulus applied to a stem cell-collagen sponge construct in culture influences the expression of the type I collagen gene as well as linear stiffness. The second objective was to establish if gene expression, measured using a fluorescence method, correlates with an established method, real time quantitative RT-PCR (qRT-PCR). We hypothesized that: 1) mechanical stimulation would increase type I collagen gene expression as well as linear stiffness and 2) the fluorescence data and qRT- PCR data would be positively correlated.

8.2. Materials and Methods

Experimental Design

Bone marrow was harvested from both the femur and tibia of six (6) six-week to eightweek old double transgenic mice as previously described [395, 439]. After isolating and expanding the MSCs to second or third passage using a previously-published protocol[439], fifteen (15) constructs were created from each animal by seeding the cells at a concentration of 0.5×10^6 cells/construct in a type I collagen sponge (P1076, Kensey Nash Corp., Exton, PA).

Hence a total of 90 samples were created using cells from six (6) animals. For each animal, the resulting fifteen (15) constructs were assigned to five (5) treatment groups- Day 0 Non-Stimulated, Day 7 Stimulated, Day 7 Non-Stimulated, Day 14 Stimulated and Day 14 Non-Stimulated (Table 8.1). Thus, three (3) constructs were available per animal from each treatment group. One construct was assigned to evaluate GFP-T fluorescence in a spectrophotometer (measured in Relative Fluorescence Units (RFU)) [329], one construct was assigned to evaluate changes in type I collagen and glyceraldehyde-3-phosphate dehydrogenase (GAPDH) gene expression using real-time quantitative reverse transcriptase polymerase chain reaction (gRT-PCR) [167], and the remaining construct was failed in tension to determine its linear stiffness [171]. Stimulated (S) constructs were stretched between 0% and 2.4% peak strain [158] at 1 Hz for 20 seconds followed by a 100-second rest period. This pattern was repeated for five (5) hours/day. Peak amplitude, frequency and duty cycle were chosen based on studies performed in our lab on the effects of tensile stimulation on rabbit MSCs seeded in collagen scaffolds [167, 171]. The non-stimulated constructs (NS) served as controls. The sample size of six (6) animals was sufficient to detect a 30% treatment effect with an 80% power. Differences were considered significant for p < 0.05.

Double transgenic mice

The mouse containing the transgene pOBCol3.6GFPtpz was acquired courtesy of David Rowe, University of Connecticut Health Center. This transgene contains a 3.6 Kb fragment of the rat Col1a1 promoter, enhancer sequence and GFP-T. GFP-T expression in these transgenic mice is evident in skin, tendon and osseous tissues[176].

Plasmid pCol2-ECFP was derived by replacing the β-gal gene with ECFP (Clontech, Palo Alto, CA), in the expression gene containing the mouse type II collagen promoter and enhancer [179] (provided by W. Horton, Northeastern Ohio College of Medicine). This pCol2-ECFP construct was injected into mice blastocysts. Founder mice showed high levels of ECFP expression in cartilaginous tissue.

Treatment Conditions	Constructs/ Treatment Condition/Animal	Response Measures (Assignment Of Constructs)
Day 0 Non- Stimulated	3	Gene expression by qRT-PCR (n=1) Gene expression by RFU (n=1) Linear Stiffness (n=1)
Day 7 Stimulated	3	Gene expression by qRT-PCR (n=1) Gene expression by RFU (n=1) Linear Stiffness (n=1)
Day 7 Non- Stimulated	3	Gene expression by qRT-PCR (n=1) Gene expression by RFU (n=1) Linear Stiffness (n=1)
Day 14 Stimulated	3	Gene expression by qRT-PCR (n=1) Gene expression by RFU (n=1) Linear Stiffness (n=1)
Day 14 Non- Stimulated	3	Gene expression by qRT-PCR (n=1) Gene expression by RFU (n=1) Linear Stiffness (n=1)

Table 8.1. Treatment conditions, response measures, and assignment of constructs.

Mice transgenic for either pOBCol3.6GFPtpz or pCol2-ECFP were then bred to produce double transgenic animals. No apparent phenotypic differences were observed between the double and non-transgenic mice. Both GFP-T and ECFP have a 24- hour half life which indicates that fluorescence decays to half of its original by 24 hours.

Mesenchymal stem cells

Mice were euthanized by CO_2 according to Institutional Animal Care and Use Committee (IACUC) protocols. Their long bones were excised, transected, and placed in adapted centrifuge tubes [395, 439]. Marrow was extracted by centrifugation for one minute at 400g. Extracted cells were plated at 7.5 x 10^6 cells/100 mm dishes and fed media with supplements (MesenCultTM Basal Medium for Mouse Mesenchymal Stem Cells and Mouse Mesenchymal Stem Cell Stimulatory Supplements, Stem Cell Technologies, Vancouver, BC). Cells were allowed to attach for two days, washed with phosphate buffered saline (PBS, Gibco BRL/Life Technologies Inc., Gaithersburg, MD) to remove non-adherent cells, and fed fresh media. Adherent cells were allowed to grow for seven to ten days before first passage. Cells were then trypsinized and replated at 1 x 10^6 cells/100mm dishes and cultured for one additional week.

Cells at P2 and P3 were then sub-cultured at a density of 1 x 10^6 cells/100 mm dish and cultured for another week.

Scaffold and construct preparation

Sterilized collagen type I sponges (94% pore volume; 62 µm mean pore diameter) were provided by Kensey Nash Corporation (Exton, PA). Scaffolds were cut from these sponges such that they fit in the wells of a silicone dish [171]. Before seeding with cells, each scaffold was soaked overnight in PBS and then placed in each well of the dish. MSCs were suspended in media at a concentration of 2x10⁶ cells/ml. 250µl of this cell suspension was pipetted on top of each scaffold. All constructs were placed in an incubator (Steri-Cult Model 3033, Forma Scientific, Marietta, OH; 37°C, 5% CO₂, 95% relative humidity) for 2 weeks and fed three times weekly with Advanced DMEM, 1% antibiotic/antimycotic, 1% glutamax and 10% FBS.

Mechanical stimulation

After 2 days of incubation, the silicone dishes containing the constructs to be stimulated were placed into a computer controlled five station pneumatic mechanical stimulation system housed within an incubator (Steri-Cult Model 3033, Forma Scientific, Marietta, OH) [171] and stretched using the pattern described in Experimental Design. Dishes were removed from the incubator after either 7 or 14 days and constructs were then prepared for either evaluation of gene expression using fluorescence analysis and qRT-PCR or failure testing in tension.

Fluorescence Microscopy

Each construct was washed in PBS for 1 hour with gentle shaking to remove media and then visualized for GFP-T fluorescence in a fluorescence microscope (Axiovert 25, Carl Zeiss Inc., Göttingen, Germany) equipped with filter sets for visualizing GFP-T and ECFP (XF104-2 and XF114-2, respectively; Omega). To rule out cell auto-fluorescence, each construct was also visualized for rhodamine using a specific filter set (11002VZ, Chroma).

Spectrophotometric Analysis

After imaging, constructs were digested for 40 minutes in 4 ml of 100U/ml type I collagenase (Sigma Chemical, St. Louis, MO) in an incubator with gentle shaking. The resulting collagen fragments were further digested for 20 minutes in 2 ml of trypsin (Invitrogen-Gibco BRL/Life Technologies Inc., Gaithersburg, MD). Digests were then centrifuged at 2000 rpm for 6 minutes and the supernatant was discarded. The remaining pellets were re-suspended in 1 ml of PBS and filtered using a 100µm nylon mesh (BD Falcon, Bedford, MA). These digests were then pipetted into a black bottom microplate (200µl per well in three wells). GFP-T fluorescence in these digests was quantified in Relative Fluorescence Units (RFUs) [329] by reading the microplate in a spectrophotometer (Spectra Max M2, Molecular Devices) using an excitation wavelength of 491 nm and an emission wavelength of 529 nm, with a cut off filter of 519 nm. Pilot studies on GFP-T expressing fibroblasts harvested from the ribs of new born mice did not show any qualitative differences in GFP-T fluorescence when subjected to the above mentioned collagenase, trypsin digestion and filtration steps.

RNA extraction and conventional gene expression analysis

RNA extraction and conventional gene expression analysis were performed according to previously published protocols [167]. Briefly, the constructs were stored in RNAlater (QIAGEN Inc., Valencia, CA) for 2 days at -4°C. The RNAlater was aspirated carefully and the constructs were frozen in liquid nitrogen to prevent RNA degradation. RNA from each construct was extracted using a RNeasy mini kit (QIAGEN Inc., Valencia, CA). Conventional reverse transcriptase (RT) reaction (MuLV reverse transcriptase, Applied Biosystems, Foster City, CA) was performed [167] to create first-strand complementary deoxyribonucleic acid (cDNA). Mouse specific primers were used for type I collagen and glyceraldehyde-3-phosphate dehydrogenase (GAPDH) gene expression. The forward and reverse primer sequences and the resultant products are summarized in Table 8.2. Before use in the experiment, all primers were tested under conventional and real-time quantitative RT-PCR conditions to ensure specificity with only
one band by electrophoresis. The conventional PCR of the reverse transcribed RNA was performed according to previously published protocols [167]. The amplified products were verified by 2% agarose gel electrophoresis in Tris-Acetate-EDTA (TAE) and SYBR safe DNA gel stain (Invitrogen Molecular Probes, Eugene, OR).

Gene	Primer sequence	Product size (bp)	Annealing temperature
Collagen I	TGT GTG CGA TGA CGT GCA AT	132	58°C
	GGG TCC CTC GAC TCC TAC A		
GAPDH	AAT GGT GAA GGT CGG TGT G	200	55°C
	CCT TCG GGT AGT GGT AGA AG		

Table 8.2. Sequence of primers used for gene expression analysis and product size in base pairs

Real-time quantitative RT-PCR

Real-time quantitative RT-PCR (qRT-PCR) was performed to quantify mRNA levels of the genes according to previously published protocols [167]. qRT-PCR was perfomed by monitoring SYBR Green fluorescent dye (SYBR Green PCR master mix, Applied Biosystems, Foster City, CA) bound to double-strand DNA with a continuous fluorescence detector (DNA Engine Opticon 2 System, MJ Research Incorporated, Waltham, MA). All samples were run in duplicate since differences in cycle number values between samples were less than 0.3 cycles. Standard curves were created for each target gene to quantify gene expression for each cellsponge construct. The absolute amount of the corresponding gene mRNA in each construct was obtained from the corresponding gene standard curve. Gene expression was normalized by calculating the ratio between type I collagen and GAPDH genes for each sample.

Biomechanical evaluation of the constructs

After 0, 7 or 14 days in culture, constructs for biomechanical analysis were placed in cryovials and frozen at -80^oC. On the day of testing, constructs were removed from the freezer

and thawed to room temperature. Small squares of gauze were used to cover the post holes in each end of the constructs. These gauzes provided a surface that would minimize slippage and the premature failure of the specimens in the grips. Constructs were inserted into custom grips in a materials testing system (100R6, TESTRESOURCES, Shakopee, Minnesota) and fixed at a gauge length of 12 mm to minimize Saint Venant's gripping effects on specimen properties[440]. The specimens were then failed under displacement control at a rate of 10%/sec in a PBS bath at room temperature [51]. Linear stiffness was calculated from the linear region of the force-elongation curve generated by the constructs during failure testing.

Statistical analysis

The duplicate gene expression measures and triplicate RFU measures for each cell line from each of the six mice were averaged prior to statistical analysis. Gene expression and mechanical properties of the stimulated vs. non-stimulated stem cell-collagen sponge constructs were compared using a mixed-effects model (SAS proc mixed) with culture time and stimulation as fixed factors and animal as a random factor [334]. All conclusions regarding the significance of mechanical stimulation on gene expression and mechanical properties were made for p<0.05.

8.3. Results

Cells did not fluoresce in any of the constructs at day 0. However, GFP-T fluorescent cells were found in all 7 and 14 day constructs. No fluorescence signal was evident when using the ECFP and rhodamine filter sets, indicating the specificity of the GFP-T signal and the lack of detectable levels of type II collagen gene expression, even in the end regions where the construct attaches to the posts. The fluorescing cells appeared elongated in all constructs at both time periods of stimulation (Figs. 8.1A -8.1D). Elongated cells in both the controls (Figs. 8.1A & 8.1C) and the stimulated constructs (Figs. 8.1B & 8.1D) were randomly oriented with respect to the direction of applied tensile strain.

Tensile stimulation significantly increased measured RFU values at both time points. At day 7, RFU values in the S constructs increased by 2.2 fold compared to those for the NS



Fig. 8.1. Elongated GFP-T fluorescent cells in A) non-stimulated constructs and B) stimulated constructs at day 7 as well as C) non-stimulated constructs and D) stimulated constructs at day 14. Original magnification x10. Greater fluorescence was observed in the stimulated vs. non-stimulated control constructs at 7 days.

constructs (p=0.0002) (Fig. 8.2). At day 14, RFU values in the S constructs increased by 1.5 fold compared to those for the NS constructs (p=0.01) (Fig. 8.2). RFU values for the S vs. NS construct groups averaged 6.3 ± 0.22 vs. 2.8 ± 0.69 at day 7 and 4.63 ± 0.16 vs. 3.04 ± 0.97 at day 14 (mean \pm SD). No difference was seen between NS constructs at day 7 and 14 (p>0.5) (Fig. 8.2). There was a significant decrease in RFUs in S constructs between day 7 and 14 (p=0.0001) (Fig. 8.2).



at same time interval (p< 0.02). **Significantly different from day 0 (p<0.00001) ***Significantly different from day 7 stimulated constructs (p<0.0002). Data represented as mean \pm standard deviations. N=6 for all groups.

Seven and fourteen days of in-vitro mechanical stimulation significantly increased type I collagen gene expression in stem cell-collagen sponge constructs measured using qRT-PCR. S constructs showed a 12-fold increase in type I collagen gene expression (p = 0.001) relative to NS controls at day 7 and a 5-fold increase at day 14 (p = 0.01) (Fig. 8.3). Type I collagen gene expression as measured by qRT-PCR for the S vs. NS construct groups averaged 7.8 ± 2.6 vs. 0.6 ± 0.5 at day 7 and 2.8 ± 1.1 vs. 0.5 ± 0.4 at day 14 (mean ± SD). No difference was seen between NS constructs at day 7 and 14 (p>0.5) (Fig. 8.3) but there was a significant decrease in type I collagen gene expression in the S constructs between day 7 and 14 (p=0.005) (Fig. 8.3).



Fig. 8.3. Tensile stimulation increased type I collagen gene expression by qRT-PCR. *Significantly different from non-stimulated controls at same time interval (p< 0.02). **Significantly different from day 0 (p<0.0003) ***Significantly different from day 7 stimulated constructs (p<0.006). Data represented as means ± standard deviations. N=6 for all groups.

Mechanical stimulation did not significantly increase GAPDH gene expression (p>0.5) at day 7 or day 14 in culture. GAPDH values for the S vs. NS construct groups averaged 5.8 E-07 \pm 1.7 E-07 vs. 5.6 E-07 \pm 1.3 E-07 at day 7 and 5.4 E-07 \pm 1.4 E-07 vs. 5.3 E-07 \pm 1.2 E-07 (mean \pm SD). The GAPDH values for day 0 constructs were 5.8 E-07 \pm 1.7 E-07. The results from qRT-PCR and fluorescence analysis were positively correlated (Fig. 8.4, r² = 0.79). The slope of the linear regression curve was 0.56.

Both 7 and 14 days of tensile stimulation increased the construct's linear stiffness compared to NS constructs but had no effect on dimensions (p>0.05; Table 8.3). S constructs showed a 2.3-fold increase in linear stiffness (p = 0.0006) relative to NS controls at day 7 and a 1.6-fold increase at day 14 (p = 0.003) (Fig. 8.5). Linear stiffness for the S vs. NS construct groups averaged 0.035 \pm 0.006 N/mm vs. 0.015 \pm 0.003 N/mm at day 7 and 0.029 \pm 0.004 N/mm vs. 0.0175 \pm 0.4 N/mm at day 14 (mean \pm SD). No difference was seen between NS

constructs at day 7 and day 14 (p=0.1) (Fig. 8.5). Tensile stiffness of both 7 and 14 day constructs were significantly less than the corresponding stiffness values of day 0 constructs (p=0.005) (Fig. 8.5).



Fig. 8.4. Type I collagen gene expression as measured by relative fluorescence units (RFU) and by qRT-PCR were positively correlated ($r^2 = 0.79$). Each point represents the average of 6 samples (means ± standard deviations). Circle denotes Day 0 constructs, triangles denote non-stimulated constructs and squares denote stimulated constructs.

Evaluation time points	Length(mm)	Width (mm)	Thickness (mm)
Day 0	23.2 ±0.5	8.5 ±0.7	2.7±0.1
Day 7 Stimulated	23.8 ±0.4	8.3 ±0.5	2.6 ±0.4
Day 7 Non-Stimulated	23.1 ±0.6	8.1 ±0.5	2.5 ±0.5
Day 14 Stimulated	23.6. ±0.3	8.0 ±0.6	2.5 ±0.5
Day 14 Non-Stimulated	23.2 ±0.4	8.1 ±0.9	2.4 ±0.5

Table 8.3. Tensile stimulation did not significantly affect construct dimensions (p>0.05). Shown are means + standard deviations for all three parameters.



Fig. 8.5. Tensile stimulation increased linear stiffness. *Significantly different from non-stimulated controls at same time interval (p<0.004). **Significantly different from day 0(p<0.0005). ***Significantly different from day 7 stimulated (p<0.05). Data represented as means ± standard deviations. N=6 for all groups.

8.4. Discussion

Understanding how to control matrix production and assembly during culture is one of the primary objectives in developing functional tissue engineered replacements for load-bearing connective tissues [159]. Although externally-applied deformations have been shown to strengthen tissue engineered constructs [171-172, 324, 417, 437], long culture times are typically required for cells to synthesize enough matrix to improve resulting mechanical properties that can remotely approach the levels of the native tissue. This slow process could reflect the fact that mechanical stimuli have yet to be optimized for inducing such cellular-based benefits. In particular, we still know little about how various components of the stimulation profile (e.g. strain amplitude, frequency, and duty cycle) affect construct quality and functionality in culture and after surgery. We do know, however, that determining the importance of each component can be a very expensive and time-consuming process. Therefore, a strategy to

evaluate how these factors influence the production of extracellular matrix components like type I (tendon and ligament mid-substance) or type II collagen (soft tissue insertions, articular cartilage, meniscus, intervertebral disc) would appear to be a very beneficial step in this development effort. It was for these reasons that we bred the double transgenic mouse model for the current study. By using this fluorescence technology we could repeatedly track changes in gene expression during rest periods between bouts of mechanical stimulation in culture rather than use destructive microarray or qRT-PCR techniques that could only detect final changes after stimulation. Also this technology provides a unique way to detect near real-time changes in expression in culture by one or both of these genes and to readjust the mechanical signal applied to the evolving construct to obtain the desired spatial and temporal gene expression patterns.

This study utilizing MSCs harvested from these mice was designed to evaluate the effects of effects of uniaxial tension on gene expression and biomechanics in cell-collagen constructs. We hypothesized that: 1) mechanical stimulation would increase type I collagen gene expression as well as linear stiffness and 2) the fluorescence data and qRT- PCR data would be positively correlated. In our study we found that tensile stimulation increased GFP-T fluorescence and type I collagen gene expression compared to controls, thus validating the first part our hypothesis. The increases in both measures (Figs. 8.2 & 8.3) that we observed in stimulated constructs reinforce the importance of stimulating a cell-scaffold construct with a defined mechanical signal in a cell culture system. The fact that the tensile stimulus chosen in this study significantly increased GFP-T fluorescence in the construct's cells without affecting ECFP activity suggests that tensile stimulation did not affect collagen type II gene expression.

Several reasons could explain the increases in GFP-T fluorescence (type I collagen gene expression) we observed after seven and fourteen days of tensile stimulation. Tensile strains are known to trigger the creation of cell-surface stretch receptors and integrins, activating a cascade of genes responsible for the synthesis and secretion of ECM components

[441]. Of course the stimulus we applied may have also produced other cellular changes (cell proliferation, mass transfer rates of nutrients, metabolites and waste materials [173], expression and synthesis of various growth factors and cytokines), some of which are known to be sensitive to mechanical stimulation [396, 398, 442-443]. Several of these factors will be examined in greater detail now that we have demonstrated a significant positive effect of mechanical stimulation on gene expression in this model system.

Our findings are generally consistent with previous reports examining how tensile stimulation affects gene expression of cells and cell-based constructs, although the specific treatments, conditions and response measures vary greatly among studies. Investigators using human fibroblasts [417-418, 444] as well as human mesenchymal[290, 420] and human and bovine stromal cells [173] have demonstrated positive effects of mechanical stimulation on type I collagen gene expression and protein synthesis. However, these studies were conducted either using cells on monolayer [420] or after placing cells in three-dimensional polyurethanes[417] and collagen gels [173, 290]. These differences in local matrix environment can be important to driving cell phenotype and gene expression patterns. The 2.4 % peak strain that we delivered in the current study is far less than the 5% and 10% [417-418, 420, 444] peak strain treatments used by other groups and not as complex as reported biaxial strain patterns[173] While the studies by Park et.al and Noth et.al [290, 420] matched the frequency (1 Hz) delivered in our study and the 14 days for treatment assessment [173, 290] was identical to our time interval, other aspects of these studies differed markedly. In fact, direct comparison of our results to those used by other investigators was difficult given the broad range of frequencies (0.0167 Hz[173], 0.25 Hz[418], and 0.167 Hz [444]) and time assessment periods (from 6 to 24 hrs) [417-418, 420, 444] reported in the literature. Studies will ultimately need to either limit these stimulation conditions across studies or examine the interactive effects of these treatment components if we are to understand how to control and optimize gene and protein expression in tissue engineering.

Our findings are also different from a previous study in the literature [173]. Although Altman and co-workers[173] examined mechanically-induced changes in gene expression using stem cells over the same time interval used in the current study, they used collagen gels rather than sponges and found that tensile stimulation significantly increased type I collagen gene expression at day 14 but not at day 7. Such differences between our two outcomes could be due to different cell sources (bovine vs. murine) or mechanical strain patterns (combined axial and torsional strains vs. uniaxial tensile strains), although other differences may have also contributed. For example, a cell's biosynthetic activity is strongly related to its ability to attach to its collagen scaffold [445-450]. Once accomplished, subsequent collagen synthesis within these three-dimensional collagen lattices is then regulated by transcriptional and post-transcriptional mechanisms mediated by α and β integrins [448]. Similarly, synthetic scaffolds can also provide a conducive environment, as shown by Chastain and co-workers [451] who found increases in $\alpha 6$ and $\beta 6$ gene expression over a 5-week culture period using mesenchymal stem cells deposited in PLGA scaffolds. Regardless of scaffold type, increases in type I collagen gene expression will clearly require the appropriate signals and microenvironment for effective cellbased therapy.

We also found in this study that collagen type I gene expression measured using real time quantitative RT-PCR and fluorescence analysis were positively correlated (Fig. 8.4) thus validating our second hypothesis. This correlation serves as a validation for using fluorescence as an indicator of average gene expression within tissue engineered constructs. Additional correlations of this type may soon allow investigators to use fluorescence as a reasonable estimate of expression of genes like type II collagen than using more traditional approaches like qRT-PCR. These correlations may also be possible at even finer levels in the tissue engineered constructs, including expression by groups of cells or even individual cells in the future.

We also found that applying both seven (7) and fourteen (14) days of tensile stimulation increased the construct's linear stiffness compared to non-stimulated constructs in culture (Fig.

8.5) validating the last part of our first hypothesis. Other investigators have noted positive effects of stimulating collagen-based constructs on biomechanics as well as gene and protein expression using different cell types (tendon fibroblasts [398], smooth muscle cells [396], ligament fibroblasts [444], and MSCs [173, 420] and methods of stimulation [173, 396, 398, 418, 420, 444]. Our results are also similar to another study in our laboratory where mechanical stimulation of rabbit mesenchymal stem cells in collagen sponge constructs not only increased construct stiffness but also produced increases in type I and III collagen mRNA expression using qRT-PCR [167].

Day 7 and day 14 constructs showed significantly lower stiffness compared to day 0 constructs (Fig. 8.5). It has been shown that when chondrocytes [452] and fibroblasts [453] are seeded in collagen based constructs, they produce MMPs which degrade the collagen scaffold. It is plausible that in our study the mouse MSCs also produced MMPs which degraded the collagen scaffold and in turn, decreased its stiffness. Such decreases in stiffness are likely offset by tensile stimulation. In the future, we plan to also monitor MMP production in our stimulated and control constructs.

The stiffness of our constructs is still orders of magnitude less than that of that of native tendons[171] and hence cannot currently replace the entire ruptured tendon. Instead, we have been using these constructs for "biological augmentation" in a load-protected regime to repair central defect injuries to the rabbit patellar tendon. Implanting such stimulated cell-scaffolds invivo can speed up the repair process as the cells are already acclimatized in-vitro to some of the deformations that they might experience in-vivo. Further these cells are already synthesizing type I collagen in-vitro and might fill the repair site much faster. We have shown that such compliant constructs produce rabbit patellar tendon repairs whose average stiffness is three orders of magnitude greater than the stiffness of initial constructs at surgery and capable of matching the stiffness of normal patellar tendon up to 150% of peak in-vivo force values [171].

The results obtained from this study have the potential to be translated to higher species such as humans. If we can demonstrate through in-vitro studies that murine cell-based constructs respond to mechanical stimuli (and chemical stimuli) in ways similar to rabbit cell-based constructs by a range of response measures, it is then our hope to be able to apply the results of genetically-based experiments in the mouse to new therapies in larger animals where such tools do not yet exist. The benefits that we formulate and test in lower species and then validate in larger preclinical models can then be translated to improved repair in humans.

Our study has limitations. 1) In this study we recorded non-destructively the average effects of uniaxial tension on GFP-T expression in MSCs. We are currently examining techniques to measure fluorescence without destroying the constructs. 2) We did not quantify the number of fluorescing cells. This prevented us from distinguishing whether the increases in fluorescence were due to increased numbers of fluorescing cells and/or increased production of GFP-T by already fluorescing cells. Determining which of these effects predominates will require more local recordings of gene expression at the individual cell level combined with in-situ hybridization. Fluorescing cells could also be counted in a non-destructive way by applying statistical methods if we had the optical systems that could scan through the entire depth of the opaque constructs. 3) We selected each fluorescent protein to have a 24-hour half life, thus providing near, rather than actual, real-time gene expression. While it would have been ideal to have instantaneous feedback of gene up-regulation much like a conventional transducer, the longer period ensured that we could detect measurable changes in fluorescence before signal dissipation. 4) Cell viability within the constructs was not determined. Diminished cell viability in individual constructs could have adversely affected our RFU results for GFP-T fluorescence and thus increased inter-specimen variability. 5) This method of fluorescence analysis is limited to the specific transgenic species described with two modified fluorescent proteins. 6) We chose to measure only the construct's linear stiffness because we were concerned that the lower aspect ratio of 2:1 (less than the ideal ratio of 3:1 to avoid St. Venant effects) might alter the structure's

failure properties. 7) These constructs can be only used to repair tendons in a load-protected regime. We are currently investigating ways to further improve the stiffness of our constructs.

In conclusion, this study has demonstrated that short-term tensile stimulation increases type I collagen gene expression and linear stiffness of murine stem cell-collagen sponge constructs and that two measures used to track changes in type I collagen gene expression are correlated. Future studies will focus on optimizing the effects of these in-vitro signal components on expression of the type I collagen gene and the degree to which they affect construct and then repair biomechanics. Given that construct stiffness is positively correlated with repair stiffness after implanting rabbit autologous MSC-collagen constructs in patellar tendon defects [171] our efforts to optimize stiffness in culture might also positively impact tendon repair biomechanics and gene expression across multiple species and injury scenarios. We also envision that once optical technologies have been developed to visualize fluorescence in opaque constructs in a bioreactor at magnifications consistent with light microscopy, we and others should be capable of tracking real-time, local changes in gene fluorescence. Ultimately, we hope to speed tendon repair and match the tangent stiffness of normal tendons to levels well above peak in-vivo force levels [171].

Chapter 9

Tensile Strain Downregulates Col2 Gene Expression In Double Transgenic Murine Chondrocytes Seeded In Collagen Sponges

The objectives of this study were to determine how tensile displacements, applied to murine chondrocyte-collagen tissue engineered constructs (TECs), influence Col1 and Col2 gene expression as well as construct linear stiffness. We hypothesized that up to 28 days of tensile stimulation would downregulate Col2 expression, upregulate Col1 expression and increase TEC stiffness. Chondrocytes were harvested from the ribs of six newborn double transgenic mice carrying transgenes with enhanced cyan fluorescent protein (ECFP) and green fluorescent protein (GFP-T) as reporters for expression from the Col2a1 and Col1a1 promoters, respectively. Twenty-seven TECs were created per animal by seeding 0.5 x 10⁶ chondrocytes in type I collagen scaffolds. Twelve of these TECs were stimulated (S constructs) for 7, 14, 21 or 28 days in a pneumatic system (n = 3 each) housed in an incubator and another twelve matching constructs were not stimulated (NS controls) but exposed to identical culture conditions and time intervals. The remaining three TECs per animal served as Day 0 controls. ECFP-fluorescing cells were present in all constructs and time intervals but no GFP-T expressing cells were observed. Compared to time-matched NS controls, 21 and 28 days of stimulation significantly decreased ECFP fluorescence and 28 days of stimulation significantly decreased linear stiffness. This study reveals that tensile stimulation not only downregulates Col2 gene expression but fails to upregulate detectable Col1 gene expression within 4 weeks. The inability of these specialized cells to express the Col1 gene would be consistent with an unaltered TEC stiffness, but the observed reduction in stiffness suggests that Col2 expression patterns may also control tensile stiffness even without a perceived change in chondrocyte phenotype. Current studies in the lab are examining type I and type II collagen protein levels in

these TECs and whether longer time intervals and different components of tensile stimulation in culture might result in dedifferentiation of these chondrocytes and corresponding positive effects of Col1 expression on TEC biomechanical stiffness.

9.1. Introduction

Articular cartilage maintains homeostasis as it responds to normal internal stresses produced by dynamic chemical equilibrium and modified by complex external physical forces applied to joint surfaces [454-456]. Although articular cartilage primarily resists compressive external loading, these loads create transverse tensile and shear strains, particularly in the tissue's superficial layer. Efforts to model rapid loading on articular cartilage also reveal the presence of tensile forces acting at the cartilage-bone interfaces as well as in type II collagen fibers near the articular surface [457]. Larger forces associated with traumatic injury can induce cracks or fissures in the articular surface leading to degeneration and long-term arthritis [458-463]. Continued loading of these damaged regions can generate regions of high tensile stress near the crack tips, resulting in tensile strains on neighboring chondrocytes [464]. These altered stress environments can also lead to progressive tissue degradation in adjacent and otherwise normal articular cartilage.

Many forms of mechanical stimulation modulate chondrocyte matrix synthesis both in tissue explants and in 3-D culture environments [166, 366, 465-472]. Static and cyclic mechanical compression and cyclic shear can enhance as well as inhibit extracellular matrix (ECM) synthesis and gene expression by chondrocytes [366, 469-472]. However, relatively few studies have explored how uniaxial or biaxial tensile signals modulate chondrocyte phenotype or matrix synthesis [464, 473-477]. Cyclic tensile strains (10%-20% peak-to-peak) applied to chondrocyte-fibrin gel constructs inhibit proteoglycan synthesis whereas lower peak strains (5%) show no such effect [477]. In another study, tensile strains ranging between 0% and 10% increased cell proliferation but actually decreased matrix synthesis [464]. Tensile strain-induced matrix deformation can also regulate proliferation and differentiation of chondrocytes cultured in

3-D collagen scaffolds [473, 475]. Understanding the role of tension on chondrocyte metabolism is important because tissue engineering approaches to repairing damaged articular cartilage may involve inserting a scaffold with markedly different material properties than the surrounding tissue into a defect site. Under functional loading, regions of the repair or native tissues near this material property mismatch may experience substantial tensile strains[464]. This altered local mechanical environment could in turn affect the development of the repair tissue as well as the maintenance or the continued degradation of the native cartilage [464].

Tensile stimulation has been shown to upregulate collagen I gene expression in other cell types like fibroblasts and MSCs. Groups have shown that tensile stimulation applied to human fibroblasts [416-418], mesenchymal stem cells [419-420] and stromal cells [421] as well as bovine stromal cells [421] increases Col1 gene expression and type I collagen protein synthesis. Similarly, our group has demonstrated that tensile displacements applied to tissue engineered constructs containing either rabbit [167] or murine [163] MSCs upregulate Col1 gene expression. Whether such tensile signals might downregulate Col2 expression and upregulate Col1 expression in chondrocytes is intriguing especially if expression by both genes could be monitored in near real time.

Our laboratory has created novel transgenic technologies to study the effects of mechanical stimulation on gene expression by both undifferentiated and specialized cells. In collaboration with Cincinnati Children's Hospital, we created a double transgenic murine model in which the Col1 and Col2 genes are linked to Green Fluorescent Protein- Topaz (GFP-T) and Enhanced Cyan Fluorescent Protein (ECFP), respectively [2, 163]. Cells from these mice fluoresce either green or cyan if either Col1 gene or Col2 gene is activated respectively. We have shown that after suspending different cell types from these mice in scaffolds, MSCs show increased Col1 gene expression in response to tensile strains [163] and chondrocytes show

demonstrated that when these fluorescence patterns are quantified in a spectrophotometer, the resulting relative fluorescence units or RFUs are positively correlated with qRT-PCR [2, 163].

Using this double transgenic murine technology, the objective of the current study was to examine how cyclic tensile displacements influence tissue engineered constructs containing chondrocytes. We hypothesized that compared to non-stimulated constructs at equivalent time points, tensile stimulation would: 1) Decrease ECFP fluorescence, a reporter of Col II gene expression; 2) Increase GFP-T fluorescence, a measure of Col1 gene expression; 3) Decrease the construct's linear stiffness. 4) We also hypothesized that non-stimulated constructs would maintain ECFP fluorescence over equivalent time periods.

9.2. Materials and Methods

Experimental Design

Chondrocytes were harvested from the ribs of six (6) newborn double transgenic mice and passaged until P2 [2]. Type I collagen sponges (Kensey Nash Corp., Exton, PA) previously shown to increase Col1 gene expression in murine and rabbit MSCs [163, 167] was used as the scaffold. Chondrocyte-collagen constructs were created by seeding 0.5 x 10⁶ cells/construct. Twenty seven constructs were created per animal. Twelve constructs were assigned to stimulated (S) groups and another twelve constructs were assigned to non-stimulated (NS) controls at days 7, 14, 21 and 28. The remaining three (6) constructs served as day 0 NS controls. Stimulated (S) constructs were elongated between 0% and 2.4% peak strain at 1 Hz for 20 seconds followed by a 100-second rest period. This pattern was repeated for five (5) hours/day. Peak amplitude, frequency and duty cycle were chosen based on studies performed in our lab on the effects of tensile stimulation on murine MSCs [163]. The six (6) S and six (6) NS constructs at each time point were assigned as follows: one for ECFP/GFP-T fluorescence in a spectrophotometer (measured in Relative Fluorescence Units (RFU)) [2, 163, 446], one for cell viability using an MTT assay[478], and one for determining linear stiffness[163].

Chondrocyte isolation and expansion

Newborn mice were euthanized using protocols in accordance with protocols approved by Institutional Animal Care and Use Committee (IACUC) protocols. Chondrocytes were harvested from the rib cages and expanded using previously-published procedures [2, 333, 479]. Chondrocytes were passaged until P2. While ECFP expression was evident, no intracellular GFP-T expression was seen in cultured chondrocytes from P0 through P2.

Scaffold and construct creation

Scaffolds were cut from sponges provided by Kensey Nash Corporation (Exton, PA). The dimensions of the scaffolds are such that they can fit in the wells of silicone dishes. Holes were created in each scaffold to fit over the posts in each well of the dish [163, 171]. Before seeding with cells, each scaffold was soaked in 70% ethanol overnight and rinsed three times in PBS. Chondrocytes were suspended in media at the chosen concentration (2x10⁶ cells/ml). 250µl of this cell suspension was pipetted on top of each scaffold. All constructs were placed in an incubator (Steri-Cult Model 3033, Forma Scientific, Marietta, OH; 37°C, 5% CO2, 95% relative humidity) for two days to permit the cells to acclimate to the environment. Constructs were fed three times weekly with Advanced DMEM, 1% antibiotic/antimycotic, 1% glutamax and 10% FBS.

Mechanical stimulation

After letting the cells acclimatize in the collagens sponges for 2 days of incubation (I denoted as Day 0), only the silicone dishes containing the S constructs were stimulated using the pattern described in the Experimental Design. This stimulation was done in a pneumatic mechanical stimulation system housed in a standard incubator (Steri-Cult Model 3033, Forma Scientific, Marietta, OH) [163, 171]. The details of this pneumatic system can be obtained here [2]. Dishes were removed from the incubator at the prescribed intervals and constructs were

then prepared for evaluation of gene expression using fluorescence analysis, failure testing in tension, or cell viability.

Fluorescence Analysis

ECFP and GFP-T fluorescence in constructs were analyzed qualitatively and quantitatively. Each S or NS construct was visualized in a fluorescent microscope (Zeiss Axiovert) with a filter set (XF1142, Omega) for ECFP fluorescence and a filter set (XF104-2, Omega) for GFP-T fluorescence [2, 163]. Further visualization using filter set for red fluorescence (11002VZ, Chroma) ruled out autofluorescence. Constructs were digested as previously described to extract cells [163]. These digests were then pipetted into a black bottom microplate (200µl per well in three wells). ECFP/GFP-T RFUs were quantified in a spectrophotometer as previously described [2, 163].

Determination of Cell Viability

At each time point, cell viability in constructs was determined using an MTT assay (Vybrant MTT Cell Profileration Assay Kit, Invitrogen, Carlsbad, CA), following the manufacturer's instructions. Details of the assay are described elsewhere[478].

Biomechanical evaluation of the cell-collagen constructs

After the specific time periods in culture, constructs for biomechanical analysis were placed in cryovials and frozen at -80°C. On the day of testing, constructs were removed from the freezer and thawed to room temperature. TECs were inserted into custom grips in a materials testing system (100R6, TESTRESOURCES, Shakopee, Minnesota) and then failed under displacement control at a strain rate of 10%/sec [163]. Linear stiffness was calculated from the linear region of the force-elongation curve generated by the constructs during failure testing [163].

Statistical Analysis

The presence of treatment-related differences among values was determined for all response measures using a two-way ANOVA with Tukey's HSD post hoc testing (SPSS

software). Animal was chosen as a random factor, culture time and tensile stimulation as fixed factors, and response measures as dependent variables. Significance was set at p < 0.05.

9.3. Results

Fluorescence Microscopy

Fluorescence microscopy using ECFP and GFP-T filter sets revealed ECFP-expressing cells in all constructs (Fig 9.1) but no GFP-expressing cells in any construct. Fluorescence was observed in the red filter set, thus ruling out autofluorescence. ECFP fluorescent cells were found in all constructs at all time points of culture. However, no apparent differences in ECFP fluorescence were observed among non-stimulated and stimulated constructs at the three time points. ECFP expressing cells appeared elongated and were not aligned in the direction of the applied tension.

Spectroscopic Analysis

Quantitative results showed that applying a dynamic tensile displacement significantly decreased ECFP expression in the 3-D constructs, but only at later time points. Compared to time-matched, NS controls, compressive stimulation significantly decreased RFU values by 0.71 fold and 0.56 fold at, 21 and 28 days, respectively (p<0.05; Fig. 9.2).

Linear Stiffness

Stiffness of non-stimulated constructs appeared to increase over time but the increases were not significant (p>0.05; Fig. 9.3). Tensile stimulation significantly decreased the stiffness of constructs at 28 days in culture. S constructs showed a 0.79 fold decrease in linear stiffness compared to NS controls at day 28 (p<0.05; Fig. 9.3).

Cell Viability

No significant changes occurred in cell viability due to mechanical stimulation or for increasing time in culture (p>0.05; Fig. 9.4). Day 28 constructs showed 6% lower cell viability than day 0 constructs.





Fig. 9.1. Elongated ECFP expressing chondrocytes in collagen sponges











Fig. 9.4. No significant differences were measured in cell viability between stimulated and non-stimulated constructs after 0, 7, 14, 21 and 28 days (p>0.05). Cell viability exceeded 94% at all time periods in culture. Data represented as mean + one standard deviation. N=6 for all groups.

9.5. Discussion

Understanding how chondrocytes respond to altered mechanical loading environments offers both in-vitro and in-vivo benefits. Determining how these cells respond to tensile signals can help tissue engineers better understand how to manipulate cartilage constructs in-vitro. These experiments also have the potential to help investigators discover how the post-implantation loading environment further alters cartilage replacement tissues. These goals motivated our group to develop the model, to isolate double transgenic murine chondrocytes, to examine the effects of tension on chondrocyte-collagen scaffolds, and ultimately to determine the plasticity of these cells to mechanical loading. Using this specialized transgenic murine model also allowed us to test our hypotheses that applying tension to chondrocyte-seeded gels would decrease ECFP fluorescence, increase GFP-T fluorescence and thereby increase the construct's linear stiffness.

Our quantitative fluorescence results support our first hypothesis that tensile stimulation significantly decreases ECFP fluorescence in chondrocytes. The fact that tensile stimulation of S constructs produces significant decreases in ECFP RFU values relative to NS controls at 21 and 28 days (Fig. 9.2) indicates that specialized cells do respond to tensile signals in a somewhat plastic manner, albeit with a decline in gene expression. However, these temporal changes are delayed in time compared to the more rapid increase in Col2 gene expression that we and others have observed when chondrocytes are exposed to dynamic compression [2, 336-337]. These changes are also slower than the response of double transgenic murine MSCs to similar patterns of tensile stimulation [163, 167, 416-421]. At the same time, it should be noted that the Col2 gene is not exclusively sensitive to mechanical signals in chondrocytes. Genes such as MMPs, CTGF, and runx2/cbfa1 in chondrocytes also respond depending on the mode of stimulation [480]. So while chondrocytes respond to oscillatory tensile stimulation in a fundamentally different manner than they do to compression, monitoring the sensitivity of other structural genes like Col1 are also important.

Our second hypothesis that GFP-T fluorescence, an indicator of Col1 gene expression, would increase in chondrocytes exposed to tensile stimulation was not accepted. While a tensile stimulus upregulates Col1 gene expression in other cells types [163, 167, 416-421], none of the S or NS constructs showed GFP-T positive cells during the course of the experiments. The mechanical profile we delivered to these constructs was identical to the one we used when stimulating murine MSC-collagen sponge scaffolds [163]. However, unlike that study where tensile signals significantly upregulated Col1 gene expression using MSCs, we found no such effect using chondrocytes after 28 days of stimulation. It is possible that longer periods of stimulation are required to show GFP-T expression in these more terminally differentiated chondrocytes or that these levels are below the level of detection using this transgenic technology. Studies are ongoing in our laboratory to test each of these possibilities and to

discover when and if chondrocytes are capable of showing GFP-T expression in response to different tensile stimuli in culture.

Our third hypothesis that tensile displacements applied to chondrocyte-collagen constructs would increase TEC stiffness was also refuted. We were initially surprised that tensile displacements actually decreased linear stiffness of stimulated constructs compared to non-stimulated controls at day 28.Given the fact that tensile stimuli have no effect on Col1 gene expression in these constructs, we might have expected no change rather than a decline in TEC stiffness. Previous studies have shown that compressive stimulation of chondrocyte-seeded collagen constructs increases matrix protein [166, 481-482] and proteoglycan synthesis [426-427] and application of tensile forces produces the opposite effect [483]. Others have also shown that tensile stimulation of chondrocyte-scaffold constructs increases MMP gene expression [164, 480].Thus, it is conceivable that such decreases in protein, proteoglycan synthesis and increase in MMPs could explain the decrease in linear stiffness observed in our constructs. Our future studies will also be monitoring these other gene and protein measures to determine if our chondrocyte-collagen scaffolds demonstrate similar effects.

Our quantitative fluorescence results do support our final hypothesis that Col2 gene expression will not decrease with culture time in non-stimulated constructs. Quantitative ECFP RFU data (Fig. 9.1) shows that there are slight decreases in ECFP RFU in NS constructs over time, but such decreases are not statistically significant. Previous studies have shown that collagen scaffolds have the ability to maintain chondrocytic characteristics [166, 426-427, 481-482, 484-485]. Collagen gels has been shown to provide an environment in which chondrocytes can expand [427] and produce matrix proteins [166, 481-482], proteoglycans [426-427] and maintain Col2 gene expression [166, 485]. Protein synthesis is also enhanced when collagen-chondrocyte constructs are exposed to growth factors [484]. Our results confirm that type I collagen sponges do have the ability to maintain Col2 gene expression in chondrocytes.

Our study is not without limitations. We pooled our cells within a sample in order to detect sufficient fluorescence levels. This pooling means that we only know the average effects and not the specific effects of tension on individual cells. We are currently examining techniques like in situ hybridization to examine more local effects of tension on individual cells. We did not track cell-induced changes in the production of proteins and proteoglycans nor changes in integrin expression. These factors likely play an important role in regulating intracellular gene expression and the ability of cells to sense their near environment. Our future studies will track the synthesis of these constituents and their effect on our outcome measures. We applied only one stimulus pattern to the constructs. We still need to systematically vary components of the mechanical signal and determine their effects on gene expression, protein synthesis and biomechanical properties. Finally, we chose to measure only the construct's linear stiffness. Stiffness was selected because we were concerned that the lower aspect ratio of 2:1 (less than the ideal ratio of 3:1 to avoid St. Venant effects) might alter the structure's failure properties. Future studies will examine how constructs with larger aspect ratios influence our results.

We conclude that tension applied to cell-seeded TECs decreased Col2 gene expression but did not induce increase Col1 gene expression in mouse chondrocytes. Tension also decreased the linear stiffness of constructs. Expanding our studies to examine longer time intervals and different components of tensile stimulation as well as monitoring protein and proteoglycan synthesis should provide a more comprehensive picture of the ability of specialized chondrocytes to undergo dedifferentiation or transdifferentiation in response to more complex mechanical signals. Ultimately, such culture experiments using double transgenic stem and differentiated cells should help investigators in selecting cell sources and scaffold types that more rapidly accelerate the repair of diseased and injured connective tissues.

Chapter 10

Discussion and Conclusions

Chapters 2 through 9 describe a series of studies using a fluorescent promoter transgenic technology in the mouse model to examine Col1 and Col 2 gene expression during natural healing of patellar tendon defect injuries, during growth and development, and following mechanical stimulation of tissue engineered constructs containing primitive or specialized cells. The major findings from these studies are summarized and discussed in this chapter. This chapter also highlights the overarching influence of all of these studies on tendon tissue engineering.

<u>Punch defect healing of murine patellar tendons produce spatial and temporal changes</u> in Col1 gene expression and results in inferior tissues.

Our results from Chapter 2 document the spatial and temporal changes in swelling and disorganization levels, Col1 and Col2 gene expression levels, following punch defect and sham injuries. Both swelling and disorganization decreased with healing time. Col 1 gene expression after injury increased and then decreased in intensity while also becoming less localized. This emphasizes how an entire tissue can be affected by a local punch injury, possibly as a result of the disruption of vascular supply as well as loss of biomechanical tension to the tendon and its tenocytes proximal and distal to the defect site. Our temporal Col1 expression results are in agreement with others in the literature, which shows Col1 expression occurring in healing tendons between 10-24 days [256] [255] [253] [249]. Our Col1 gene expression patterns might indicate which phase of healing is occurring in these tendons and establish timelines for these phases. Very little Col1 gene expression (together with observed swelling and disorganization) at 1 week indicates that the healing tendons are still in the inflammatory phase. High levels of

Col1 expression at 2 weeks suggest that the healing tendons are in the proliferative phase. Decreases in Col1 (together with decreases in swelling and disorganization) at 4 weeks suggests that the tendons may have entered the remodeling phase.

No ECPF expression (Col2 expression) was seen in the entire tendon. This indicates that the Col2 gene is not involved during the repair of tendon mid-substance. This is not surprising as Col2 expression is not seen in the mid-substance of the normal tendons. Since Col2 expression is localized to the insertions, Col2 expression might be a factor in the healing of insertions. The temporal patterns of Col1 expression we observed during mid-substance healing suggests that temporal patterns of Col2 expression might also exist during insertional healing This is supported by preliminary insertional injury studies done in our lab which show elevated Col2 gene expression in the insertions by 2 weeks (not shown). These insertional injuries have to be examined further to determine the complete temporal and spatial expression patterns for Col2 expression there.

Force-elongation and stress-strain curves demonstrate that while not different from each other at 4 and 6 weeks, healing tendons were still significantly compliant and weak compared to normal tendons (p>0.05). Poor biomechanics of natural healing was expected as an early study in our lab using rabbit PT window defects showed that natural healing was inferior [257]. What is surprising is that even long term healing (63 days) of murine intrasynovial tendon adhesions produce inferior tissues [256]. These results taken together with our own reinforce that even in the long term, true tendon regeneration is not occurring. Thus, new strategies are required to augment such repair. One promising approach is to determine what happens during normal tissue development as to provide possible stimuli that we might impose during adult healing. This was the motivation for the studies on growth and development.

In summary this repair model and models like it should allow researchers to develop a roadmap for tissue engineering that more closely simulates normal development, potentially leading to regeneration rather than repair of tendon injuries.

The current study was not without limitations. 1. Our design did not permit us to determine when Col1 and Col2 expression levels first became detectable since we only started monitoring the expression of both genes at 7 days. We are currently examining expression after only 3 days. Moreover, once gene expression can be truly monitored in real time, we can answer this question. 2. The biomechanical results for the healing regions included some adjacent strut tissue due to the extremely small size of the repair area. Thus, our biomechanical results may have overestimated the true properties of the repair tissues. This inclusion of adjacent struts might also explain why 4 and 6 week healing properties were not different. 3. We recognize that the small size of these tendons increases the difficulty of inducing repeatable injuries in the murine model. Future studies, particularly those involving tissue engineering, may require using larger animals if genetic markers can be incorporated. 4. We could not directly explain our biomechanical results based on gene expression patterns. The improvements we observed in biomechanical properties between 4 and 6 weeks were quite encouraging, but our experimental design did not include an assessment of either gene expression or tissue structure at 6 weeks post injury. However, based on gene expression results between 2 and 4 weeks combined with our 6-week biomechanical results, we can speculate that gene expression levels likely undergo further decreases toward control levels with corresponding reductions in tissue thickness and more parallel alignment of collagen fibers. 5. These acute punch defects do not replicate the *clinical problems.* This punch defect model provides a way to biologically and mechanically evaluate the effects of natural healing. Rabbit models which are closer to the clinical conditions are being developed in our lab and should also be extended to the mouse model in the future.

<u>Developing knee structures undergo spatial and temporal changes in Col1 and Col2 gene</u> expressions, BMP-2 expression and TGF-β1 expression.

Results from chapter 3 indicate that there are spatial and temporal changes in ECFP/GFP-T fluorescence patterns in several knee structures (i.e. Col1/Col2 gene expression).

These changes in fluorescence around birth suggest that knee structures undergo regulatory changes which could be controlled by either biologic or physical factors. A recent study of rat supraspinatus entheses indicates that physical factors do not play a significant role in enthesis maturation in the first 14 days postnatally, implying that biologic factors may drive early postnatal development [272]. Thus our Col1/Col2 expression patterns could be a result of differential expression of biological factors, especially BMP-2 and TGF- β 1, as these have been shown to have important roles during joint development [486].

Immunohistochemistry did reveal differences in growth factor patterns across age groups. For example, BMP-2 was only evident in the PT mid-substance until P1 and in the tibia at P14. BMP signaling has been suggested to control regional differentiation in tendon. Generally BMP signaling is inhibited in the tendon mid-substance, allowing tendon fiber formation, but is activated at the insertion site, allowing fibrocartilage formation. The appearance of BMP-2 in the tendon mid-substance is contrary to what we expected. It has been suggested that BMP-2 acts a regulator of TGF-β1 signaling which is important in the mid-substance.

The absence of BMP-2 in the tendon mid-substance beyond P1 is also significant suggesting that BMP-2 is likely important only during early development of the insertions and that other members of the BMP family could play important roles during later development.

pSmad1,5,8 expression did not mirror that of BMP-2 expression. Since only the BMP family acts through the pSmad1,5,8 pathway, other members of the BMP family are also active during development. Future studies are necessary to determine which BMPs are expressed at these developmental time points. Once these BMPs have been identified, these should be used to stimulate TECs in culture to see if such stimulation improves not only the in-vitro mechanical properties of TECs but also the resulting in-vivo repair mechanical properties.

Unlike BMP-2, TGFβ1 was seen in the patellar tendons at all ages. TGF-β1 has been shown to have a central role during morphogenesis and tissue formation by influencing the differentiation of connective tissue precursors into tendon forming cells [275]. Thus this

expression of TGF- β 1 can help to explain the increases in Col1 gene expression we observed. Also the appearance of TGF- β 1 at all time points suggests that this growth factor is likely important not only during early development, but also during later development of the tendon mid-substance. This suggested role of TGF- β 1 during development taken together with results in the literature showing improvement in collagen synthesis [276-277] and improvements in healing of tendon mechanics [278] makes this a key factor to be investigated during the development of tissue engineered constructs for tendon repair. In fact a preliminary study in our laboratory demonstrated that adding 10 ng/ml of TGF- β 1 significantly increases rabbit MSCcollagen stiffness which was similar to the stiffness produced by optimization of the mechanical stimulus.

pSmad2,3 expression mirrors that of TGF- β 1 expression. This suggests that the expressed TGF- β 1 acts on the cells in the tendon mid-substance and further emphasizes the importance of TGF- β 1 in the formation of tendon mid-substance and possibly delivery of this growth factor locally to TECs.

Improving the integrity of tendon tissue engineered constructs in culture is one of the most important problems facing tendon tissue engineers. Our growth and development results motivate new strategies to solve this problem. For example, we observed that Col1-expressing cells increased fluorescence levels and were already well-aligned with the tendon axis by P3. Such knowledge about organization and gene expression of cells in the developing tendon could be used to try and recreate more useful tissue engineered constructs, e.g. by the application of tensile displacements to improve their integrity.

Recreating tendon zonal attachment sites after injury is another important problem facing tendon tissue engineers [171] [62, 267-269]. Our growth and development results motivate new strategies to solve this problem. Our results showing: 1) appearance of Col2 before Col1 during development, 2) early co-expression of Col1 and Col2 in insertion regions, and 3) differences in expression of BMP-2 vs.TGF- β 1, suggest that to recreate insertions during

tendon repair, investigators should monitor and attempt to enhance the co-expression of Col1 and Col2 in the end regions of tendon constructs. Researchers should also strive to regionalize the expression of BMP-2/TGF- β 1, if these are lacking in the current set of FTE tendon constructs. Since mechanical signals are known to modulate gene expression, the next set of in vitro studies were conducted to identify appropriate mechanical signals and cell source (primitive vs specialized) necessary to create constructs showing both Col1 and Col2 gene expression.

An important question which arises from these growth and development studies is the correlation of our murine results with findings from larger species, especially humans. It is important to first establish if there are correlations between the mouse and another higher species (for example, rabbit) and then the next higher species (for example, goat) and so on until humans. It is plausible that there are some correlations which can be found by taking into account differences between mice and humans. We might have to deliver different concentrations based on either body mass or tissue size as well as corresponding periods in the life cycle (e.g. early skeletal maturity and in senescence).

In summary, the results from this study illustrate some of the problems that need to be addressed to effectively repair tendon injuries. We need to reconstitute sufficient key regulatory events from the normal spatial and temporal differentiation of the patellar tendon to enhance the maturation of cells in vitro, and thus enhance tendon repair. Our growth and development data begin to suggest biological benchmarks with which to compare tissue engineered constructs and repair. These biological benchmarks complement the mechanical benchmarks our group has been developing over the past decade.

The current studies were not without limitations. 1) *We only monitored GFP-T and ECFP fluorescence as early indicators for Col1 and Col2 gene expression in sections of the murine knees*. While these fluorescent markers may appear to be subjective measures, we have previously shown positive significant correlations between relative fluorescence units

or RFUs and qRT-PCR for Col1 and Col2, respectively [2, 163, 279]. 2) We did not measure actual Type I or Type II collagen protein expression from these sections. However, we have shown in culture that Type II collagen protein is present within 7 days after upregulation of Col2 gene expression [2, 163, 279]. 3). Although our double transgenic model should allow us to track changes in Col1 and Col2 expression in near real time, currently we must sacrifice animals and section knees to image the fluorescence. Once we can identify powerful in vivo optical technologies (e.g. dual photon methods) that we can use to visualize at the cellular level, we should be able to image the entire knee with the required magnification and conduct cost- and time-efficient longitudinal studies that eliminate assignment of animals for gene expression patterns at different ages and healing times.

<u>Cell-culture conditions can be created and maintained in a bioreactor housed in the ELF</u> <u>system. This system delivers precise displacements to compliant TECs.</u>

Results from chapter 4 indicate that the number of number of viable, proliferative cells within the bioreactor was not significantly different from that of cells cultured in the standard incubator. This result confirms our ability to create and maintain cell-culture conditions within the bioreactor.

Results from chapter 4 also indicate that the actuator displacement recorded by the internal LVDT of the electromagnetic system is similar to the displacements recorded by a sonomicrometry system. This confirms our ability to apply precise displacements to compliant TECs.

Several features of our bioreactor emphasize its potential for Functional Tissue Engineering. Our bioreactor combines both the culture and stimulation in a one step process. We can also deliver multiple precise stimulation profiles including in vivo profiles to multiple specimens. We can record loads and displacements during culture thus permitting real-time calculation of stiffness. We can compare these values with those for normal tissues, thus

providing biomechanical feedback for adjusting the stimulation to better "challenge" the constructs and to accelerate their maturation in culture for more rapid delivery at surgery.

The current studies were not without limitations. 1) *We only monitored cell viability in monolayer*. However, we have previously shown that we can maintain over 90% viability in MSC-agarose and chondrocyte-agarose constructs. 2) *We only monitored cell viability up to 14 days in culture*. Important to note is the fact that we have also maintained over 90% cell viability in 3-D constructs up to 28 days in culture.

Positive correlations exist between response measures.

Our results from chapters 5 and 8 and Fig 10.1 suggest that fluorescence and collagen mRNA expression are positively correlated and type II collagen content and aggregate modulus are positively correlated. These correlations serve as a validation for using fluorescence as an indicator of average gene expression within these tissue engineered constructs. These correlations are important as they suggest that RFU measures can be used to indirectly estimate gene expression in constructs and type II collagen content can be used to indirectly estimate aggregate modulus or vice-versa. Such correlations could definitely speed up the tissue engineering process. While it takes about 3-5 hours to determine the RFUs in constructs, it takes about 2-3 days to evaluate mRNA content from cells in these TECs. Also these correlations will reduce the costs associated with performing real time PCR and with using multiple samples.

Although the correlations between these response measures is encouraging, further studies are needed to determine if such correlations are found in other cell types or other model systems (e.g. in the Col2 rabbit). Discovering correlations of this type in larger models like the rabbit may soon allow investigators to use fluorescence as a reasonable surrogate of gene expression as opposed to more traditional approaches like qRT-PCR.



Fig. 10. 1. Positive correlation exists between aggregate modulus and type II collagen content in chondrocyte agarose constructs.

Mechanical stimulation affects gene expression when compared with non-stimulated

<u>constructs.</u>

Results from chapters 5 to 9 show the ability of mechanical stimulation to either increase or decrease gene expression compared to non-stimulated controls. Generally, compression increased Col2 gene expression (chapters 5, 6) and decreased Col1 gene expression (chapter 7) while tension increased Col 1 gene expression (chapter 8) and decreased Col2 gene expression (chapter 9).

Our findings for different cell types regarding the effects of compressive and tensile stimulation on collagen gene expression are consistent with other reports in the literature. For example, our Col2 gene expression results showing increases in our chondrocyte-agarose and chondrocyte-MSC systems are similar to results reported in the literature [336] [487] [390, 399] [336-339, 488]. Our Col1 gene expression increases in MSC-collagen constructs are similar to

the results of other studies in the literature [417-418, 444] that have demonstrated positive effects of mechanical stimulation on type I collagen gene expression. Our results are similar to other studies in which fibroblasts exposed to cyclical contraction [423] and hydrostatic pressure [206] expressed significantly lower Col1 mRNA.

Several mechanisms have been put forth by investigators to explain such changes in Col2/Col1 gene expression. Compression has been shown to induce transcriptional activation of SP1 binding sites residing in the proximal region of the col2a1 gene promoter [340], cell/tissue strain, fluid pressurization and flow, electrokinetic phenomena, convective transport, and release of cytokines/growth factors which can trigger the complex chain of events that modulate the production of ECM proteins like Type II collagen [341-342, 489]. Tensile strains are known to trigger the creation of cell-surface stretch receptors and integrins, to activate a cascade of genes responsible for the synthesis and secretion of ECM components [441], to produce cellular changes (cell proliferation, mass transfer rates of nutrients, metabolites and waste materials [173], expression and synthesis of various growth factors and cytokines) [396, 398, 443, 490]. Of particular interest in the above mentioned mechanisms is the expression and synthesis of various growth factors and cytokines) [396, determine which cells express which growth factors during mechanical stimulation. Once identified, experiments can then be performed to determine if stimulating these constructs with growth factors during mechanical stimulation further improves their gene expression.

Taking the results of all studies in this dissertation together it becomes evident that when using mechanical stimulation as a treatment, to increase Col1 expression one has to apply tension and to increase Col2 gene expression one has to apply compression. Thus to create constructs with both Col1 and Col2 gene expression one should apply both compressive and tensile signals to constructs. By designing grips which can provide compression to the end of constructs and by utilizing them in our pneumatic system we can create constructs with Col1
expressing cells in the middle and Col2 expressing cells in the ends. Such constructs containing dual Col1 and Col2 gene expression would be necessary for tendon insertions into bone.

<u>Mechanical stimulation affects mechanical properties when compared with non-</u> stimulated constructs.

Results from chapters 5 to 9 show the ability of mechanical stimulation to either increase or decrease the construct's mechanical properties over non-stimulated controls. Both compression and tension increased the aggregate modulus of agarose constructs (chapters 5 to 7) and linear stiffness of collagen constructs (chapter 8), respectively. The only exception was when chondrocytes were subjected to tension (chapter 9). The increase we observed in stimulated constructs reinforces the importance of pre-conditioning a cell-scaffold construct with a defined mechanical signal in a cell culture system as we have shown that in vitro stiffness correlates with in vivo repair stiffness [171]..

The increases in mechanical properties could be attributed to increases in collagen content of our constructs. Results from chapters 4 and 5 confirm that mechanical stimulation increases type II collagen content of the constructs. Other factors could have also played a role. For example, accumulated proteoglycans such as aggrecan in agarose constructs could have led to increases water content and aggregate modulus. Collagen orientation could have improved in our MSC-collagen constructs. We have previously shown that the sub-failure properties are correlated with collagen content[491]. Hence future studies should try to measure proteoglycans by the use of ELISA and collagen orientation by the use of SALS.

The decreases in linear stiffness of chondrocyte-collagen constructs can be explained by the decrease in protein and proteoglycan synthesis [483]. Tensile stimulation has also been shown to increase MMP gene expression [164, 480]. Such decreases in protein and proteoglycan synthesis combined with increases in MMPs could explain the decrease in linear stiffness observed in our constructs. Future studies will need to track MMP and protein production by ELISA.

<u>Compressive mechanical stimulation increases Type II collagen content of stimulated</u> <u>constructs.</u>

Results from chapters 5 and 6 show the ability of compressive mechanical stimulation to increase type II collagen content in agarose constructs. This increase in type II collagen content (Table 10.1) at later points (days 21 and 28) is similar to the results reported for MSCs [389,

	Day 0	Day 7		Day14		Day 21		Day 28	
		NS	S	NS	S	NS	S	NS	S
СА	0.036 (0.004)	0.043 (0.008)	0.039 (0.003)	0.038 (0.006)	0.044 (0.012)	0.07 (0.004)	0.09* (0.005)	0.1 (0.004)	0.13* (0.005)
MA	-	-	-	0.03 (0.003)	0.027 (0.007)	0.03 [^] (0.002)	0.05*^ (0.003)	0.04^ (0.008)	0.072*^ (0.004)

Table 10.1. Compressive stimulation increased Type II collagen content (%ww) in S constructs. Type II collagen content were higher in CA constructs than MA constructs. * Significant from NS constructs, ^ Significant from CA constructs. N=6.

400] and chondrocytes [488]. For equivalent design and testing conditions, chondrocytes produce inherently greater gene (Table 10.2) and matrix expression than do MSCs (Table 10.1), leading to improved mechanical function. These studies suggest that chondrocytes could be an excellent cell source for regenerating cartilage. However, a major limiting factor using chondrocytes is whether sufficient autologous cells can be harvested, especially in elderly and osteoarthritic patients who exhibit decreased chondrogenic capacity. Hence MSCs are a promising alternative cell source for cartilage repair due to their ease of isolation and expansion and their chondrogenic differentiation potential. These findings suggest that an inherent limitation in matrix formation by MSCs and that culture conditions must be optimized to create constructs whose properties are similar to that of chondrocytes.

	Day Day 7 0		,	Day14		Day 21		Day 28	
		NS	S	NS	S	NS	S	NS	S
CA	11.2	18.1	24.3*	21.6	32.3*	23.2	29.6*	18.9	23.7*
	(1.2)	(2.2)	(2.6)	(4.3)	(2.3)	(3.5)	(2.1)	(1.1)	(3.2)
MA	2.8^	2.9^	3.2^	3.2^	3.3^	3.7^	5.4*^	4.3^	7.1*^
	(1.3)	(1.7)	(1.9)	(1.4)	(2.0)	(1.7)	(1.9)	(2.7)	(2.2)

Table 10.2. Compressive stimulation increased ECFP RFU in S constructs. ECFP RFU higher in CA constructs than MA constructs. * Significant from NS constructs, ^ Significant from CA constructs

While my experimental design did not allow me to quantify collagen content in fibroblastagarose constructs, scaffold material prevented me from quantifying newly synthesized collagen content in MSC-collagen and chondrocyte-collagen constructs. Future studies should try to use 1) radio-labeling techniques to quantify newly synthesized collagen in collagen constructs and 2) ELISA in fibroblast-agarose constructs. Such determination of collagen content is necessary to see if there are correlations between collagen content and biomechanical properties. If such correlation exists then collagen content might become an important predictor for subsequent biomechanics.

Differential maturation and structure-function relationship in chondrocyte-agarose and

MSC-agarose systems.

Results from chapters 4 and 5 and Figs 10.1 A &B show temporal and stimulationinduced relationships among gene expression, protein expression and biomechanics. If additional studies could corroborate these findings in the murine model and in other model systems, gene and protein expression patterns might then serve as predictors of biomechanical response, both in vitro and ultimately in vivo.

In the chondrocyte-agarose constructs (chapter 4 and Fig 10.2 A), the earlier increases in Col2 gene expression translated to later increases in Type II collagen content and still later

increases in aggregate modulus [162].. These results are not surprising. While gene expression must obviously precede protein deposition, any biomechanical benefits arise only after the protein is assembled into a more functional matrix.

For the MSC-agarose model system (chapter 5 and Fig 10.2 B), the increases we observed in Col2 gene expression mirrored the increases in Type II collagen content and aggregate modulus suggesting differential maturation and structure–function relationships in MSC- vs. chondrocyte-seeded agarose TECs. Our results are similar to reports in the literature which also show differential maturation and structure-function relationships between chondrocyte- and MSC-seeded constructs [492-493]. It is not surprising that the undifferentiated MSCs do not respond to biophysical stimulation in a similar manner to fully differentiated chondrocytes. Long-term stimulation of MSCs might be needed to create constructs whose characteristics match those of chondrocytes stimulated for a shorter time. It is possible such long-term stimulation might be actually detrimental. One recent study showed that long-term mechanical compression (up to 48 days) produced lower compressive modulus and GAG content than non-stimulated constructs [400].

It is important to note that these studies were done using cells from different animals, hence making straight-forward comparisons more difficult. Future studies should utilize chondrocytes and MSCs from the same animal to eliminate inter-animal effects.

<u>Specialized cells respond more rapidly and to a greater extent than MSCs when</u> <u>stimulated with signals that they normally experience in vivo.</u>

Results from chapters 4 and 5 show that chondrocytes upregulate Col2 gene expression more rapidly than MSCs when cell-seeded TECs are exposed to compression (Fig. 10.3). MSCladen constructs also possessed properties lower than those of chondrocyte-seeded constructs at equivalent time points (Table 10.1, Table 10.2 and Table 10.3). These findings for gene expression, protein content and biomechanical properties of MSCs vs. chondrocytes are consistent with observations by others [492, 494-495]. This suggests that methods for inducing MSC-based chondrogenesis have yet to be optimized to produce cells whose functional matrix-forming potential matches that of native chondrocytes. This also suggests that if one's primary aim is to tissue engineer cartilage and sufficient chondrocytes are available, then using chondrocytes will likely be more beneficial than using MSCs.



Fig 10.2 Different relationship among RFU, type II collagen content and aggregate modulus between A) Chondrocyte-agarose and B) MSC-agarose systems. Squares denote RFU, triangles denote Type II collagen content and circles denote aggregate modulus

I only performed tensile experiments on MSCs and not on fibroblasts. But preliminary studies performed in our lab (thanks to Nat Dyment and Andrea Lalley) suggest that fibroblasts in TECs show greater Col1 gene expression when exposed to tension than when MSCs in TECs are placed in tension (Fig 10.4). This also suggests that if one's primary aim is to rapidly tissue engineer tendon mid-substance, then using fibroblasts may be more beneficial than using MSCs.

Comparisons between MSCs vs chondrocytes exposed to compression and MSCs vs fibroblasts exposed to tension lead me to conclude that specialized cells respond more rapidly to signals that they experience normally in vivo than do MSCs. It should be noted that these results were obtained using cells from different animals. To eliminate inter-animal effects, future studies should be performed using cells from the same animal.

	Day 0	Day 7		Day14		Day 21		Day 28	
		NS	S	NS	S	NS	S	NS	S
СА	11.5	12.9	15.9	14.2	16.0	15.7	18.5	14.9	25.1*
	(1.0)	(3.4)	(2.5)	(1.5)	(2.4)	(3.1)	(4.4)	(2.3)	(1.5)
MA	9.4	9.6	9.8	10.1	9.7	9.8 [^]	13.5*^	10.6 [^]	16.5*^
	(1.1)	(3.4)	(2.5)	(1.6)	(2.3)	(1.2)	(1.5)	(2.4)	(3.2)

Table 10.3. Compressive stimulation increased aggregate modulus (KPa) in S constructs. Aggregate moduli were higher in CA constructs than in MA constructs. * Significantly different from NS constructs, ^ Significantly different from CA constructs

Specialized cells respond more slowly than MSCs when stimulated with signals they do

not normally experience in vivo

Our quantitative fluorescence results in chapters 7 and 9 show that stimulation of specialized cells with signals they do not normally experience in vivo decreases Col2 fluorescence in chondrocytes and Col1 expression in fibroblasts but only at later points of stimulation (Fig. 10.5 A&B). The fact that such decreases occur only at later time intervals in culture indicates that specialized cells show modest plasticity. In particular, these temporal changes are delayed in time compared to the more rapid increases in Col2 gene expression that we and others have observed when chondrocytes are exposed to dynamic compression (chapter 5) [2, 336-337]. These changes are also slower than the response of double transgenic murine MSCs to similar patterns of compressive (chapter 6) and tensile stimulation (chapter 8)

[163, 167, 416-421]. These results lead one to conclude that MSCs are generally more plastic than specialized cells (i.e., they respond more rapidly than specialized cells to unconventional mechanical signals in culture).



Fig 10.3. Compressive response of chondrocytes vs MSCs in TECs. Chondrocytes show more rapid increases in ECFP expression than MSCs when stimulated in compression.



Fig 10.4 Tensile response of Fibroblasts vs MSCs in TECs. Fibroblasts show more rapid increases in GFP-T expression than MSCs when stimulated in tension.



Fig 10.5. A. Chondrocytes respond more slowly than MSCs when stimulated in tension B. Fibroblasts respond more slowly than MSCs when stimulated in compression

MSCs may be a better candidate to create constructs where both Col1 and Col2 gene

expressions are desired

Even after 28 days in culture, fibroblasts did not express the Col2 gene (chapter 7) nor did chondrocytes express the Col1 gene (chapter 9). Though the mechanical profile we delivered was the same as the one which upregulated Col1 and Col2 gene expression in MSC-scaffolds (chapters 6 and 8), it might be plausible that the specialized cells require a different mechanical signal and longer time in culture to express the other gene. It is also plausible that these genes were expressed at such low levels that we could not detect the fluorescence. RT-

PCR studies are needed in the future to determine which genes are actually being expressed and their spatial and temporal profiles.

Our results also show that mechanically stimulated TECs containing MSCs can significantly express both Col1 and Col2 after 14 days in culture (chapters 6 and 8) (Fig 10.6). These results have huge implications when one is trying to engineer tissue with both Col1 and Col2 gene expression. Results from our studies suggest that MSCs rather than specialized cells are a suitable cell source if we need to create structures containing both Col1 and Col2 expressing cells.



Fig 10.6 Compressive vs tensile response of MSCs in TECs Our in vitro studies are not without limitations. 1) *We did not track real-time changes*

in collagen gene expression in the intact construct. These structures are currently too thick and opaque to perform such non-destructive imaging. We continue to seek novel imaging technologies to observe such changes in cellular fluorescence that would permit us to perform longitudinal studies on the same constructs over time. 2) *The fact that we pooled our cells to record fluorescence means that we only know the average effect of dynamic compression and not the specific effects of compression on individual cells*. It has been previously shown that gene expression varies between the core and the annulus region of the hydrogel [389]. In the future, we plan to evaluate gene expression in both regions of the agarose constructs as well as in the mid and end regions of the collagen constructs. 3) Autofluorescence of agarose gel and collagen fragments could increase overall RFU *values*. In these studies, we assumed that these changes were constant across groups. 4) We did not specifically track changes in the production of proteoglycans like aggrecan in agarose constructs. Increases in aggrecan content could reflect improvements in a construct's aggregate modulus. In the future we intend to track the production of specific proteoglycans like dermatan sulfate. 5) We chose to measure only the collagen construct's linear stiffness because we were concerned that the lower aspect ratio of 2:1 (less than the ideal ratio of at least 3:1 to avoid St. Venant effects) might alter the structure's failure properties. In the future we are planning to utilize longer constructs to increase the aspect ratio and hence minimize St. Venant effects. 6) We applied only one stimulus pattern to the constructs. We still need to systematically vary components of the mechanical signal so as to optimize gene and protein expression as well as construct biomechanics in the shortest possible time interval. We are, however, performing these studies using rabbit MSCs as part of another research project. 7) Our plasticity experiments on specialized cells were only performed up to 28 days in culture. We may need to compressively stimulate our TECs for longer durations to induce Col1 gene expression in chondroyctes as well as Col2 gene expression in fibroblasts.

Technical problems to be overcome in order to perform in vivo studies in the mouse

One of the important findings in the rabbit model is that in-vitro construct stiffness and the in-vivo repair stiffness are positively correlated (302). Such correlations may also exist in the mouse model but in-vivo studies are necessary to establish such correlations. The small size of the mouse not only makes in vivo surgeries a challenge but also makes it difficult to evaluate the effect of implanting tissue engineered constructs to repair defects in the patellar tendon. One of the major problems encountered is securing the tissue engineered constructs in the defect site. While this is achieved with the help of sutures in the rabbit model, such a process will not work in the mouse as the sutures are quite large relative to width and thickness of the tendon and would create a significant injury to the tissues. Also, simply inserting the constructs in the defect site does not produce a 100% success rate. We conducted preliminary studies in our laboratory where Di-I labeled MSCs were seeded in either KN sponges or ESL-GAG sponges. These TECs were then used to fill the punch defect site in six animals (n=3 for each sponge type). After one week, Di-I labeled cells were seen in the repair sites of only two animals implanted with KN sponges and in none of the repair sites containing the ESL-GAG sponges. These preliminary results strongly emphasize the need to find ways to secure the constructs in the defect site without inducing ancillary effects unrelated to the treatment and to translate our results to larger animals where more reproducible surgery is possible.

Several factors could have influenced the results of the in-vitro plasticity experiments

Three experimental factors could have influenced the results of my in-vitro experiments. These include: 1) animal variability, 2) type of biomaterial, and 3) stimulation profile. I will discuss each of these factors and how they might have affected the outcomes.

Comparing the results of studies using different animals and ages can be difficult. While I was able to control the age of animals from which I harvested fibroblasts and MSCs, I was forced to use newborn rib chondrocytes in my chondrocyte-agarose constructs as the yield of chondrocytes from adult ribs was very low and rather heterogeneous (i.e. they contained fibroblasts). Such differences in age make comparisons of results between MSCs, chondrocytes and fibroblasts difficult. Also I also used cells from different animals in these plasticity experiments once again making comparisons difficult. To minimize such animal effects, future studies should directly compare and contrast the plasticity of cells obtained from the same animal.

Use of different biomaterials in the plasticity experiments made comparisons between studies more challenging. I used agarose for compression studies and collagen for tensile studies. This was because agarose does not resist tension very well and collagen does not resist compression. Use of different biomaterials makes straight-forward comparisons difficult because of the different cell-matrix interactions. It is well known that cell-matrix interactions have a major influence on the biosynthetic response of cells (466, 489). Use of different biomaterials makes it hard to decouple cell-matrix interaction effects and mechanical stimulation effects while comparing cell compression vs cell tension studies.

In this dissertation I used only one profile for compression and one profile for tension for the in-vitro cell studies. The results of these studies suggest that MSCs are more plastic than specialized cells. The conclusions that I have reached are only valid for the experimental conditions described in the chapters. It is possible that by varying the profiles of mechanical stimulation we might find that specialized cells are more plastic than MSCs and future studies should focus on these issues.

Other factors could have also influenced the results of the plasticity experiments. Observed fluorescence is an indicator of the gene expression of only that promoter to which it is linked. It is important to keep in mind that Col1 and Col2 are under the regulation of many promoters. In our mouse model the topaz and the cyan fluorescent proteins are linked to only one promoter for the Col1 and Col2 genes, respectively. Hence these fluorescent proteins do not indicate the activity of all promoters for Col1 and Col2 genes but only the promoter to which each is linked. These could have played a role in the in-vitro experiments. For example, tensile stimulation could have activated the promoter linked to GFP-T in MSCs, but not the promoter of interest in the case of chondrocytes. This could explain why we did not see GFP-T expressing cells when chondrocytes were stimulated in tension. This could also explain why no ECFP-expressing cells were seen when fibroblasts were exposed to compression.

It is also possible that in chondrocytes exposed to tension, GFP-T was produced but in such low quantities that it was impossible to detect. The same problem could have occurred when fibroblasts were exposed to compressive strains. Hence in the future, real time PCR should be performed in these plasticity experiments of specialized cells to see the levels of Col1 gene expression found in chondrocytes and the levels of Col2 gene expression observed in fibroblasts. We will be initiating these experiments shortly.

Translating results from murine to rabbit models is possible

The main limitation of the mouse model is that we cannot use it to reproducibly test our tissue engineered constructs after surgery. Still this mouse model has its advantages in tissue engineering especially if we take advantage of the genetic tools that are available to perform in vitro studies and then translate the results to larger animal models such as the rabbit where validation of these in vitro studies can be performed in-vivo with ease. Establishing correlations in TEC behavior across these two species might be important if we are to more effectively repair tendon injuries in larger models.

These studies are being performed in our laboratory in order to determine if transgenic and wild type mouse and rabbit cells respond in similar ways to mechanical and chemical signals in culture. For example, tensile stimulation of constructs containing either double transgenic (DT) murine MSCs or rabbit MSCS show similar patterns of increasing Col1 gene expression (307,406). Similarly, compressive stimulation of constructs containing either DT murine chondrocytes or single transgenic rabbit chondrocytes show similar patterns of Col2 gene expression in both species. These results are encouraging and suggest that we can create corresponding animal models and translate our results across species. These comparison studies are ongoing in our laboratory.

To formally link these results across species and unify our mouse and rabbit protocols, we will need to perform similar experiments using cells from both species. Currently differences exist in the way we make and mechanically stimulate mouse and rabbit MSC constructs. These limitations make comparison of results somewhat difficult. Hence following the same protocols for both species becomes important. It should be noted, however, that we may not be able to completely unify these protocols. For example, we already have shown that culturing mouse MSCs requires a special formulation of media unlike that for rabbits. Still, creating conditions as similar as possible across species will make interpretation of our inter-species data easier.

Can chemical stimulation be used to modulate Col1 and Col2 gene expression?

In this dissertation, I chose to mechanically stimulate TECs to modulate Col1 and Col2 gene expression levels in primitive and specialized cells. I could have chosen chemical stimulation (e.g., growth factors) to modulate these genes within TECs. Some of the candidates for chemical stimulation are transforming growth factor beta (TGF- β), basic fibroblast growth factor (b-FGF), insulin-like growth factor (IGF) and bone morphogenetic protein (BMP). TGF- β has been shown to stimulate collagen and glycosaminoglycan synthesis by fibroblasts [276-277]. TGF-β1 induces a dose-dependent increase in the expression of procollagen type I and III mRNA which improves failure load and stiffness in the healing tendons [278]. Investigators have also shown a dosedependent increase in the number of proliferating cells and the level of expression of type III collagen after in vivo supplementation with b-FGF [496]. This growth factor offers the potential to stimulate the proliferation of fibroblasts in culture and increase the speed of healing of tendon repairs. Another growth factor, insulin-like growth factor (IGF), has been reported to elicit dosedependent effects on collagen and proteoglycan synthesis in rabbit flexor tendon [497-498]. BMPs have also been shown to increase the expression of Col2 in a variety of cell types. Hence chemical stimulation could conceivably be used in place of or in addition to mechanical stimulation to modulate Col1 and Col2 genes within TECs.

The important question which naturally arises when comparing these two methods of stimulation is whether chemical stimulation is as effective as mechanical stimulation in altering gene and protein expression as well as biomechanical outcome. With some exceptions, preliminary studies suggest that chemical stimulation may be better than mechanical stimulation when preparing cells for tissue engineering applications. When we exposed DT murine MSC's to BMPs, the cells began to express ECFP by seven days. When these MSC's were mechanically stimulated, however, ECFP expressing cells were only present at fourteen days. Similarly, addition of 10 ng/ml of TGF-β1 to rabbit MSCs led to significantl increases in construct stiffness to levels

produced by the optimized mechanical stimulus [499]. Clearly the effects of chemical stimulation have to be examined further.

But the use of chemical stimulation to modulate genes of interest and to improve the biomechanical properties of constructs has some drawbacks. First and foremost is the cost associated with the use of growth factors as these are are very expensive. We still need to optimize the components of the chemical signals (e.g dosage concentration, frequency and total time of stimulation) and remain concerned about duration of effect and potential adverse reactions should high concentrations of factors be introduced.

Overarching Conclusions and Discussions

Inferior biomechanics of natural healing of patellar tendons even after 6 weeks of healing emphasizes the need for strategies such as FTE to improve and speed the repair process. Normal growth and development data suggests that we should create constructs that mimic both Col1 and Col2 gene expression patterns to recreate zonal insertions. One strategy to create constructs with both Col1 and Col2 gene expressions is to introduce primitive cells such as bone marrow-derived mesenchymal stem cells or MSCs into biological scaffolds and then precondition these MSC-based constructs to produce tendon-like cells in the mid-substance and fibrocartilage-like cells near the attachments. We might accomplish this by using different stimulation methods in the mid-substance vs. the ends. Another possible strategy would be to use specialized cells (fibroblasts or chondroctyes) in different regions of the constructs and then apply mechanical strains to create constructs with tendon-like cells in the mid-substance and fibrocartilage-like cells in the insertions. Examining the plasticity of MSCs vs. chondrocytes and fibroblasts helps us in choosing which of the two strategies mentioned above is more effective. Results from all these studies taken together leads me to conclude the following. Specialized cells (fibroblasts and chondrocytes) respond more rapidly and to a greater extent than MSCs when stimulated with signals that they normally experience in vivo. MSCs respond more rapidly and to a greater extent than specialized cells when constructs containing the latter cells are

stimulated with signals they do not normally experience in vivo. If we want to create a TEC with both Col1 and Col2 expressing cells, then MSCs seem to be a better candidate than using two specialized cell types like chondrocytes and fibroblasts. In total, precursor MSCs are generally more appropriate to use in a tissue application where cells experience multiple types of mechanical stimuli and hence may be more suitable for composite tendon tissue engineering.

The conclusions that I have reached are only valid for the experimental conditions described in the chapters. It is possible that by varying the profiles of mechanical stimulation we might find that specialized cells are more plastic than MSCs. It is also entirely plausible that by changing the biomaterial, we might find that specialized cells are more plastic than MSCs under these new conditions. Further in vitro and in vivo studies must be conducted to answer these questions of plasticity, in-vitro response and in-vivo repair outcome.

Chapter 11

Recommendations

By applying the principles of FTE our group has been successful in using autologous mesenchymal stem cell MSC-collagen constructs to accelerate repair of tendon injuries in the adult rabbit model [51, 150, 257, 500-502]. In spite of early encouraging success, our group faces numerous challenges in tendon tissue engineering. 1) The repair often consists of inferior scar tissue. 2) The zonal insertion of tendon into bone is not restored, making it vulnerable to re-injury. 3) The healing time is too long. 4) Cultured TE constructs can only be used in a load protected regime. 5) The tendon tissue engineering process is iterative with incremental success.

To overcome these challenges we clearly need new strategies. 1) One strategy is to mimic events occurring during growth and development during the tissue culture process. 2) Another strategy is to use either primitive cells and or specialized cells to produce tendon like cells in the mid-substance and fibrocartilage-like cells near the attachments. Our double transgenic mouse system with fluorescent transgenes for Col1 and Col2 becomes very useful as it allows us to investigate the above mentioned strategies quickly and effectively.

Using these transgenic mice, the first part of my dissertation has been focused on understanding Col1 and Col2 gene expression patterns in natural healing and normal development. Using cells from these transgenic mice As well as bioreactors, the second part of my dissertation research has focused on how mechanical signaling of cell-based TECs in culture affects the plasticity of primitive MSCs and specialized chondrocytes and fibroblasts. The results from these studies are presented in chapters 2 through 8. Based on the findings from the studies, this Chapter gives recommendations so that future investigators might improve their understanding of these aspects, a step essential to achieving tendon repair with recreated zonal insertions while decreasing the time post surgery to achieve this desired repair outcome.

This Chapter also describes some of the limitations of this research and ways to overcome these limitations.

11.1. Measurement of Fluorescence in Constructs

11.1.1. Average Effects of Stimulation on Fluorescence

The fact that we pooled our cells to record fluorescence means that we only know the average effect of stimulation. This average effect might mask differences in expression in different regions of the constructs. For example it has been previously shown that gene expression varies in the core and the annulus region of the hydrogel [389]. Future studies should look at the gene expression in the annulus and core regions of compressed hydrogel and also around the posts and mid-regions of the collagen constructs.

This average effect also does not let us determine the specific effects of stimulation on individual cells. Two techniques could be applied to determine the effects on individual cells. One is in-situ hybridization and another one uses laser capture micro dissection (LCM) in conjugation with micro array analysis to quantify local expression. This knowledge about the local gene expression of cells becomes even more powerful if we can couple it with the local stress and strain fields in the neighborhood of these cells. Local stresses and strains in our constructs could be computed using Finite Element Analysis.

11.1.2. Quantification of Fluorescing Cells

We did not quantify the number of fluorescing cells in constructs containing primitive cells. This prevented us from distinguishing whether the increases in fluorescence were due to increased numbers of fluorescing cells and/or increased production of ECFP/GFP-T by already fluorescing cells. Determination of the number of ECFP and GFP-T fluorescing cells becomes even more important when we have constructs with a mixture of ECFP and GFP-T cells. The number of fluorescing cells can be obtained by passing the collagen digests and agarose homogenates through FACS. Future studies should explore the option of utilizing FACS to count the number of fluorescent cells.

11.1.3. Non-destructive Assessment of Constructs

We are currently utilizing non-destructive methods to quantify fluorescence, as these constructs are currently too thick and opaque to perform non-destructive imaging. I strongly recommend that we continue to seek novel imaging technologies to observe such changes in cellular fluorescence that would permit us to perform longitudinal studies on the same constructs over time. This will not only speed up the tissue engineering process, but also reduce cost and the number of animals used in the study.

The other limitation that I faced is to determine when Col1 and Col2 expression levels first became detectable as we only image these constructs at set time points. This limitation can be overcome if we can couple imaging technologies with our existing bioreactor systems.

11.2. Differentiation of Cells in Constructs

One important limitation that our group faces while using rabbit autologous-MSCs is that we cannot determine what if any percentage of MSCs have differentiated into fibroblasts in these constructs. This is because of the lack of cell surface markers in the rabbit model. These surface markers are available in the mouse model. I recommend that the future studies utilize these markers to study the level of differentiation in our cellular constructs as Col1 and Col2 gene expression does not provide information regarding the state of the differentiation of cells. This information is essential to determine if the cells are undergoing differentiation or de-differentiation and therefore their degree of plasticity.

11.3. Experimental Conditions

As mentioned in the discussion chapter three experimental factors could have influenced the results of my in-vitro experiments. These include 1) animal factors, 2) type of biomaterial and 3) stimulation profile. Future studies should be done to either minimize or eliminate these factors. To eliminate animal factors, future studies should compare and contrast plasticity of cells obtained from the same animal. While I was able to control the age of animals from which I harvested fibroblasts and MSCs, I was forced to use newborn rib chondrocytes as the yield of chondrocytes

from adult ribs was very low and heterogeneous. These differences in ages could have influenced the results of the plasticity experiments. It is plausible that chondrocytes from older animals (e.g. 10 week old) might behave differently. It is also plausible that new born fibroblasts and MSCs might behave differently than cells harvested from 10 week old animals. Hence future studies should compare and contrast the response of new born and juvenile cells.

Use of different biomaterials in the plasticity experiments made comparisons between studies more difficult. I strongly suggest that we identify a novel biomaterial that can resist both compression and tension and use that for the plasticity experiments. PLGA seems to be an ideal candidate as investigators have been already been using this material in tendon and ligament repair it can be stimulated in both in tension and compression, it is biocompatible, and it has been shown to enhance cellular attachment, proliferation and matrix production [503-506].

Also I used only one compressive profile and one tensile profile for in-vitro cell studies. The results of these studies suggest that MSCs are more plastic than specialized cells. The conclusions that I have reached are only valid for the experimental conditions described in the chapters. It is possible that by varying the profiles of mechanical stimulation we might find that specialized cells are more plastic than MSCs and future studies should focus on these issues.

11.4. Growth and Development of Knee Structures

We only performed histological studies to map spatial and temporal expression of Col1 and Col2. We did not measure type I and II collagen production. Future studies should try to quantify type I and II collagen proteins and try to correlate these with gene expressions.

Although our double transgenic model should allow us to track changes in Col1 and Col2 expression in near real time, currently we must sacrifice animals and section knees to image the fluorescence. Similar to what I have said in section 10.1.3 we need to identify powerful in-vivo optical technologies that can visualize at the cellular level. With this technology we should be able to image the entire knee with the required magnification and conduct cost- and time-efficient

longitudinal studies that eliminate assignment of animals for gene expression patterns at different ages.

Our growth and development data in the double transgenic (DT) mouse provides an initial, semi-quantitative assay of Col1 and Col2 gene expression during embryonic and early post-natal tendon development. We did not measure type I and II collagen production. Future studies should try to quantify type I and II collagen proteins and try to correlate these with gene expression levels and patterns.

Little information is available regarding the pathways that are required for tendon development, nor its regionalization. We need to identify key agonists/antagonists and signaling pathways in both the tendon mid-substance and insertions. This process can help us to identify the genes truly critical in developing a functional tendon. This will allow us to better understand mechanisms of normal development, which nearly always produce functional tendon, and contrast these with adult natural healing, which most often produce a nonfunctional tendon and cell-based healing which produces a somewhat functional tendon. These studies can be performed by combining LCM with microarray techniques. Future studies should be done to identify these candidate markers and signaling pathways.

11.5. Injury Model in the Mouse Tendon

We could not directly explain our biomechanical results based on gene expression patterns. The improvements we observed in biomechanical properties between 4 and 6 weeks were quite encouraging, but my experimental design did not include an assessment of either gene expression or tissue structure at 6 weeks post injury. Future studies should also perform biomechanics after 1 and 2 weeks of healing and also perform histological studies at 6 weeks of healing.

We only investigated one type of injury i.e. the circular punch defect in the mouse tendon. We tried to create a full length defect in the patellar tendon and did not pursue it further for the following reasons. 1) Our techniques did not produce reproducible injuries. 2) Full length defects

led to rather severe limping of the mouse. 3) It was conveyed to us from Dr. Soslowsky that these defects led to spontaneous rupture. Because of all these reasons, full length defects may not be suitable for the mouse tendon. I suggest that we try half length defects. Such defects can be easily created reproducibly by using a rectangular punch and by modifying the backing which slides under the tendon so as to investigate both soft tissue and insertional healing.

Healing of the mid-substance injury did not induce any ECFP expression which is not surprising. This injury model in my opinion does not utilize the potential of our double transgenic mice, the value of which is in the insertion sites where both Col1 and Col2 is expressed. Hence we should be inducing injuries in the insertions of the tendon. Preliminary studies suggest that Col2 is upregulated during healing in insertions and these might provide clues for recreating zonal insertions. I strongly suggest that we perform healing studies of the insertions in the future.

11.6. Translational of Results from Murine to Rabbit Models

As mentioned in the discussion chapter, establishing correlations in TEC behavior across murine and rabbits might be important if we are to more effectively repair tendon injuries in the animal model. Studies performed in lab show that both transgenic and wild type mouse and rabbit cells behave in the same way. For example, tensile stimulation of MSCs increases Col1 gene expression in both species. Also compressive stimulation of chondrocytes increases Col2 gene expression in both species. These results are encouraging and suggest that we can translate results across species.

We need to perform similar experiments in both species and to make the translational of results easy we need to unify our mouse and rabbits protocols. Currently differences exist in the way we make and stimulate mouse and rabbit MSC constructs and these limitations make comparison of results difficult. Hence following the same protocols for both species become important.

11.7. In-vitro Testing System of Tensile Constructs

Our in-vitro mechanical testing protocol and the testing system itself should also be improved. Gripping issues lead to the low aspect ratio ($\approx 2:1$) of our constructs resulting in biaxial strains near the grips. It is not surprising, therefore, that failures frequently initiated near the grips due to these elevated stress values. These grip failures likely altered failure mechanisms and thus failure biomechanical properties due to St. Venant's effects [507]. Therefore future studies should explore methods to eliminate these end effects, such as cutting a "dog-bone" shaped test specimen.

The biomechanical properties of the constructs could also have been affected by the friction in the mechanical testing system. Although the magnitude of the friction was kept very low, the error produced by this friction could have been substantial due to very small forces acting on the construct. We have explored the possibilities of using vertical systems and found that these systems have very little friction. Therefore, I recommend that future studies utilize this testing system to test the tensile properties of constructs.

Bibliography

- 1. Sharma, P. and N. Maffulli, *Basic biology of tendon injury and healing*, in *Surgeon*. 2005. p. 309-316.
- 2. Butler, D.L., et al., *Using functional tissue engineering and bioreactors to mechanically stimulate tissue-engineered constructs.* Tissue Eng Part A, 2009. **15**(4): p. 741-9.
- 3. Kannus, P., *Structure of the tendon connective tissue*. Scandinavian Journal of Medicine and Science in Sports, 2000. **10**(6): p. 312-320.
- 4. Praemer, A., S. Furner, and D. Rice, *Musculoskeletal Condition in the United States*. Second ed. 1999, Parke Ridge, IL: American Academy of Orthopaedic Surgeons. 182.
- 5. Praemer, A., S. Furner, and D. Rice, *Musculoskeletal Condition in the United States*. First ed. 1992, Parke Ridge, IL: American Academy of Orthopaedic Surgeons. 199.
- 6. Cofield, R.H., et al., *Surgical repair of chronic rotator cuff tears. A prospective longterm study.* J Bone Joint Surg Am, 2001. **83-A**(1): p. 71-7.
- 7. Huang, H.H., A.A. Qureshi, and J.J. Biundo Jr., *Sports and other soft tissue injuries, tendinitis, bursitis, and occupation-related syndromes.* Current Opinion in Rheumatology, 2000. **12**(2): p. 150-154.
- Wong, J., V. Barrass, and N. Maffulli, *Quantitative review of operative and nonoperative management of Achilles tendon ruptures*. American Journal of Sports Medicine, 2002. 30(4): p. 565-575.
- 9. Butler, D.L., *Anterior cruciate ligament: Its normal response and replacement.* Journal of Orthopaedic Research, 1989. **7**(6): p. 910-921.
- 10. Kvist, M., Achilles tendon injuries in athletes. Ann Chir Gynaecol, 1991. **80**(2): p. 188-201.
- 11. Robert, L., M. Moczar, and M. Robert, *Biogenesis, maturation and aging of elastic tissue*. Experientia, 1974. **30**(2): p. 211-2.
- 12. O'Brien, M., Anatomy of Tendons, in Tendon Injuries Basic Science and Clinical Medicine, N. Mafully, P. Renstrom, and W. Leadbetter, Editors. 2005, Springer-Verlog: London.
- 13. Malaviya, P., et al., *An in vivo model for load-modulated remodeling in the rabbit flexor tendon.* Journal of Orthopaedic Research, 2000. **18**(1): p. 116-125.
- 14. Benjamin, M., S. Qin, and J.R. Ralphs, *Fibrocartilage associated with human tendons and their pulleys.* Journal of Anatomy, 1995. **187**: p. 625-633.
- 15. Mills, D.K. and J.C. Daniel, *Development of Functional Specializations within the Maturing Rabbit Flexor Digitorum Profundus Tendon*. Connective Tissue Research, 1993. **30**(1): p. 37-57.
- 16. Hoffmann, A. and G. Gross, *Tendon and ligament engineering in the adult organism: mesenchymal stem cells and gene-therapeutic approaches.* Int Orthop, 2007. **31**(6): p. 791-7.
- 17. Goh, J.C., et al., *Tissue-engineering approach to the repair and regeneration of tendons and ligaments*. Tissue Eng, 2003. **9 Suppl 1**: p. S31-44.
- 18. Woo, S., et al., *Structure and function of ligaments and tendons*, in *Basic Orthopaedic Biomechanics and Mechano-Biology*. 2005, Lippincott Williams&Williams.
- 19. Linsenmayer, T.F., et al., *Type V collagen: molecular structure and fibrillar organization of the chicken alpha 1(V) NH2-terminal domain, a putative regulator of corneal fibrillogenesis.* J Cell Biol, 1993. **121**(5): p. 1181-9.

- 20. Birk, D.E. and R. Mayne, *Localization of collagen types I, III and V during tendon development. Changes in collagen types I and III are correlated with changes in fibril diameter.* Eur J Cell Biol, 1997. **72**(4): p. 352-61.
- 21. Liu, S.H., et al., *Collagen in tendon, ligament, and bone healing. A current review.* Clin Orthop Relat Res, 1995(318): p. 265-78.
- 22. Niyibizi, C., et al., *Type V collagen is increased during rabbit medial collateral ligament healing*. Knee Surg Sports Traumatol Arthrosc, 2000. **8**(5): p. 281-5.
- 23. Fukuta, S., et al., *Identification of types II, IX and X collagens at the insertion site of the bovine Achilles tendon*. Matrix Biology, 1998. **17**(1): p. 65-73.
- 24. Niyibizi, C., et al., *Identification and immunolocalization of type X collagen at the ligament-bone interface*. Biochemical and Biophysical Research Communications, 1996. **222**(2): p. 584-589.
- 25. Sagarriga Visconti, C., et al., *Biochemical analysis of collagens at the ligament-bone interface reveals presence of cartilage-specific collagens*. Archives of Biochemistry and Biophysics, 1996. **328**(1): p. 135-142.
- 26. Matyas, J.R., et al., *Stress governs tissue phenotype at the femoral insertion of the rabbit MCL*. Journal of Biomechanics, 1995. **28**(2): p. 147-157.
- 27. Derwin, K.A., et al., *Proteoglycans and glycosaminoglycan fine structure in the mouse tail tendon fascicle*. Journal of Orthopaedic Research, 2001. **19**(2): p. 269-277.
- 28. Frank, C., et al., *Optimisation of the biology of soft tissue repair*. Journal of Science and Medicine in Sport, 1999. **2**(3): p. 190-210.
- 29. Kuhn, K., *The classical collagens: Types I, II, and III,* in *Stucture and function of collagen types.* 1987, Academic Press Inc. p. 1-42.
- 30. Yurchenco, P.D., D.E. Birk, and R.P. Mecham, eds. *Extracellular matrix assembly and structure*. Biology of Extracellular Matrix. 1994, Academic Press: San Diego : . 468 p.
- 31. Orgel, J.P., et al., *The in situ supermolecular structure of type I collagen*. Structure (Camb), 2001. **9**(11): p. 1061-9.
- 32. Best, T.M. and W.E. Garrett, *Basic science of soft tissue: Muscle and tendon*. Orthopaedic Sports Medicine, 1994: p. 25-45.
- 33. Sharma, P. and N. Maffulli, *Biology of tendon injury: Healing, modeling and remodeling.* Journal of Musculoskeletal Neuronal Interactions, 2006. **6**(2): p. 181-190.
- 34. Sharma, P. and N. Maffulli, *Tendon injury and tendinopathy: Healing and repair*. Journal of Bone and Joint Surgery Series A, 2005. **87**(1): p. 187-202.
- 35. Kirkendall, D.T. and W.E. Garrett, *Function and biomechanics of tendons*. Scandinavian Journal of Medicine and Science in Sports, 1997. **7**(2): p. 62-66.
- 36. O'Brien, M., *Functional anatomy and physiology of tendons*. Clinics in Sports Medicine, 1992. **11**(3): p. 505-520.
- 37. Oxlund, H., *Relationships between the biomechanical properties, composition and molecular structure of connective tissues.* Connective tissue research, 1986. **15**(1-2): p. 65-72.
- 38. Carlstedt, C.A. and M. Nordin, *Biomechanics of tendons and ligaments*. Basic Biomechanics of the Musculoskeletal System, 1989: p. 59-74.
- 39. Viidik, A., *Tendons and ligaments*. Extracellular Matrix, 1996. 1: p. 303-327.
- 40. Fyfe, I. and W.D. Stanish, *The use of eccentric training and stretching in the treatment and prevention of tendon injuries.* Clinics in Sports Medicine, 1992. **11**(3): p. 601-624.

- 41. Diamant, J., et al., *Collagen; ultrastructure and its relation to mechanical properties as a function of ageing.* Proceedings of the Royal Society of London. Series B. Biological sciences, 1972. **180**(60): p. 293-315.
- 42. Hess, G.P., et al., *Prevention and treatment of overuse tendon injuries*. Sports Medicine, 1989. **8**(6): p. 371-384.
- 43. Butler, D.L., et al., *Biomechanics of ligaments and tendons*. Exercise and Sport Sciences Reviews, 1978. **6**: p. 125-181.
- 44. Viidik, A., *Functional properties of collagenous tissues*. International Review of Connective Tissue Research, 1973. **Vol. 6**: p. 127-215.
- 45. Zernicke, R.F. and B.J. Loitz, *Exercise-related adaptations in connective tissue*. The Encyclopaedia of Sports Medicine. Strength and Power in Sport, 2002. **3**: p. 93-113.
- 46. Mosler, E., W. Folkhard, and E. Knorzer, *Stress-induced molecular rearrangement in tendon collagen.* Journal of Molecular Biology, 1985. **182**(4): p. 589-596.
- 47. Curwin, S. and W.D. Stanish, *Tendonitis: Its Etiology and Treatment*. Tendinitis: Its Etiology and Treatment, 1984.
- 48. Kastelic, J. and E. Baer, *Deformation in tendon collagen*. Symposia Soc Exp Biol, 1980.
 34: p. 397-435.
- 49. Sasaki, N., et al., *Time-resolved X-ray diffraction from tendon collagen during creep using synchrotron radiation*. Journal of Biomechanics, 1999. **32**(3): p. 285-292.
- 50. Dressler, M.R., et al., *A potential mechanism for age-related declines in patellar tendon biomechanics*. J Orthop Res, 2002. **20**(6): p. 1315-22.
- 51. Awad, H.A., et al., *Repair of patellar tendon injuries using a cell-collagen composite.* J Orthop Res, 2003. **21**(3): p. 420-31.
- 52. Tohyama, H. and K. Yasuda, *Extrinsic cell infiltration and revascularization accelerate mechanical deterioration of the patellar tendon after fibroblast necrosis.* J Biomech Eng, 2000. **122**(6): p. 594-9.
- 53. Lewis, G. and K.M. Shaw, *Tensile properties of human tendo Achillis: effect of donor age and strain rate.* J Foot Ankle Surg, 1997. **36**(6): p. 435-45.
- 54. Hyman, J. and S.A. Rodeo, *Injury and repair of tendons and ligaments*. Physical Medicine and Rehabilitation Clinics of North America, 2000. **11**(2): p. 267-288.
- 55. Lin, T.W., L. Cardenas, and L.J. Soslowsky, *Biomechanics of tendon injury and repair*. Journal of Biomechanics, 2004. **37**(6): p. 865-877.
- 56. Woo, S.L., K.N. An, and C.B. Frank, *Anatomy, biology, and biomechanics of tendon and ligament*. Orthopaedic Basic Science, 2000: p. 581-616.
- 57. Woo, S.L.Y., et al., *Injury and repair of ligaments and tendons*. Annual Review of Biomedical Engineering, 2000. **2**(2000): p. 83-118.
- 58. Silver, I.A., et al., *A clinical and experimental study of tendon injury, healing and treatment in the horse.* Equine veterinary journal. Supplement, 1983(1): p. 1-43.
- Taylor, D.C., et al., *Experimental muscle strain injury. Early functional and structural deficits and the increased risk for reinjury.* American Journal of Sports Medicine, 1993.
 21(2): p. 190-194.
- 60. Walsh, S. and C. Frank, *Two methods of ligament injury: a morphological comparison in rabbit model*. Journal of Surgical Research, 1988. **45**(2): p. 159-166.
- 61. O'Donoghue, D.H., et al., *Repair of knee ligaments in dogs.I. The lateral collateral ligament.* J Bone Joint Surg, 1961. **43**: p. 1167-1178.

- 62. Silva, M.J., et al., *The insertion site of the canine flexor digitorum profundus tendon heals slowly following injury and structure repair.* Journal of Orthopaedic Research, 2002. **20**(3): p. 447-453.
- 63. Gelberman, R.H., et al., *The effect of gap formation at the repair site on the strength and excursion of intrasynovial flexor tendons: An experimental study on the early stages of tendon-healing in dogs.* Journal of Bone and Joint Surgery Series A, 1999. **81**(7): p. 975-982.
- 64. Chan, B.P., et al., *Pyridinoline in relation to ultimate stress of the patellar tendon during healing: An animal study.* Journal of Orthopaedic Research, 1998. **16**(5): p. 597-603.
- 65. Kubota, H., et al., *Effect of motion and tension on injured flexor tendons in chickens*. Journal of Hand Surgery, 1996. **21**(3): p. 456-463.
- 66. Erhard, L., et al., *Reproducible volar partial lacerations in flexor tendons: A new device for biomechanical studies.* Journal of Biomechanics, 2002. **35**(7): p. 999-1002.
- 67. Gomez, M., *The physiology and biochemistry of soft tissue healing*, in *Rehabilitation of the Injured Knee 2nd Edition*. 1995. p. 34-44.
- 68. Montgomery, R.D., *Healing of muscle, ligaments, and tendons.*, in *Seminars in veterinary medicine and surgery (small animal).* 1989. p. 304-311.
- 69. Murphy, P.G., et al., Influence of exogenous growth factors on the synthesis and secretion of collagen types I and III by explants of normal and healing rabbit ligaments., in Biochemistry and cell biology = Biochimie et biologie cellulaire. 1994. p. 403-409.
- 70. Oakes, B.W., *Tissue healing and repair: Tendons and ligaments*, in *Rehabilitation of Sports Injuries: Scientific Basis*. 2003. p. 56-98.
- 71. Gelberman, R.H., et al., *Angiogenesis in healing autogenous flexor-tendon grafts*. Journal of Bone and Joint Surgery Series A, 1992. **74**(8): p. 1207-1216.
- 72. Fenwick, S.A., B.L. Hazleman, and G.P. Riley, *The vasculature and its role in the damaged and healing tendon*. Arthritis Research, 2002. **4**(4): p. 252-260.
- 73. Jozsa, L. and P. Kannus, *Human tendons: Anatomy, physiology, and pathology*. Human Tendons: Anatomy, Physiology, and Pathology, 1997.
- 74. Grinnell, F., *Fibronectin and wound healing*. Journal of Cellular Biochemistry, 1984. **26**(2): p. 107-116.
- 75. Hooley, C.J. and R.E. Cohen, *A model for the creep behaviour of tendon*, in *International Journal of Biological Macromolecules*. 1979. p. 123-132.
- 76. Abrahamsson, S.O., *Matrix metabolism and healing in the flexor tendon. Experimental studies on rabbit tendon.*, in *Scandinavian journal of plastic and reconstructive surgery and hand surgery. Supplementum.* 1991. p. 1-51.
- Woo, S.L. and J.A. Buckwalter, AAOS/NIH/ORS workshop. Injury and repair of the musculoskeletal soft tissues. Savannah, Georgia, June 18-20, 1987. Journal of orthopaedic research : official publication of the Orthopaedic Research Society, 1988. 6(6): p. 907-931.
- 78. Buckwalter, J.A. and E.B. Hunziker, *Healing of bones, cartilages, tendons, and ligaments: A new era.* Lancet, 1996. **348**(SUPPL.2): p. 18.
- 79. Manske, P.R., et al., *Intrinsic flexor-tendon repair. A morphological study in vitro.* Journal of Bone and Joint Surgery Series A, 1984. **66**(3): p. 385-396.
- 80. Potenza, A.D., *Tendon healing within the flexor digital sheath in the dog.* J Bone Joint Surg, 1962. **44** A(1): p. 49-64.

- 81. Gelberman, R.H., P.R. Manske, and J.S. Vande Berg, *Flexor tendon repair in vitro: A comparative histologic study of the rabbit, chicken, dog, and monkey, in Journal of Orthopaedic Research.* 1984. p. 39-48.
- 82. Potenza, A., *Mechanisms of healing of digital flexor tendons*. Hand, 1969. 1: p. 40-41.
- 83. Potenza, A.D., *Critical evaluation of flexor tendon healing and adhesion formation within artificial digital sheaths.* J Bone Joint Surg, 1963. **45** A: p. 1217-1233.
- 84. Potenza, A.D., *Prevention of adhesions to healing digital flexor tendons*. JAMA, 1964. **187**: p. 99-103.
- 85. Strickland, J.W., *Flexor tendons Acute injuries*, in *Green's Operative Hand Surgery*. 1999. p. 1851-1897.
- 86. Lundborg, G., F. Rank, and B. Heinau, *Intrinsic tendon healing*. A new experimental *model*. Scandinavian Journal of Plastic and Reconstructive Surgery, 1985. **19**(2): p. 113-117.
- 87. Mass, D.P. and R.J. Tuel, *Intrinsic healing of the laceration site in human superficialis flexor tendons in vitro*. Journal of Hand Surgery, 1991. **16**(1): p. 24-30.
- 88. Garner, W.L., et al., Autonomous healing of chicken flexor tendons in vitro, in Journal of Hand Surgery. 1988. p. 697-700.
- 89. Manske, P.R., R.H. Gelberman, and P.A. Lesker, *Flexor tendon healing*, in *Hand Clinics*. 1985. p. 25-34.
- 90. Becker, H., et al., *Intrinsic tendon cell proliferation in tissue culture*, in *Journal of Hand Surgery*. 1981. p. 616-619.
- 91. Russell, J.E. and P.R. Manske, *Collagen synthesis during primate flexor tendon repair in vitro*, in *Journal of Orthopaedic Research*. 1990. p. 13-20.
- 92. Lundborg, G., et al., Superficial repair of severed flexor tendons in synovial environment. An experimental, ultrastructural study on cellular mechanisms, in Journal of Hand Surgery. 1980. p. 451-461.
- 93. Ingraham, J.M., R.M. Hauck, and H.P. Ehrlich, *Is the tendon embryogenesis process resurrected during tendon healing?*, in *Plastic and Reconstructive Surgery*. 2003. p. 844-854.
- 94. Fujita, M., S. Hukuda, and Y. Doida, *Experimental study of intrinsic healing of the flexor tendon: Collagen synthesis of the cultured flexor tendon cells of the canine*, in *Journal of the Japanese Orthopaedic Association*. 1992. p. 326-333.
- 95. Tokita, Y., A. Yamaya, and Y. Yabe, An experimental study on the repair and restoration of gliding function after digital flexor tendon injury. Part I: Repair of the sutured digital flexor tendon within the digital sheath (Japanese), in Journal of the Japanese Orthopaedic Association. 1974. p. 107-127.
- 96. Gelberman, R.H., P.R. Manske, and W.H. Akeson, *Kappa Delta Award paper: Flexor tendon repair*, in *Journal of Orthopaedic Research*. 1986. p. 119-128.
- 97. Mast, B.A., et al., *In vivo degradation of fetal wound hyaluronic acid results in increased fibroplasia, collagen deposition, and neovascularization, in Plastic and Reconstructive Surgery.* 1992. p. 503-509.
- 98. Uhthoff, H.K. and K. Sarkar, Surgical repair of rotator cuff ruptures. The importance of the subacromial bursa, in Journal of Bone and Joint Surgery Series B. 1991. p. 399-401.
- 99. Hampson, K., et al., *Tendon Tissue Engineering*, in *Topics in Tissue Engineering*, Vol. 4, N. Ashammakhi, R. Reis, and F. Chiellini, Editors. 2008.

- 100. Taras, J.S. and M.J. Lamb, *Treatment of flexor tendon injuries: Surgeons' perspective*, in *Journal of Hand Therapy*. 1999. p. 141-151.
- 101. Wang, E.D., Tendon repair, in Journal of Hand Therapy. 1998. p. 105-110.
- 102. Mullaney, M.J., et al., *Weakness in end-range plantar flexion after achilles tendon repair*. American Journal of Sports Medicine, 2006. **34**(7): p. 1120-1125.
- 103. Van Der Linden-Van Der Zwaag, H.M.J., R.G.H.H. Nelissen, and J.B. Sintenie, *Results of surgical versus non-surgical treatment of Achilles tendon rupture*. International Orthopaedics, 2004. **28**(6): p. 370-373.
- 104. Butler, D.L. and H.A. Awad, *Perspectives on cell and collagen composites for tendon repair*. Clinical Orthopaedics and Related Research, 1999(367 SUPPL.): p. S324-S332.
- 105. Holden, J.P., et al., *Biomechanics of fascia lata ligament replacements: Early postoperative changes in the goat.* Journal of Orthopaedic Research, 1988. **6**(5): p. 639-647.
- 106. Noyes, F.R., D.L. Butler, and E.S. Grood, *Biomechanical analysis of human ligament grafts used in knee-ligament repairs and reconstructions*. Journal of Bone and Joint Surgery Series A, 1984. **66**(3): p. 344-352.
- 107. Juergensen, K., J.C. Edwards, and R.P. Jakob, *Positioning of the posterior cruciate ligament*. Knee Surgery, Sports Traumatology, Arthroscopy, 1994. **2**(3): p. 133-137.
- 108. Bonamo, J.J., R.M. Krinick, and A.A. Sporn, *Rupture of the patellar ligament after use of its central third for anterior cruciate reconstruction. A report of two cases.* Journal of Bone and Joint Surgery Series A, 1984. **66**(8): p. 1294-1297.
- 109. Hughston, J.C., *Complications of anterior cruciate ligament surgery*. Orthopedic Clinics of North America, 1985. **16**(2): p. 237-240.
- 110. Beynnon, B.D., et al., *Biomechanical assessment of the healing response of the rabbit patellar tendon after removal of its central third*. American Journal of Sports Medicine, 1995. **23**(4): p. 452-457.
- Proctor, C., D. Jackson, and T. Simon, *Characterization of the repair tissue after removal of the central one-third of the patellar ligament. An experimental study in a goat model.* JOURNAL OF BONE AND JOINT SURGERY. AMERICAN VOLUME, 1997. **79**(Jul): p. 997-1006.
- 112. Burks, R.T., R.C. Haut, and R.L. Lancaster, *Biomechanical and histological observations* of the dog patellar tendon after removal of its central one-third. American Journal of Sports Medicine, 1990. **18**(2): p. 146-153.
- 113. Fromm, B., et al., *Reconstruction of the anterior cruciate ligament with a cyropreserved ACL allograft - A microangiographic and immunohistochemical study in rabbits.* International Orthopaedics, 1996. **20**(6): p. 378-382.
- 114. Goertzen, M.J., et al., Anterior cruciate ligament reconstruction using cryopreserved irradiated bone-ACL-bone-allograft transplants. Knee Surgery, Sports Traumatology, Arthroscopy, 1994. 2(3): p. 150-157.
- 115. Fideler, B.M., et al., *Effects of gamma irradiation on the human immunodeficiency virus.* A study in frozen human bone-patellar ligament-bone grafts obtained from infected cadavera. Journal of Bone and Joint Surgery - Series A, 1994. **76**(7): p. 1032-1035.
- 116. Salehpour, A., et al., *Dose-dependent response of gamma irradiation on mechanical properties and related biochemical composition of goat bone-patellar tendon-bone allografts.* J Orthop Res, 1995. **13**(6): p. 898-906.

- 117. Rasmussen, T.J., et al., *The effects of 4 Mrad of* \hat{I}^3 *irradiation on the initial mechanical properties of bone-patellar tendon-bone grafts.* Arthroscopy, 1994. **10**(2): p. 188-197.
- 118. Roberts, T.S., et al., Anterior cruciate ligament reconstruction using freeze-dried, ethylene oxide-sterilized, bone-patellar tendon-bone allografts: Two year results in thirty-six patients. American Journal of Sports Medicine, 1991. **19**(1): p. 35-41.
- 119. Lima, E.G., et al., *Functional tissue engineering of chondral and osteochondral constructs*. Biorheology, 2004. **41**(3-4): p. 577-90.
- 120. Maletius, W. and K. Messner, *The effect of partial meniscectomy on the long-term prognosis of knees with localized, severe chondral damage. A twelve- to fifteen-year followup.* Am J of Sports Med, 1996. **24**: p. 258-62.
- 121. Bolton, C.W. and W.C. Bruchman, *The GORE-TEX expanded polytetrafluoroethylene prosthetic ligament. An in vitro and in vivo evaluation.* Clin Orthop, 1985(196): p. 202-13.
- 122. Woods, G.A., P.A. Indelicato, and T.J. Prevot, *The Gore-Tex anterior cruciate ligament prosthesis. Two versus three year results.* Am J Sports Med, 1991. **19**(1): p. 48-55.
- 123. Amis, A.A., et al., Proceedings of the ESSKA Scientific Workshop on Reconstruction of the Anterior and Posterior Cruciate Ligaments. Knee Surg Sports Traumatol Arthrosc, 1994. 2(3): p. 124-32.
- 124. Beris, A.E., et al., *Two-stage flexor tendon reconstruction in zone II using a silicone rod and a pedicled intrasynovial graft.* Journal of Hand Surgery, 2003. **28**(4): p. 652-660.
- 125. Mahoney, J.L., L.G. Farkas, and W.K. Lindsay, *Quality of tendon graft healing in silastic pseudosheaths: breaking strength studies*. Surgical Forum, 1976. Vol.27: p. 572-573.
- 126. Bader, K.F. and J.W. Curton, *A successful silicone tendon prosthesis*. Archives of Surgery, 1968. **97**(3): p. 406-411.
- 127. Makisalo, S., et al., *Reconstruction of anterior cruciate ligament with carbon fiber. An experimental study on pigs.* Am J Sports Med, 1988. **16**(6): p. 589-93.
- 128. Elliot, D., et al., *The rupture rate of acute flexor tendon repairs mobilized by the controlled active motion regimen.* J Hand Surg [Br], 1994. **19**(5): p. 607-12.
- 129. Jenkins, D.H. and B. McKibbin, *The role of flexible carbon-fibre implants as tendon and ligament substitutes in clinical practice. A preliminary report.* J Bone Joint Surg Br, 1980. **62-B**(4): p. 497-9.
- 130. Weiss, A.B., et al., *Ligament replacement with an absorbable copolymer carbon fiber scaffold--early clinical experience*. Clin Orthop, 1985(196): p. 77-85.
- 131. Kitsis, C.K., et al., *Controlled active motion following primary flexor tendon repair: a prospective study over 9 years.* J Hand Surg [Br], 1998. **23**(3): p. 344-9.
- 132. Arnoczky, S.P., R.F. Warren, and J.P. Minei, *Replacement of the anterior cruciate ligament using a synthetic prosthesis. An evaluation of graft biology in the dog.* Am J Sports Med, 1986. **14**(1): p. 1-6.
- 133. Giorgetti, M., E. Giannessi, and M.P. Ricciardi, *Expanded polytetrafluoroethylene as tendon replacement: an experimental study in chickens.* Scand J Plast Reconstr Surg Hand Surg, 2001. **35**(1): p. 23-7.
- 134. Kato, Y.P., et al., *Regeneration of Achilles tendon with a collagen tendon prosthesis*. *Results of a one-year implantation study*. J Bone Joint Surg Am, 1991. **73**(4): p. 561-74.
- 135. Goldstein, J.D., et al., *Development of a reconstituted collagen tendon prosthesis. A preliminary implantation study.* J Bone Joint Surg Am, 1989. **71**(8): p. 1183-91.

- 136. Wredmark, T. and B. Engstrom, *Five-year results of anterior cruciate ligament reconstruction with the Stryker Dacron high-strength ligament*. Knee Surg Sports Traumatol Arthrosc, 1993. **1**(2): p. 71-5.
- 137. Harner, C.D., et al., *Detailed analysis of patients with bilateral anterior cruciate ligament injuries*. Am J Sports Med, 1994. **22**(1): p. 37-43.
- 138. Corsetti, J.R. and D.W. Jackson, *Failure of anterior cruciate ligament reconstruction: the biologic basis.* Clin Orthop, 1996(325): p. 42-9.
- 139. Dennis, J.E., et al., Osteogenesis in marrow-derived mesenchymal cell porous ceramic composites transplanted subcutaneously: effect of fibronectin and laminin on cell retention and rate of osteogenic expression. Cell Transplant, 1992. **1**(1): p. 23-32.
- 140. Campisi, J., Mechanisms of aging that might contribute to musculoskeletal impairment: Diminished cell proliferative capacity, in Musculoskeletal soft-tissue aging: Impact on mobility. AAOS Workshop, 1994: p. 73-78.
- 141. Lanza, R.P., R.S. Langer, and J. Vacanti, eds. *Principles of tissue engineering*. 2nd ed., ed. R.P. Lanza, R.S. Langer, and J. Vacanti. 2000, Academic Press: San Diego. 995.
- 142. Cao, Y., et al., Bridging tendon defects using autologous tenocyte engineered tendon in a hen model, in Plastic and Reconstructive Surgery. 2002. p. 1280-1289.
- 143. Cooper Jr., J.A., et al., *Biomimetic tissue-engineered anterior cruciate ligament replacement*, in *Proceedings of the National Academy of Sciences of the United States of America*. 2007. p. 3049-3054.
- 144. Huang, D., G. Balian, and A.B. Chhabra, *Tendon Tissue Engineering and Gene Transfer: The Future of Surgical Treatment*, in *Journal of Hand Surgery*. 2006. p. 693-704.
- 145. Cooper Jr., J.A., et al., Evaluation of the anterior cruciate ligament, medial collateral ligament, achilles tendon and patellar tendon as cell sources for tissue-engineered ligament, in Biomaterials. 2006. p. 2747-2754.
- 146. Badami, A.S., et al., *Effect of fiber diameter on spreading, proliferation, and differentiation of osteoblastic cells on electrospun poly(lactic acid) substrates, in Biomaterials.* 2006. p. 596-606.
- 147. Ouyang, H.W., et al., *The efficacy of bone marrow stromal cell-seeded knitted PLGA fiber scaffold for Achilles tendon repair.* Ann N Y Acad Sci, 2002. **961**: p. 126-9.
- 148. Ouyang, H.W., et al., *Knitted poly-lactide-co-glycolide scaffold loaded with bone marrow stromal cells in repair and regeneration of rabbit Achilles tendon*. Tissue Eng, 2003. 9(3): p. 431-9.
- 149. Ouyang, H.W., et al., *Knitted poly-lactide-co-glycolide scaffold loaded with bone marrow stromal cells in repair and regeneration of rabbit achilles tendon*, in *Tissue Engineering*. 2003. p. 431-439.
- 150. Young, R.G., et al., *Use of mesenchymal stem cells in a collagen matrix for Achilles tendon repair.* J Orthop Res, 1998. **16**(4): p. 406-13.
- 151. Tuan, R.S., G. Boland, and R. Tuli, *Adult mesenchymal stem cells and cell-based tissue engineering*. Arthritis Research and Therapy, 2003. **5**(1): p. 32-45.
- 152. Harris, M.T., et al., Mesenchymal stem cells used for rabbit tendon repair can form ectopic bone and express alkaline phosphatase activity in constructs, in Journal of Orthopaedic Research. 2004. p. 998-1003.
- 153. Pittenger, M.F., et al., *Multilineage potential of adult human mesenchymal stem cells*, in *Science*. 1999. p. 143-147.

- 154. Wolfman, N.M., et al., *Ectopic induction of tendon and ligament in rats by growth and differentiation factors 5, 6, and 7, members of the TGF-Î² gene family, in Journal of Clinical Investigation.* 1997. p. 321-330.
- 155. Altaian, G., R. Horan, and I. Martin, *Cell differentiation by mechanical stress*, in *FASEB* J. 2001. p. 270.
- 156. Butler, D.L., et al., *Functional tissue engineering for tendon repair: A multidisciplinary strategy using mesenchymal stem cells, bioscaffolds, and mechanical stimulation.* Journal of Orthopaedic Research, 2008. **26**(1): p. 1-9.
- 157. West, J., et al., Characterization of In-vivo Achilles Tendon Forces in Rabbits during Treadmill Locomotion at Varying Speeds and Inclinations. J Biomech, 2004. **37**(11): p. 1647-1653.
- 158. Juncosa, N., et al., *In vivo forces used to develop design parameters for tissue engineered implants for rabbit patellar tendon repair.* J Biomech, 2003. **36**(4): p. 483-8.
- 159. Butler, D.L., S.A. Goldstein, and F. Guilak, *Functional tissue engineering: the role of biomechanics*. J Biomech Eng, 2000. **122**(6): p. 570-5.
- 160. Guilak, F., et al., Functional Tissue Engineering. 2003, New York: Springer-Verlag.
- 161. Butler, D.L., et al., *Functional Tissue Engineering Parameters toward Designing Repair and Replacement Strategies.* Clin Orthop Rel Res, 2004. **427S**(427 Suppl): p. S190-9.
- 162. Chokalingam, K., et al., 3D-In vitro Effects of Compression and Time in Culture on Aggregate Modulus and on Gene Expression and Protein content of Collagen Type II in Murine Chondrocytes. Tissue Eng Part A, 2009.
- 163. Chokalingam, K., et al., *Tensile Stimulation of Murine Stem Cell-Collagen Sponge Constructs Increases Collagen Type I Gene Expression and Linear Stiffness.* Tissue Eng Part A, 2009.
- 164. Huang, J., L.R. Ballou, and K.A. Hasty, *Cyclic equibiaxial tensile strain induces both anabolic and catabolic responses in articular chondrocytes.* Gene, 2007. **404**(1-2): p. 101-109.
- 165. Hunter, C., J.K. Mouw, and M.E. Levenston, *Dynamic compression of chondrocyte-seeded fibrin gels: Effects on matrix accumulation and mechanical stiffness*. Osteoarthritis and Cartilage, 2004. **12**(2): p. 117-130.
- 166. Hunter, C.J., et al., Mechanical compression alters gene expression and extracellular matrix synthesis by chondrocytes cultured in collagen I gels, in Biomaterials. 2002. p. 1249-1259.
- 167. Juncosa-Melvin, N., et al., *Mechanical stimulation increases collagen type I and collagen type III gene expression of stem cell-collagen sponge constructs for patellar tendon repair*. Tissue Engineering, 2007. **13**(6): p. 1219-1226.
- 168. Kisiday, J.D., et al., *Effects of dynamic compressive loading on chondrocyte biosynthesis in self-assembling peptide scaffolds*. Journal of Biomechanics, 2004. **37**(5): p. 595-604.
- 169. Mauck, R.L., et al., Synergistic action of growth factors and dynamic loading for articular cartilage tissue engineering. Tissue Engineering, 2003. **9**(4): p. 597-611.
- 170. Mauck, R.L., et al., *Functional tissue engineering of articular cartilage through dynamic loading of chondrocyte-seeded agarose gels.* J Biomech Eng, 2000. **122**(3): p. 252-60.
- 171. Juncosa-Melvin, N., et al., *Effects of mechanical stimulation on the biomechanics and histology of stem cell-collagen sponge constructs for rabbit patellar tendon repair.* Tissue Engineering, 2006. **12**(8): p. 2291-2300.

- 172. Mauck, R.L., et al., *Influence of seeding density and dynamic deformational loading on the developing structure/function relationships of chondrocyte-seeded agarose hydrogels*. Ann Biomed Eng, 2002. **30**(8): p. 1046-56.
- 173. Altman, G.H., et al., *Cell differentiation by mechanical stress*. Faseb J, 2002. **16**(2): p. 270-2.
- 174. Grant, T.D., et al., *Col2-GFP reporter marks chondrocyte lineage and chondrogenesis during mouse skeletal development*. Developmental Dynamics, 2000. **218**(2): p. 394-400.
- 175. Harvey, K.J., D. Lukovic, and D.S. Ucker, *Membrane-targeted green fluorescent protein reliably and uniquely marks cells through apoptotic death*. Cytometry, 2001. **43**(4): p. 273-278.
- 176. Kalajzic, I., et al., *Stage specific inhibition of osteoblast lineage differentiation by FGF2 and noggin.* Journal of Cellular Biochemistry, 2003. **88**(6): p. 1168-1176.
- 177. Ma, X., et al., *The Ly-6A (Sca-1) GFP transgene is expressed in all adult mouse hematopoietic stem cells.* Stem Cells, 2002. **20**(6): p. 514-521.
- 178. Mignone, J.L., et al., *Neural Stem and Progenitor Cells in Nestin-GFP Transgenic Mice*. Journal of Comparative Neurology, 2004. **469**(3): p. 311-324.
- 179. Tsumaki, N., et al., Role of CDMP-1 in skeletal morphogenesis: Promotion of mesenchymal cell recruitment and chondrocyte differentiation. Journal of Cell Biology, 1999. **144**(1): p. 161-173.
- 180. Beredjiklian, P.K., *Biologic aspects of flexor tendon laceration and repair*, in *Journal of Bone and Joint Surgery Series A*. 2003. p. 539-550.
- 181. Dourte, L.M., A.F. Kuntz, and L.J. Soslowsky, *Twenty-five years of tendon and ligament research*. Journal of Orthopaedic Research, 2008. **26**(10): p. 1297-1305.
- 182. Gelberman, R.H., et al., *The early stages of flexor tendon healing: A morphologic study of the first fourteen days*, in *Journal of Hand Surgery*. 1985. p. 776-784.
- 183. Frank, C., S.L.Y. Woo, and D. Amiel, *Medial collateral ligament healing. A multidisciplinary assessment in rabbits*, in *American Journal of Sports Medicine*. 1983. p. 379-389.
- 184. Frank, C., et al., *Collagen fibril diameters in the healing adult rabbit medial collateral ligament.*, in *Connective Tissue Research.* 1992. p. 251-263.
- 185. Frank, C., D. Mcdonald, and N. Shrive, Collagen fibril diameters in the rabbit medial collateral ligament scar: A longer term assessment, in Connective Tissue Research. 1997. p. 261-269.
- 186. Ingber, D.E., et al., *Tissue engineering and developmental biology: Going biomimetic.* Tissue Engineering, 2006. **12**(12): p. 3265-3283.
- 187. Verheyen, E.M., *Opposing Effects of Wnt and MAPK on BMP/Smad Signal Duration*. Developmental Cell, 2007. **13**(6): p. 755-756.
- 188. Schweitzer, R., et al., *Analysis of the tendon cell fate using Scleraxis, a specific marker for tendons and ligaments.* Development, 2001. **128**(19): p. 3855-3866.
- 189. Lincoln, J., A.W. Lange, and K.E. Yutzey, *Hearts and bones: Shared regulatory mechanisms in heart valve, cartilage, tendon, and bone development.* Developmental Biology, 2006. **294**(2): p. 292-302.
- 190. Mendias, C.L., K.I. Bakhurin, and J.A. Faulkner, *Tendons of myostatin-deficient mice are small, brittle, and hypocellular*. Proceedings of the National Academy of Sciences of the United States of America, 2008. **105**(1): p. 388-393.

- 191. Lima, E.G., et al., The beneficial effect of delayed compressive loading on tissueengineered cartilage constructs cultured with $TGF-\hat{I}^23$. Osteoarthritis and Cartilage, 2007. **15**(9): p. 1025-1033.
- 192. Mauck, R.L., et al., *Regulation of cartilaginous ECM gene transcription by chondrocytes and MSCs in 3D culture in response to dynamic loading*. Biomechanics and Modeling in Mechanobiology, 2007. **6**(1-2): p. 113-125.
- 193. Démarteau, O., et al., *Dynamic compression of cartilage constructs engineered from expanded human articular chondrocytes*. Biochemical and Biophysical Research Communications, 2003. **310**(2): p. 580-588.
- 194. Davisson, T., et al., *Static and dynamic compression modulate matrix metabolism in tissue engineered cartilage*. Journal of Orthopaedic Research, 2002. **20**(4): p. 842-848.
- 195. Mauck, R.L., et al., *Influence of seeding density and dynamic deformational loading on the developing structure/function relationships of chondrocyte-seeded agarose hydrogels.* Annals of Biomedical Engineering, 2002. **30**(8): p. 1046-1056.
- 196. Lee, D.A., et al., *The influence of mechanical loading on isolated chondrocytes seeded in agarose constructs.* Biorheology, 2000. **37**(1-2): p. 149-161.
- 197. Buschmann, M.D., et al., *Mechanical compression modulates matrix biosynthesis in chondrocyte/agarose culture*. Journal of Cell Science, 1995. **108**(4): p. 1497-1508.
- 198. Angele, P., et al., Cyclic hydrostatic pressure enhances the chondrogenic phenotype of human mesenchymal progenitor cells differentiated in vitro. Journal of Orthopaedic Research, 2003. **21**(3): p. 451-457.
- 199. Nicoll, S.B., et al., Modulation of proteoglycan and collagen profiles in human dermal fibroblasts by high density micromass culture and treatment with lactic acid suggests change to a chondrogenic phenotype. Connective Tissue Research, 2001. **42**(1): p. 59-69.
- 200. Angele, P., et al., Cyclic, mechanical compression enhances chondrogenesis of mesenchymal progenitor cells in tissue engineering scaffolds. Biorheology, 2004. 41(3-4): p. 335-346.
- 201. Elder, S.H., et al., *Chondrocyte differentiation is modulated by frequency and duration of cyclic compressive loading*. Annals of Biomedical Engineering, 2001. **29**(6): p. 476-482.
- 202. Walker, E.A., et al., *Effect of compressive loading on chondrocyte differentiation in agarose cultures of chick limb-bud cells*. Journal of Orthopaedic Research, 2000. 18(1): p. 78-86.
- 203. Huang, C.-Y.C., et al., *Effects of cyclic compressive loading on chondrogenesis of rabbit bone-marrow derived mesenchymal stem cells.* Stem Cells, 2004. **22**(3): p. 313-323.
- 204. Mauck, R.L., X. Yuan, and R.S. Tuan, *Chondrogenic differentiation and functional maturation of bovine mesenchymal stem cells in long-term agarose culture*. Osteoarthritis and Cartilage, 2006. **14**(2): p. 179-189.
- 205. He, Y., et al., *Compression and tension: Differential effects on matrix accumulation by periodontal ligaments fibroblasts in vitro.* Connective Tissue Research, 2004. **45**(1): p. 28-39.
- 206. Shim, J.W. and S.H. Elder, *Influence of cyclic hydrostatic pressure on fibrocartilaginous metaplasia of achilles tendon fibroblasts*. Biomechanics and Modeling in Mechanobiology, 2006. **5**(4): p. 247-252.
- 207. Vanderploeg, E.J., et al., Oscillatory tension differentially modulates matrix metabolism and cytoskeletal organization in chondrocytes and fibrochondrocytes. Journal of Biomechanics, 2004. **37**(12): p. 1941-1952.

- 208. Butler, D., Kappa Delta Award paper. Anterior cruciate ligament: its normal response and replacement. J Orthop Res, 1989. 7: p. 910-21.
- 209. Ferretti, A., Epidemiology of Jumper's knee. Sports Medicine, 1986. 3(4): p. 289-295.
- 210. Kannus, P., *Tendons--a source of major concern in competitive and recreational athletes.* Scandinavian journal of medicine & science in sports, 1997. **7**(2): p. 53-54.
- 211. Kannus, P., *Tendon pathology: Basic science and clinical applications*. Sports Excercise and Injury, 1997. **3**(2): p. 62-75.
- 212. Lian, Ø.B., L. Engebretsen, and R. Bahr, *Prevalence of jumper's knee among elite athletes from different sports: A cross-sectional study.* American Journal of Sports Medicine, 2005. **33**(4): p. 561-567.
- 213. Cree, C., et al., Bilateral patellar tendon ruptures: A missed diagnosis : Cse report and literature review. Knee Surgery, Sports Traumatology, Arthroscopy, 2007. **15**(11): p. 1350-1354.
- 214. Katz, T., et al., *Bilateral simultaneous rupture of the quadriceps tendon in an adult amateur tennis player*. Journal of Clinical Rheumatology, 2006. **12**(1): p. 32-33.
- 215. Ozkan, C., et al., *Bilateral dislocation of the knee with rupture of both patellar tendons. A case report.* Knee, 2006. **13**(4): p. 333-336.
- 216. Saidi, H., et al., *Bilateral patellar tendon rupture in a footballer. A case report.* Rupture bilate?rale du tendon patellaire chez un footballeur. A? propos d'un cas, 2008. **25**(1): p. 54-57.
- 217. Siwek, C.W. and J.P. Rao, *Ruptures of the extensor mechanism of the knee joint*. Journal of Bone and Joint Surgery Series A, 1981. **63**(6): p. 932-937.
- 218. Webb, L.X. and E.B. Toby, *Bilateral rupture of the patella tendon in an otherwise healthy male patient following minor trauma*. Journal of Trauma, 1986. **26**(11): p. 1045-1048.
- 219. Juergensen, K., J.C. Edwards, and R.P. Jakob, *Positioning of the posterior cruciate ligament*. Knee Surg Sports Traumatol Arthrosc, 1994. **2**(3): p. 133-7.
- Jackson, D.W., et al., A comparison of patellar tendon autograft and allograft used for anterior cruciate ligament reconstruction in the goat model. Am J Sports Med, 1993.
 21(2): p. 176-85.
- 221. Vasseur, P.B., et al., *Replacement of the anterior cruciate ligament with a bone-ligamentbone anterior cruciate ligament allograft in dogs*. Clin Orthop, 1987(219): p. 268-77.
- 222. Konrath, G.A., et al., *Outcomes following repair of quadriceps tendon ruptures*. J Orthop Trauma, 1998. **12**(4): p. 273-9.
- 223. Kartus, J., et al., Serial magnetic resonance imaging of the donor site after harvesting the central third of the patellar tendon. A prospective study of 37 patients after arthroscopic anterior cruciate ligament reconstruction. Knee Surg Sports Traumatol Arthrosc, 1999. 7(1): p. 20-4.
- Burks, R.T., R.C. Haut, and R.L. Lancaster, *Biomechanical and histological observations* of the dog patellar tendon after removal of its central one-third. Am J Sports Med, 1990. 18(2): p. 146-53.
- 225. Ng, G.Y., et al., *Biomechanics of patellar tendon autograft for reconstruction of the anterior cruciate ligament in the goat: three-year study.* J Orthop Res, 1995. **13**(4): p. 602-8.

- 226. Drez, D.J., Jr., et al., Anterior cruciate ligament reconstruction using bone-patellar tendon- bone allografts. A biological and biomechanical evaluation in goats. Am J Sports Med, 1991. **19**(3): p. 256-63.
- 227. Kasperczyk, W.J., et al., *Staging of patellar tendon autograft healing after posterior cruciate ligament reconstruction. A biomechanical and histological study in a sheep model.* Clin Orthop, 1993(286): p. 271-82.
- 228. Kvist, M., Achilles tendon injuries in athletes. Sports Med, 1994. 18(3): p. 173-201.
- 229. Rupp, S., S. Tempelhof, and E. Fritsch, *Ultrasound of the Achilles tendon after surgical repair: morphology and function.* Br J Radiol, 1995. **68**(809): p. 454-8.
- 230. Milgrom, C., et al., *Rotator-cuff changes in asymptomatic adults. The effect of age, hand dominance and gender.* J Bone Joint Surg Br, 1995. **77**(2): p. 296-8.
- 231. Yamanaka, K. and T. Matsumoto, *The joint side tear of the rotator cuff. A followup study by arthrography*. Clin Orthop, 1994(304): p. 68-73.
- 232. Miniaci, A. and M. MacLeod, *Transfer of the latissimus dorsi muscle after failed repair* of a massive tear of the rotator cuff. A two to five-year review. J Bone Joint Surg Am, 1999. **81**(8): p. 1120-7.
- 233. Celli, L., et al., *Transplantation of teres major muscle for infraspinatus muscle in irreparable rotator cuff tears.* J Shoulder Elbow Surg, 1998. **7**(5): p. 485-90.
- 234. Galatz, L.M., et al., *The outcome and repair integrity of completely arthroscopically repaired large and massive rotator cuff tears.* J Bone Joint Surg Am, 2004. **86-A**(2): p. 219-24.
- 235. Gerber, C., B. Fuchs, and J. Hodler, *The results of repair of massive tears of the rotator cuff.* J Bone Joint Surg Am, 2000. **82**(4): p. 505-15.
- 236. Jost, B., et al., *Clinical outcome after structural failure of rotator cuff repairs*. J Bone Joint Surg Am, 2000. **82**(3): p. 304-14.
- 237. Klepps, S., et al., *Prospective Evaluation of the Effect of Rotator Cuff Integrity on the Outcome of Open Rotator Cuff Repairs*. Am J Sports Med, 2004. **32**(7): p. 1716-1722.
- 238. Worland, R.L., et al., *Repair of massive rotator cuff tears in patients older than 70 years.* J Shoulder Elbow Surg, 1999. **8**(1): p. 26-30.
- 239. Hughston, J., R. Haut, and R. Lancaster, *Biomechanical and histological observations of the dog patellar tendon after removal of its central third*. Am J Sports Med, 1990. **18**: p. 146-153.
- 240. Beynnon, B., et al., *Biomechanical assessment of the healing response of the rabbit patellar tendon after removal of its central third.* Am J Sports Med, 1995. **23**(Jul-Aug): p. 452-7.
- 241. Drez, D.J., Jr., et al., Anterior cruciate ligament reconstruction using bone-patellar tendon-bone allografts. A biological and biomechanical evaluation in goats. Am J Sports Med, 1991. **19**(3): p. 256-63.
- 242. Bruns, J., et al., *Achilles tendon rupture: experimental results on spontaneous repair in a sheep-model.* Knee Surg Sports Traumatol Arthrosc, 2000. **8**(6): p. 364-9.
- 243. Beredjiklian, P.K., et al., *Regenerative versus reparative healing in tendon: A study of biomechanical and histological properties in fetal sheep.* Annals of Biomedical Engineering, 2003. **31**(10): p. 1143-1152.
- 244. Langer, R. and J.P. Vacanti, *Tissue engineering*. Science, 1993. 260(5110): p. 920-926.
- 245. Juncosa-Melvin, N., et al. Effect of mechanical stimulation on the biomechanics of stem cell Collagen sponge constructs for patellar tendon repair. in Proceedings of the ASME Summer Bioengineering Conference 2007, SBC 2007. 2007.
- 246. Juncosa-Melvin, N., et al., *Effects of cell-to-collagen ratio in stem cell-seeded constructs for Achilles tendon repair*. Tissue Engineering, 2006. **12**(4): p. 681-689.
- 247. Juncosa-Melvin, N., et al., *The effect of autologous mesenchymal stem cells on the biomechanics and histology of gel-collagen sponge constructs used for rabbit patellar tendon repair.* Tissue Engineering, 2006. **12**(2): p. 369-379.
- 248. Juncosa-Melvin, N., et al., *Effects of cell-to-collagen ratio in mesenchymal stem cell-seeded implants on tendon repair biomechanics and histology*. Tissue Engineering, 2005. **11**(3-4): p. 448-457.
- 249. Lin, T.W., et al., *Tendon healing in interleukin-4 and interleukin-6 knockout mice*. Journal of Biomechanics, 2006. **39**(1): p. 61-69.
- 250. Awad, H.A., et al., *Repair of patellar tendon injuries using a cell-collagen composite*. Journal of Orthopaedic Research, 2003. **21**(3): p. 420-431.
- 251. Awad, H.A., et al., *In vitro characterization of mesenchymal stem cell-seeded collagen scaffolds for tendon repair: Effects of initial seeding density on contraction kinetics.* Journal of Biomedical Materials Research, 2000. **51**(2): p. 233-240.
- 252. Galatz, L., et al., *Development of the supraspinatus tendon-to-bone insertion: Localized expression of extracellular matrix and growth factor genes.* Journal of Orthopaedic Research, 2007. **25**(12): p. 1621-1628.
- 253. Galatz, L.M., et al., *Characteristics of the rat supraspinatus tendon during tendon-tobone healing after acute injury.* Journal of Orthopaedic Research, 2006. **24**(3): p. 541-550.
- 254. Oshiro, W., et al., *Flexor tendon healing in the rat: A histologic and gene expression study*. Journal of Hand Surgery, 2003. **28**(5): p. 814-823.
- 255. Berglund, M., et al., *Patterns of mRNA Expression for Matrix Molecules and Growth Factors in Flexor Tendon Injury: Differences in the Regulation Between Tendon and Tendon Sheath.* Journal of Hand Surgery, 2006. **31**(8): p. 1279-1287.
- 256. Loiselle, A.E., et al., *Remodeling of murine intrasynovial tendon adhesions following injury: MMP and neotendon gene expression.* J Orthop Res, 2008.
- 257. Awad, H.A., et al., *Autologous mesenchymal stem cell-mediated repair of tendon*. Tissue Eng, 1999. **5**(3): p. 267-77.
- 258. Praemer, A., S. Furner, and D. Rice, *Musculoskeletal Condition in the United States*. Second ed. 1999, Park Ridge, IL: American Academy of Orthopaedic Surgeons. 182.
- 259. Kuwada, G.T., *An update on repair of Achilles tendon rupture. Acute and delayed.* Journal of the American Podiatric Medical Association, 1999. **89**(6): p. 302-306.
- 260. Winters, S.C., et al., Suture methods for flexor tendon repair. A biomechanical analysis during the first six weeks following repair. Annales de Chirurgie de la Main et du Membre Superieur, 1997. **16**(3): p. 229-234.
- 261. Brown Jr., C.H., et al., *The biomechanics of interference screw fixation of patellar tendon anterior cruciate ligament grafts.* American Journal of Sports Medicine, 1993. **21**(6): p. 880-886.
- 262. Noyes, F.R. and S.D. Barber-Westin, *Anterior cruciate ligament reconstruction with autogenous patellar tendon graft in patients with articular cartilage damage.* American Journal of Sports Medicine, 1997. **25**(5): p. 626-634.

- Kusayama, T., et al., Anatomical and biomechanical characteristics of human meniscofemoral ligaments. Knee Surgery, Sports Traumatology, Arthroscopy, 1994. 2(4): p. 234-237.
- 264. Proctor, C.S., D.W. Jackson, and T.M. Simon, *Characterization of the repair tissue after removal of the central one- third of the patellar ligament: An experimental study in a goat model.* Journal of Bone and Joint Surgery Series A, 1997. **79**(7): p. 997-1006.
- 265. Butler, D.L., S.A. Goldstein, and F. Guilak, *Functional tissue engineering: the role of biomechanics*. Journal of biomechanical engineering, 2000. **122**(Dec): p. 570-5.
- 266. Benjamin, M., R.N.S. Tyers, and J.R. Ralphs, *Age-related changes in tendon fibrocartilage*, in *Journal of Anatomy*. 1991. p. 127-136.
- 267. St. Pierre, P., et al., *Tendon-healing to cortical bone compared with healing to a cancellous trough: A biomechanical and histological evaluation in goats.* Journal of Bone and Joint Surgery Series A, 1995. **77**(12): p. 1858-1866.
- 268. Thomopoulos, S., et al., *The localized expression of extracellular matrix components in healing tendon insertion sites: An in situ hybridization study.* Journal of Orthopaedic Research, 2002. **20**(3): p. 454-463.
- 269. Thomopoulos, S., G.R. Williams, and L.J. Soslowsky, *Tendon to bone healing: Differences in biomechanical, structural, and compositional properties due to a range of activity levels.* Journal of Biomechanical Engineering, 2003. **125**(1): p. 106-113.
- 270. Ralphs, J.R., R.N.S. Tyers, and M. Benjamin, *Development of functionally distinct fibrocartilages at two sites in the quadriceps tendon of the rat: The suprapatella and the attachment to the patella*. Anatomy and Embryology, 1992. **185**(2): p. 181-187.
- 271. Vogel, K.G., et al., *Proteoglycans in the compressed region of human tibialis posterior tendon and in ligaments.* Journal of Orthopaedic Research, 1993. **11**(1): p. 68-77.
- 272. Thomopoulos, S., et al., *Decreased muscle loading delays maturation of the tendon enthesis during postnatal development*. Journal of Orthopaedic Research, 2007. **25**(9): p. 1154-1163.
- 273. Chhabra, A., et al., *GDF-5 deficiency in mice delays Achilles tendon healing*. Journal of Orthopaedic Research, 2003. **21**(5): p. 826-835.
- 274. Rickert, M., et al., Adenovirus-mediated gene transfer of growth and differentiation factor-5 into tenocytes and the healing rat achilles tendon. Connective Tissue Research, 2005. **46**(4-5): p. 175-183.
- 275. Würgler-Hauri, C.C., et al., *Temporal expression of 8 growth factors in tendon-to-bone healing in a rat supraspinatus model*. Journal of Shoulder and Elbow Surgery, 2007. **16**(5 SUPPL.): p. S198-S203.
- 276. Falanga, V., et al., *Transforming growth factor-beta: selective increase in glycosaminoglycan synthesis by cultures of fibroblasts from patients with progressive systemic sclerosis.* J Invest Dermatol, 1987. **89**(Jul): p. 100-4.
- 277. Bonamo, J.J., R.M. Krinick, and A.A. Sporn, *Rupture of the patellar ligament after use of its central third for anterior cruciate reconstruction. A report of two cases.* J Bone Joint Surg Am, 1984. **66**(8): p. 1294-7.
- 278. Kashiwagi, K., et al., *Effects of transforming growth factor-beta 1 on the early stages of healing of the Achilles tendon in a rat model.* Scand J Plast Reconstr Surg Hand Surg, 2004. **38**(4): p. 193-7.

- 279. Chokalingam, K., et al., 3D-In vitro Effects of Compression and Time in Culture on Aggregate Modulus and on Gene Expression and Protein content of Collagen Type II in Murine Chondrocytes
- Tissue Eng Part A 2009, 2009.
- 280. Langer, R. and J.P. Vacanti, *Tissue engineering*. Science, 1993. 260(May 14): p. 920-6.
- 281. Richmond, J.C., et al., Anterior cruciate reconstruction using a Dacron ligament prosthesis: A long-term study. American Journal of Sports Medicine, 1992. **20**(1): p. 24-28.
- 282. Moyen, B.J.L., et al., *Comparison of reconstruction of the anterior cruciate ligament with and without a Kennedy ligament-augmentation device. A randomized, prospective study.* Journal of Bone and Joint Surgery Series A, 1992. **74**(9): p. 1313-1319.
- 283. Amiel, D., J. Billings E., and W.H. Akeson, *Ligament structure, chemistry and physiology*. Knee Ligaments: Structure, Function, Injury and Repair, 1990: p. 77-91.
- 284. Jorgensen, U., et al., *Reconstruction of the anterior cruciate ligament with the iliotibial band autograft in patients with chronic knee instability.* Knee Surg Sports Traumatol Arthrosc, 2001. **9**(3): p. 137-45.
- 285. Steenbrugge, F., et al., *Repair of chronic ruptures of the anterior cruciate ligament using allograft reconstruction and a ligament augmentation device*. Acta Orthopaedica Belgica, 2001. **67**(3): p. 252-258.
- 286. Stone, K.R., Meniscus replacement. Clinics in Sports Medicine, 1996. 15(3): p. 557-&.
- 287. Bruns, J., et al., *Autologous perichondral tissue for meniscal replacement*. Journal of Bone and Joint Surgery-British Volume, 1998. **80B**(5): p. 918-923.
- 288. Kohn, D., et al., *Medial Meniscus Replacement by a Tendon Autograft Experiments in Sheep.* Journal of Bone and Joint Surgery-British Volume, 1992. **74**(6): p. 910-912.
- 289. Messner, K., *Meniscal Substitution with a Teflon Periosteal Composite Graft a Rabbit Experiment*. Biomaterials, 1994. **15**(3): p. 223-230.
- 290. Noth, U., et al., Anterior cruciate ligament constructs fabricated from human mesenchymal stem cells in a collagen type I hydrogel. Cytotherapy, 2005. **7**(5): p. 447-455.
- 291. Butler, D., M. Dressler, and H. Awad, *Functional Tissue Engineering: Assessment of Function in Tendon and Ligament Repair*, in *Functional Tissue Engineering*, F. Guilak, et al., Editors. 2003, Springer:: New York.
- 292. Butler, D., S. Goldstein, and F. Uilak, *Functional Tissue Engineering: The role of Biomechanics*. Journal of Biomechanical Engineering, 2001.
- 293. Butler, D.L., et al., *Functional tissue engineering parameters toward designing repair and replacement strategies.* Clinical Orthopaedics and Related Research, 2004(427): p. S190-S199.
- 294. Freed, L.E., et al., *Tissue engineering of cartilage in space*. Proceedings of the National Academy of Sciences of the United States of America, 1997. **94**(25): p. 13885-13890.
- 295. Vunjak-Novakovic, G., et al., *Bioreactor studies of native and tissue engineered cartilage*. Biorheology, 2002. **39**(1-2): p. 259-268.
- 296. Banes, A.J., et al., A new vacuum-operated stress-providing instrument that applies static or variable duration cyclic tension or compression to cells in vitro. Journal of Cell Science, 1985. **75**: p. 35-42.
- 297. Carvalho, R.S., E.H. Yen, and D.M. Suga, *Glycosaminoglycan synthesis in the rat articular disk in response to mechanical stress.* American journal of orthodontics and

dentofacial orthopedics : official publication of the American Association of Orthodontists, its constituent societies, and the American Board of Orthodontics, 1995. **107**(4): p. 401-410.

- 298. Sambajon, V.V., et al., *The effects of mechanical strain on synovial fibroblasts*. Journal of Oral and Maxillofacial Surgery, 2003. **61**(6): p. 707-712.
- 299. Upton, M.L., et al., *Differential effects of static and dynamic compression on meniscal cell gene expression.* Journal of Orthopaedic Research, 2003. **21**(6): p. 963-969.
- 300. Aufderheide, A.C. and K.A. Athanasiou, *A direct compression stimulator for articular cartilage and meniscal explants*. Annals of Biomedical Engineering, 2006. **34**(9): p. 1463-1474.
- 301. Frank, E.H., et al., A versatile shear and compression apparatus for mechanical stimulation of tissue culture explants. Journal of Biomechanics, 2000. **33**(11): p. 1523-1527.
- 302. Mauck, R.L., et al., *Functional tissue engineering of articular cartilage through dynamic loading of chondrocyte-seeded agarose gels.* Journal of Biomechanical Engineering, 2000. **122**(3): p. 252-260.
- 303. Ahmed, A.M. and D.L. Burke, *In-vitro measurement of static pressure distribution in synovial joints Part I: Tibial surface of the knee*. Journal of Biomechanical Engineering, 1983. **105**(3): p. 216-225.
- 304. Brown, T.D. and D.T. Shaw, *In vitro contact stress distribution on the femoral condyles*. Journal of Orthopaedic Research, 1984. **2**(2): p. 190-199.
- 305. Eckstein, F., et al., *In vivo cartilage deformation after different types of activity and its dependence on physical training status.* Annals of the Rheumatic Diseases, 2005. **64**(2): p. 291-295.
- 306. Eckstein, F., et al., *Functional analysis of articular cartilage deformation, recovery, and fluid flow following dynamic exercise in vivo.* Anatomy and Embryology, 1999. **200**(4): p. 419-424.
- 307. Sipes, N.S., J.T. Shearn, and D.L. Butler, *The effects of orientation, temperature, and displacement magnitude changes on the sonometrics system accuracy.* Journal of Biomechanics, 2005. **38**(12): p. 2486-2490.
- 308. Thirion, S. and F. Berenbaum, *Culture and phenotyping of chondrocytes in primary culture*. Methods in molecular medicine, 2004. **100**: p. 1-14.
- 309. Meyer, U., et al., *Design and performance of a bioreactor system for mechanically promoted three-dimensional tissue engineering.* British Journal of Oral and Maxillofacial Surgery, 2006. **44**(2): p. 134-140.
- 310. Muir, H., *The chondrocyte, architect of cartilage. Biomechanics, structure, function and molecular biology of cartilage matrix macromolecules.* BioEssays, 1995. **17**(12): p. 1039-1048.
- 311. Mow, V.C., A. Ratcliffe, and A.R. Poole, *Cartilage and diarthrodial joints as paradigms for hierarchical materials and structures.* Biomaterials, 1992. **13**(2): p. 67-97.
- 312. Nesic, D., et al., *Cartilage tissue engineering for degenerative joint disease*. Advanced Drug Delivery Reviews, 2006. **58**(2): p. 300-322.
- 313. Kuo, C.K., et al., *Cartilage tissue engineering: Its potential and uses.* Current Opinion in Rheumatology, 2006. **18**(1): p. 64-73.
- 314. <u>http://www.cdc.gov/arthritis/data_statistics/arthritis_related_statistics.htm#1</u>.
- 315. <u>http://www.cdc.gov/arthritis/data_statistics/arthritis_related_statistics.htm#10</u>.

- 316. Jackson, R.W. and C. Dietrichs, *The results of arthroscopic lavage and debridement of osteoarthritic knees based on the severity of degeneration: A 4-to 6-year symptomatic follow-up.* Arthroscopy-the Journal of Arthroscopic and Related Surgery, 2003. **19**(1): p. 13-20.
- 317. Reading, A.D., P. Rooney, and G.J.S. Taylor, *Quantitative assessment of the effect of* 0.05% chlorhexidine on rat articular cartilage metabolism in vitro and in vivo. Journal of Orthopaedic Research, 2000. **18**(5): p. 762-767.
- 318. Bouwmeester, S.J.M., et al., *Long-term results of rib perichondrial grafts for repair of cartilage defects in the human knee*. International Orthopaedics, 1997. **21**(5): p. 313-317.
- 319. Carranza-Bencano, A., et al., *Comparative study of the reconstruction of articular cartilage defects with free costal perichondrial grafts and free tibial periosteal grafts: An experimental study on rabbits.* Calcified Tissue International, 1999. **65**(5): p. 402-407.
- 320. Hangody, L., et al., *Autologous Osteochondral Mosaicplasty*. Journal of Bone and Joint Surgery Series A, 2004. **86**(SUPPL. 1): p. 65-72.
- 321. Jakob, R.P., et al., *Autologous osteochondral grafting in the knee: Indication, results, and reflections.* Clinical Orthopaedics and Related Research, 2002(401): p. 170-184.
- 322. Huntley, J.S., et al., *Chondrocyte death associated with human femoral osteochondral harvest as performed for mosaicplasty.* Journal of Bone and Joint Surgery-American Volume, 2005. **87A**(2): p. 351-360.
- 323. Guilak, F., Functional tissue engineering The role of biomechanics in reparative medicine, in Reparative Medicine: Growing Tissues and Organs. 2002. p. 193-195.
- 324. Shearn, J.T., et al., *Mechanical stimulation of tendon tissue engineered constructs: Effects on construct stiffness, repair biomechanics, and their correlation.* Journal of Biomechanical Engineering, 2007. **129**(6): p. 848-854.
- Hung, C.T., et al., A paradigm for functional tissue engineering of articular cartilage via applied physiologic deformational loading. Annals of Biomedical Engineering, 2004.
 32(1): p. 35-49.
- 326. Mauck, R.L., et al., *The role of cell seeding density and nutrient supply for articular cartilage tissue engineering with deformational loading.* Osteoarthritis and Cartilage, 2003. **11**(12): p. 879-890.
- 327. Waldman, S.D., et al., Long-term intermittent compressive stimulation improves the composition and mechanical properties of tissue-engineered cartilage. Tissue Engineering, 2004. **10**(9-10): p. 1323-1331.
- 328. Benya, P.D. and J.D. Shaffer, *Dedifferentiated chondrocytes reexpress the differentiated collagen phenotype when cultured in agarose gels.* Cell, 1982. **30**(1): p. 215-224.
- 329. Molecular Devices Spectramax Gemini-XS Application Note 44.
- 330. Hunter, S.A., et al., *Effects of matrix stabilization when using glutaraldehyde on the material properties of porcine meniscus.* Journal of Biomedical Materials Research Part A, 2003. **67**(4): p. 1245-1254.
- 331. Stabler, C.L., et al., *Noninvasive measurement of viable cell number in tissue-engineered constructs in vitro, using 1H nuclear magnetic resonance spectroscopy.* Tissue Engineering, 2005. **11**(3-4): p. 404-414.
- 332. <u>http://www.chondrex.com/products/captureelisa.htm</u>.
- 333. Thirion, S. and F. Berenbaum, *Culture and phenotyping of chondrocytes in primary culture*. Methods in molecular medicine, 2004. **100**: p. 1-14.

- 334. Montgomery, D., *Design and Analysis of Experiments (5th Edition).* 2001, New York, NY: John Wiley and Sons.
- 335. Wilkins, R.J., J.A. Browning, and J.P.G. Urban, *Chondrocyte regulation by mechanical load*. Biorheology, 2000. **37**(1-2): p. 67-74.
- 336. De Croos, J.N.A., et al., *Cyclic compressive mechanical stimulation induces sequential catabolic and anabolic gene changes in chondrocytes resulting in increased extracellular matrix accumulation.* Matrix Biology, 2006. **25**(6): p. 323-331.
- 337. Xie, J., et al., Mechanical loading-dependence of mRNA expressions of extracellular matrices of chondrocytes inoculated into elastomeric microporous poly(L-lactide-co-?-caprolactone) scaffold. Tissue Engineering, 2007. **13**(1): p. 29-40.
- 338. Hunter, C.J., et al., *Mechanical compression alters gene expression and extracellular matrix synthesis by chondrocytes cultured in collagen I gels.* Biomaterials, 2002. **23**(4): p. 1249-1259.
- 339. Demarteau, O., et al., *Dynamic compression of cartilage constructs engineered from expanded human articular chondrocytes*. Biochemical and Biophysical Research Communications, 2003. **310**(2): p. 580-588.
- 340. Xie, J., Z.Y. Han, and T. Matsuda, *Mechanical compressive loading stimulates the activity of proximal region of human COL2A1 gene promoter in transfected chondrocytes*. Biochemical and Biophysical Research Communications, 2006. **344**(4): p. 1192-1199.
- 341. Frank, E.H. and A.J. Grodzinsky, *Cartilage electromechanics I. Electrokinetic transduction and the effects of electrolyte pH and ionic strength*. Journal of Biomechanics, 1987. **20**(6): p. 615-627.
- 342. Garcia, A.M., et al., *Contributions of fluid convection and electrical migration to transport in cartilage: Relevance to loading.* Archives of Biochemistry and Biophysics, 1996. **333**(2): p. 317-325.
- 343. Kim, Y.-J., L.J. Bonassar, and A.J. Grodzinsky, *The role of cartilage streaming potential*, *fluid flow and pressure in the stimulation of chondrocyte biosynthesis during dynamic compression*. Journal of Biomechanics, 1995. **28**(9): p. 1055-1066.
- 344. Nirmalanandhan, V.S., et al., *Effect of scaffold material, construct length and mechanical stimulation on the in vitro stiffness of the engineered tendon construct.* Journal of Biomechanics, 2008. **41**(4): p. 822-828.
- 345. Darling, E.M. and K.A. Athanasiou, *Rapid phenotypic changes in passaged articular chondrocyte subpopulations*. Journal of Orthopaedic Research, 2005. **23**(2): p. 425-432.
- 346. Lefebvre, V., C. Peeters-Joris, and G. Vaes, *Production of collagens, collagenase and collagenase inhibitor during the dedifferentiation of articular chondrocytes by serial subcultures*. Biochimica et Biophysica Acta Molecular Cell Research, 1990. **1051**(3): p. 266-275.
- 347. Getgood, A., et al., *Articular cartilage tissue engineering: Today's research, tomorrow's practice?* Journal of Bone and Joint Surgery Series B, 2009. **91**(5): p. 565-576.
- 348. Schindler, O.S., *Cartilage repair using autologous chondrocyte implantation techniques*. Journal of perioperative practice, 2009. **19**(2): p. 60-64.
- 349. Elder, B.D. and K.A. Athanasiou, *Hydrostatic pressure in articular cartilage tissue engineering: from chondrocytes to tissue regeneration.* Tissue engineering. Part B, Reviews, 2009. **15**(1): p. 43-53.

- 350. Lattermann, C. and S.E. Romine, *Osteochondral Allografts: State of the Art.* Clinics in Sports Medicine, 2009. **28**(2): p. 285-301.
- 351. LaPrade, R.F., et al., *Refrigerated osteoarticular allografts to treat articular cartilage defects of the femoral condyles. A prospective outcomes study.* The Journal of bone and joint surgery. American volume, 2009. **91**(4): p. 805-811.
- 352. Rozen, B., et al., *The effects of prolonged deep freezing on the biomechanical properties of osteochondral allografts*. Cell and Tissue Banking, 2009. **10**(1): p. 27-31.
- 353. Gross, A.E., et al., *Fresh osteochondral allografts for posttraumatic knee defects: Longterm followup.* Clinical Orthopaedics and Related Research, 2008. **466**(8): p. 1863-1870.
- 354. Colangeli, M., et al., *Total knee replacement versus osteochondral allograft in proximal tibia bone tumours.* International Orthopaedics, 2007. **31**(6): p. 823-829.
- 355. Hennig, A. and J. Abate, *Osteochondral allografts in the treatment of articular cartilage injuries of the knee*. Sports Medicine and Arthroscopy Review, 2007. **15**(3): p. 126-132.
- 356. Davidson, P.A. and T.R. Carter, *Osteochondral Allograft Transplantation*. Operative Techniques in Sports Medicine, 2007. **15**(2): p. 53-61.
- 357. McCulloch, P.C., et al., *Prospective evaluation of prolonged fresh osteochondral allograft transplantation of the femoral condyle: Minimum 2-year follow-up.* American Journal of Sports Medicine, 2007. **35**(3): p. 411-420.
- 358. McCulloch, P.C., R. Kang, and B.J. Cole, *Osteochondral allografts for large defects in the knee*. Techniques in Knee Surgery, 2006. **5**(3): p. 165-173.
- 359. Edward Glenn Jr., R., et al., *Comparison of fresh osteochondral autografts and allografts: A canine model.* American Journal of Sports Medicine, 2006. **34**(7): p. 1084-1093.
- 360. Spak, R.T. and R.A. Teitge, *Fresh osteochondral allografts for patellofemoral arthritis: Long-term followup.* Clinical Orthopaedics and Related Research, 2006(444): p. 193-200.
- 361. Frenkel, S.R., E.N. Kubiak, and K.G. Truncale, *The repair response to osteochondral implant types in a rabbit model.* Cell and Tissue Banking, 2006. **7**(1): p. 29-37.
- 362. Jamali, A.A., et al., *Fresh osteochondral allografts: Results in the patellofemoral joint.* Clinical Orthopaedics and Related Research, 2005(437): p. 176-185.
- 363. Brittberg, M., et al., *Treatment of deep cartilage defects in the knee with autologous chondrocyte transplantation*, in *New England Journal of Medicine*. 1994. p. 889-895.
- 364. Peterson, L., et al., *Two-to 9-year outcome after autologous chondrocyte transplantation of the knee*, in *Clinical Orthopaedics and Related Research*. 2000. p. 212-234.
- 365. Gulotta, L.V., et al., Chondrocyte Death and Cartilage Degradation Following Autologous Osteochondral Transplantation Surgery in a Rabbit Model. Am J Sports Med, 2009.
- 366. Mauck, R.L., et al., *Functional tissue engineering of articular cartilage through dynamic loading of chondrocyte-seeded agarose gels*, in *Journal of Biomechanical Engineering*. 2000. p. 252-260.
- 367. Mauck, R.L., et al., *Influence of seeding density and dynamic deformational loading on the developing structure/function relationships of chondrocyte-seeded agarose hydrogels*, in *Annals of Biomedical Engineering*. 2002. p. 1046-1056.
- 368. Lima, E.G., et al., *The beneficial effect of delayed compressive loading on tissueengineered cartilage constructs cultured with TGF-?3*, in *Osteoarthritis and Cartilage*. 2007. p. 1025-1033.

- 369. Hunziker, E.B., Articular cartilage repair: Basic science and clinical progress. A review of the current status and prospects. Osteoarthritis and Cartilage, 2002. **10**(6): p. 432-463.
- 370. Barbero, A., et al., *Age related changes in human articular chondrocyte yield, proliferation and post-expansion chondrogenic capacity*, in *Osteoarthritis and Cartilage*. 2004. p. 476-484.
- 371. Tallheden, T., et al., *Proliferation and differentiation potential of chondrocytes from osteoarthritic patients.*, in *Arthritis research & therapy.* 2005.
- 372. Fan, Z., et al., Regulation of anabolic and catabolic gene expression in normal and osteoarthritic adult human articular chondrocytes by osteogenic protein-1, in Clinical and Experimental Rheumatology. 2004. p. 103-106.
- 373. Fan, Z., et al., *Freshly isolated osteoarthritic chondrocytes are catabolically more active than normal chondrocytes, but less responsive to catabolic stimulation with interleukin-1?*, in *Arthritis and Rheumatism.* 2005. p. 136-143.
- 374. Martin, J.A., S.M. Ellerbroek, and J.A. Buckwalter, *Age-related decline in chondrocyte response to insulin-like growth factor-I: The role of growth factor binding proteins*, in *Journal of Orthopaedic Research*. 1997. p. 491-498.
- 375. Aigner, T., et al., Osteoarthritis: Aging of matrix and cells Going for a remedy, in Current Drug Targets. 2007. p. 325-331.
- 376. Hwang, N.S. and J. Elisseeff, *Application of stem cells for articular cartilage regeneration*. The journal of knee surgery, 2009. **22**(1): p. 60-71.
- 377. Chen, F.H., K.T. Rousche, and R.S. Tuan, *Technology insight: Adult stem cells in cartilage regeneration and tissue engineering*. Nature Clinical Practice Rheumatology, 2006. 2(7): p. 373-382.
- 378. Caplan, A.I., *Mesenchymal stem cells: Cell-based reconstructive therapy in orthopedics.* Tissue Engineering, 2005. **11**(7-8): p. 1198-1211.
- 379. Song, L., D. Baksh, and R.S. Tuan, *Mesenchymal stem cell-based cartilage tissue engineering: Cells, scaffold and biology.* Cytotherapy, 2004. **6**(6): p. 596-601.
- 380. Caplan, A.I., *Mesenchymal stem cells*, in *Journal of Orthopaedic Research*. 1991. p. 641-650.
- 381. Iwasaki, M., et al., *Transforming growth factor-?1 stimulates chondrogenesis and inhibits osteogenesis in high density culture of periosteum-derived cells*, in *Endocrinology*. 1993. p. 1603-1608.
- 382. Johnstone, B., et al., In vitro chondrogenesis of bone marrow-derived mesenchymal progenitor cells, in Experimental Cell Research. 1998. p. 265-272.
- 383. Mackay, A.M., et al., *Chondrogenic differentiation of cultured human mesenchymal stem cells from marrow*, in *Tissue Engineering*. 1998. p. 415-428.
- 384. Nishimura, K., et al., Chondroprogenitor cells of synovial tissue, in Arthritis and Rheumatism. 1999. p. 2631-2637.
- 385. Yoo, J.U., et al., *The chondrogenic potential of human bone-marrow-derived mesenchymal progenitor cells*, in *Journal of Bone and Joint Surgery Series A.* 1998. p. 1745-1757.
- 386. Zuk, P.A., et al., Multilineage cells from human adipose tissue: Implications for cellbased therapies, in Tissue Engineering. 2001. p. 211-228.
- 387. Campbell, J.J., D.A. Lee, and D.L. Bader, *Dynamic compressive strain influences chondrogenic gene expression in human mesenchymal stem cells*, in *Biorheology*. 2006. p. 455-470.

- 388. Huang, C.-Y.C., et al., *Effects of cyclic compressive loading on chondrogenesis of rabbit bone-marrow derived mesenchymal stem cells*, in *Stem Cells*. 2004. p. 313-323.
- 389. Mauck, R.L., et al., *Regulation of cartilaginous ECM gene transcription by chondrocytes and MSCs in 3D culture in response to dynamic loading*, in *Biomechanics and Modeling in Mechanobiology*. 2007. p. 113-125.
- 390. Mouw, J.K., et al., Dynamic compression regulates the expression and synthesis of chondrocyte-specific matrix molecules in bone marrow stromal cells, in Stem Cells. 2007. p. 655-663.
- 391. Angele, P., et al., Cyclic, mechanical compression enhances chondrogenesis of mesenchymal progenitor cells in tissue engineering scaffolds, in Biorheology. 2004. p. 335-346.
- 392. Miyanishi, K., et al., Dose- and time-dependent effects of cyclic hydrostatic pressure on transforming growth factor-?3-induced chondrogenesis by adult human mesenchymal stem cells in vitro. Tissue Engineering, 2006. **12**(8): p. 2253-2262.
- 393. Miyanishi, K., et al., *Effects of hydrostatic pressure and transforming growth factor-?3* on adult human mesenchymal stem cell chondrogenesis in vitro. Tissue Engineering, 2006. **12**(6): p. 1419-1428.
- 394. Peister, A., et al., Adult stem cells from bone marrow (MSCs) isolated from different strains of inbred mice vary in surface epitopes, rates of proliferation, and differentiation potential. Blood, 2004. **103**(5): p. 1662-1668.
- 395. Dobson, K.R., et al., *Centrifugal isolation of bone marrow from bone: An improved method for the recovery and quantitation of bone marrow osteoprogenitor cells from rat tibiae and femurae.* Calcified Tissue International, 1999. **65**(5): p. 411-413.
- 396. Sumpio, B.E., et al., *Enhanced collagen production by smooth muscle cells during repetitive mechanical stretching*. Archives of Surgery, 1988. **123**(10): p. 1233-1236.
- 397. Stabler, C.L., et al., *Noninvasive measurement of viable cell number in tissue-engineered constructs in vitro, using 1H nuclear magnetic resonance spectroscopy*. Tissue Engineering, 2005. **11**(3-4): p. 404-414.
- 398. Zeichen, J., M. Van Griensven, and U. Bosch, *The proliferative response of isolated human tendon fibroblasts to cyclic biaxial mechanical strain*. American Journal of Sports Medicine, 2000. **28**(6): p. 888-892.
- 399. Bosnakovski, D., et al., Chondrogenic differentiation of bovine bone marrow mesenchymal stem cells (MSCs) in different hydrogels: Influence of collagen type II extracellular matrix on MSC chondrogenesis. Biotechnology and Bioengineering, 2006. 93(6): p. 1152-1163.
- 400. Thorpe, S.D., et al., *Dynamic compression can inhibit chondrogenesis of mesenchymal stem cells*. Biochemical and Biophysical Research Communications, 2008. **377**(2): p. 458-462.
- 401. Buschmann, M.D., et al., Mechanical compression modulates matrix biosynthesis in chondrocyte/agarose culture, in Journal of Cell Science. 1995. p. 1497-1508.
- 402. Erickson, I.E., et al., Differential maturation and structure-function relationships in mesenchymal stem cell- and chondrocyte-seeded hydrogels. Tissue Eng Part A, 2009. 15(5): p. 1041-52.
- 403. Vogel, K.G. and T.J. Koob, *Structural Specialization in Tendons under Compression*. International Review of Cytology-a Survey of Cell Biology, 1989. **115**: p. 267-293.

- 404. Wren, T.A.L., G.S. Beaupré, and D.R. Carter, *Mechanobiology of tendon adaptation to compressive loading through fibrocartilaginous metaplasia*, in *Journal of Rehabilitation Research and Development*. 2000. p. 135-143.
- 405. Giori, N.J., G.S. Beaupre, and D.R. Carter, *Cellular-Shape and Pressure May Mediate Mechanical Control of Tissue Composition in Tendons*. Journal of Orthopaedic Research, 1993. **11**(4): p. 581-591.
- 406. Okuda, Y., et al., *Biochemical, Histological, and Biomechanical Analyses of Canine Tendon.* Journal of Orthopaedic Research, 1987. **5**(1): p. 60-68.
- 407. Okuda, Y., J.P. Gorski, and P.C. Amadio, *Effect of postnatal age on the ultrastructure of six anatomical areas of canine flexor digitorum profundus tendon*. J Orthop Res, 1987. 5(2): p. 231-41.
- 408. Evanko, S.P. and K.G. Vogel, Proteoglycan synthesis in fetal tendon is differentially regulated by cyclic compression in vitro, in Archives of Biochemistry and Biophysics. 1993. p. 153-164.
- 409. Robbins, J.R., S.P. Evanko, and K.G. Vogel, *Mechanical loading and TGF-beta regulate proteoglycan synthesis in tendon*. Archives of Biochemistry and Biophysics, 1997. 342(2): p. 203-211.
- 410. Evanko, S.P. and K.G. Vogel, *Proteoglycan Synthesis in Fetal Tendon Is Differentially Regulated by Cyclic Compression in-Vitro*. Archives of Biochemistry and Biophysics, 1993. **307**(1): p. 153-164.
- 411. Koob, T.J., et al., *Compression Loading Invitro Regulates Proteoglycan Synthesis by Tendon Fibrocartilage*. Archives of Biochemistry and Biophysics, 1992. **298**(1): p. 303-312.
- 412. Gillard, G.C., et al., *The influence of mechanical forces on the glycosaminoglycan content of the rabbit flexor digitorum profundus tendon*. Connective Tissue Research, 1979. **7**(1): p. 37-46.
- 413. Elder, S.H., K.S. Fulzele, and W.R. McCulley, *Cyclic hydrostatic compression stimulates chondroinduction of C3H/10T1/2 cells*, in *Biomechanics and Modeling in Mechanobiology*. 2005. p. 141-146.
- 414. Carter, D.R., et al., *Influences of mechanical stress on prenatal and postnatal skeletal development*, in *Clinical Orthopaedics and Related Research*. 1987. p. 237-250.
- 415. Carter, D.R., P.R. Blenman, and G.S. Beaupre, *Correlations between mechanical stress history and tissue differentiation in initial fracture healing*, in *Journal of Orthopaedic Research*. 1988. p. 736-748.
- 416. Kim, S.-G., et al., *Gene expression of type I and Type III collagen by mechanical stretch in anterior cruciate ligament cells.* Cell Structure and Function, 2002. **27**(3): p. 139-144.
- 417. Webb, K., et al., Cyclic strain increases fibroblast proliferation, matrix accumulation, and elastic modulus of fibroblast-seeded polyurethane constructs. Journal of Biomechanics, 2006. **39**(6): p. 1136-1144.
- 418. Howard, P.S., et al., *Mechanical forces alter extracellular matrix synthesis by human periodontal ligament fibroblasts.* Journal of Periodontal Research, 1998. **33**(8): p. 500-508.
- 419. Nöth, U., et al., Anterior cruciate ligament constructs fabricated from human mesenchymal stem cells in a collagen type I hydrogel. Cytotherapy, 2005. **7**(5): p. 447-455.

- 420. Park, J.S., et al., *Differential effects of equiaxial and uniaxial strain on mesenchymal stem cells*. Biotechnology and Bioengineering, 2004. **88**(3): p. 359-368.
- 421. Altman, G.H., et al., *Cell differentiation by mechanical stress*. The FASEB journal : official publication of the Federation of American Societies for Experimental Biology, 2002. **16**(2): p. 270-272.
- 422. Wong, M., M. Siegrist, and K. Goodwin, Cyclic tensile strain and cyclic hydrostatic pressure differentially regulate expression of hypertrophic markers in primary chondrocytes, in Bone. 2003. p. 685-693.
- 423. He, Y., et al., *Compression and tension: Differential effects on matrix accumulation by periodontal ligaments fibroblasts in vitro*, in *Connective Tissue Research*. 2004. p. 28-39.
- 424. Ikenoue, T., et al., Mechanoregulation of human articular chondrocyte aggrecan and type II collagen expression by intermittent hydrostatic pressure in vitro, in Journal of Orthopaedic Research. 2003. p. 110-116.
- 425. Smith, R.L., et al., *In vitro stimulation of articular chondrocyte mRNA and extracellular matrix synthesis by hydrostatic pressure*, in *Journal of Orthopaedic Research*. 1996. p. 53-60.
- 426. Malemud, C.J., et al., *The proteoglycan synthesis repertoire of rabbit chondrocytes maintained in type II collagen gels*, in *Osteoarthritis and Cartilage*. 1994. p. 29-41.
- 427. Schuman, L., et al., *Chondrocyte behaviour within different types of collagen gel in vitro*, in *Biomaterials*. 1995. p. 809-814.
- 428. Huang, H.H., A.A. Qureshi, and J.J. Biundo, Jr., Sports and other soft tissue injuries, tendinitis, bursitis, and occupation-related syndromes. Curr Opin Rheumatol, 2000. 12(2): p. 150-4.
- 429. Liu, S. and T. Nguyen, *Ankle sprains and other soft tissue injuries*. Curr Opin Rheumatol, 1999. **11**(Mar): p. 132-7.
- 430. Bennett, W.F., Arthroscopic repair of massive rotator cuff tears: A prospective cohort with 2- to 4-year follow-up. Arthroscopy, 2003. **19**(4): p. 380-90.
- 431. Praemer, A., S. Furner, and D.P. Rice, *Musculoskeletal Condition in the United States*. 1999, Parke Ridge, IL: American Academy of Orthopaedic Surgeons. 182.
- 432. Bey, M.J., M.L. Ramsey, and L.J. Soslowsky, *Intratendinous strain fields of the supraspinatus tendon: effect of a surgically created articular-surface rotator cuff tear.* J Shoulder Elbow Surg, 2002. **11**(6): p. 562-9.
- 433. Soslowsky, L.J., et al., *Neer Award 1999. Overuse activity injures the supraspinatus tendon in an animal model: a histologic and biomechanical study.* J Shoulder Elbow Surg, 2000. **9**(2): p. 79-84.
- 434. Walsh, C.J., et al., *Meniscus regeneration in a rabbit partial meniscectomy model*. Tissue Engineering, 1999. **5**(4): p. 327-337.
- 435. Albright, J.C., et al., *Knee and leg: Soft-tissue trauma*. Orthopaedic Knowledge Update, 1999: p. 533-557.
- 436. McNicholas, M., et al., *Total meniscectomy in adolescence. A thirty-year follow-up.* J Bone Joint Surg [Br], 2000. **82**(Mar): p. 217-21.
- 437. Lo, S.S., et al. Mechanical loading modulates gene expression in chondrocyte-seeded agarose hydrogels. 2001.
- 438. Butler, D.L., N. Juncosa, and M.R. Dressler, *Functional Efficacy of Tendon Repair Processes*. Annu Rev Biomed Eng., 2004. **6**: p. 303-329.

- 439. Peister, A., et al., Adult stem cells from bone marrow (MSCs) isolated from different strains of inbred mice vary in surface epitopes, rates of proliferation, and differentiation potential. Blood, 2004. **103**(5): p. 1662-8.
- 440. Popov, E.P., *Engineering mechanics of soilds*. 1990 ed, Englewood Cliffs, N.J: Prentice Hall.
- 441. Toyoda, T., et al., *Tensile load and the metabolism of anterior cruciate ligament cells*. Clinical Orthopaedics and Related Research, 1998(353): p. 247-255.
- 442. Archambault, J., et al., *Stretch and interleukin-1beta induce matrix metalloproteinases in rabbit tendon cells in vitro*. J Orthop Res, 2002. **20**(1): p. 36-9.
- 443. Riser, B.L., et al., Cyclic stretching force selectively up-regulates transforming growth factor-? isoforms in cultured rat mesangial cells. American Journal of Pathology, 1996. **148**(6): p. 1915-1923.
- 444. Kim, S.-G., et al., *Gene expression of type I and Type III collagen by mechanical stretch in anterior cruciate ligament cells.* Cell Structure and Function, 2002. **27**(3): p. 139-144.
- 445. Paye, M., B.V. Nusgens, and C.M. Lapie?re, *Modulation of cellular biosynthetic activity in the retracting collagen lattice*. European Journal of Cell Biology, 1987. **45**(1): p. 44-50.
- 446. Nakagawa, S., P. Pawelek, and F. Grinnell, *Long-term culture of fibroblasts in contracted collagen gels: Effects on cell growth and biosynthetic activity.* Journal of Investigative Dermatology, 1989. **93**(6): p. 792-798.
- 447. Thie, M., et al., *Mechanical confinement inhibits collagen synthesis in gel-cultured fibroblasts.* European Journal of Cell Biology, 1989. **48**(2): p. 294-302.
- 448. Langholz, O., et al., *Collagen and collagenase gene expression in three-dimensional collagen lattices are differentially regulated by ?1?1 and ?2?1 integrins.* Journal of Cell Biology, 1995. **131**(6 II): p. 1903-1915.
- 449. Riikonen, T., et al., Integrin ?2?1 is a positive regulator of collagenase (MMP-1) and collagen ?1(I) gene expression. Journal of Biological Chemistry, 1995. **270**(22): p. 13548-13552.
- 450. Xu, J. and R.A.F. Clark, A three-dimensional collagen lattice induces protein kinase C-? activity: Role in ?2 integrin and collagenase mRNA expression. Journal of Cell Biology, 1997. **136**(2): p. 473-483.
- 451. Chastain, S.R., et al., Adhesion of mesenchymal stem cells to polymer scaffolds occurs via distinct ECM ligands and controls their osteogenic differentiation. J Biomed Mater Res A, 2006. **78**(1): p. 73-85.
- 452. RonziÃ["]re, M.-C., et al., Ascorbate modulation of bovine chondrocyte growth, matrix protein gene expression and synthesis in three-dimensional collagen sponges. Biomaterials, 2003. **24**(5): p. 851-861.
- 453. Karamichos, D., R.A. Brown, and V. Mudera, *Collagen stiffness regulates cellular contraction and matrix remodeling gene expression*. Journal of Biomedical Materials Research Part A, 2007. **83**(3): p. 887-894.
- 454. Smith, R.L., D.R. Carter, and D.J. Schurman, *Pressure and shear differentially alter human articular chondrocyte metabolism: A review*. Clinical Orthopaedics and Related Research, 2004(427 SUPPL.): p. S89-S95.
- 455. Sah, R.L.-Y., et al., *Biosynthetic response of cartilage explants to dynamic compression*. Journal of Orthopaedic Research, 1989. **7**(5): p. 619-636.

- 456. Gray, M.L., et al., *Mechanical and physicochemical determinants of the chondrocyte biosynthetic response.* Journal of Orthopaedic Research, 1988. **6**(6): p. 777-792.
- 457. Kelly, P.A. and J.J. O'Connor, *Transmission of rapidly applied loads through articular cartilage Part 1: uncracked cartilage*. ARCHIVE: Proceedings of the Institution of Mechanical Engineers, Part H: Journal of Engineering in Medicine 1989-1996 (vols 203-210), 1996. **210**(18): p. 27-37.
- 458. Wei, F., et al., *Effect of intermittent cyclic preloads on the response of articular cartilage explants to an excessive level of unconfined compression*. Journal of Orthopaedic Research, 2008. **26**(12): p. 1636-1642.
- 459. Isaac, D.I., E.G. Meyer, and R.C. Haut, *Chondrocyte damage and contact pressures* following impact on the rabbit tibiofemoral joint. Journal of Biomechanical Engineering, 2008. **130**(4).
- 460. Lewis, J.L., et al., *Cell death after cartilage impact occurs around matrix cracks*. Journal of Orthopaedic Research, 2003. **21**(5): p. 881-887.
- 461. Ewers, B.J., et al., *Rate of blunt impact loading affects changes in retropatellar cartilage and underlying bone in the rabbit patella*. Journal of Biomechanics, 2002. **35**(6): p. 747-755.
- 462. Ewers, B.J., et al., *The extent of matrix damage and chondrocyte death in mechanically traumatized articular cartilage explants depends on rate of loading.* Journal of Orthopaedic Research, 2001. **19**(5): p. 779-784.
- 463. Chin-Purcell, M.V. and J.L. Lewis, *Fracture of articular cartilage*. Journal of Biomechanical Engineering, 1996. **118**(4): p. 545-556.
- 464. Vanderploeg, E.J., et al., Oscillatory tension differentially modulates matrix metabolism and cytoskeletal organization in chondrocytes and fibrochondrocytes. Journal of Biomechanics, 2004. **37**(12): p. 1941-1952.
- 465. Hall, A.C., J.P.G. Urban, and K.A. Gehl, *The effects of hydrostatic pressure on matrix synthesis in articular cartilage*, in *Journal of Orthopaedic Research*. 1991. p. 1-10.
- 466. Hunter, C., J.K. Mouw, and M.E. Levenston, *Dynamic compression of chondrocyte-seeded fibrin gels: Effects on matrix accumulation and mechanical stiffness*, in *Osteoarthritis and Cartilage*. 2004. p. 117-130.
- 467. Jin, M., et al., Combined effects of dynamic tissue shear deformation and insulin-like growth factor I on chondrocyte biosynthesis in cartilage explants, in Archives of Biochemistry and Biophysics. 2003. p. 223-231.
- 468. Lee, D.A. and D.L. Bader, *Compressive strains at physiological frequencies influence the metabolism of chondrocytes seeded in agarose*, in *Journal of Orthopaedic Research*. 1997. p. 181-188.
- 469. Jin, M., et al., *Tissue shear deformation stimulates proteoglycan and protein biosynthesis in bovine cartilage explants*, in *Archives of Biochemistry and Biophysics*. 2001. p. 41-48.
- 470. Kim, Y.-J., et al., *Mechanical regulation of cartilage biosynthetic behavior: Physical stimuli*, in *Archives of Biochemistry and Biophysics*. 1994. p. 1-12.
- 471. Parkkinen, J.J., et al., Local stimulation of proteoglycan synthesis in articular cartilage explants by dynamic compression in vitro, in Journal of Orthopaedic Research. 1992. p. 610-620.
- 472. Sah, R.L.-Y., et al., *Biosynthetic response of cartilage explants to dynamic compression*, in *Journal of Orthopaedic Research*. 1989. p. 619-636.

- 473. Wu, Q.-Q., Y. Zhang, and Q. Chen, *Indian hedgehog Is an Essential Component of Mechanotransduction Complex to Stimulate Chondrocyte Proliferation*. Journal of Biological Chemistry, 2001. **276**(38): p. 35290-35296.
- 474. Honda, K., et al., *The effects of high magnitude cyclic tensile load on cartilage matrix metabolism in cultured chondrocytes.* European Journal of Cell Biology, 2000. **79**(9): p. 601-609.
- 475. Wu, Q.-Q. and Q. Chen, *Mechanoregulation of chondrocyte proliferation, maturation, and hypertrophy: Ion-channel dependent transduction of matrix deformation signals.* Experimental Cell Research, 2000. **256**(2): p. 383-391.
- 476. Fujisawa, T., et al., *Cyclic mechanical stress induces extracellular matrix degradation in cultured chondrocytes via gene expression of matrix metalloproteinases and interleukin-1.* Journal of Biochemistry, 1999. **125**(5): p. 966-975.
- 477. Connelly, J.T., E.J. Vanderploeg, and M.E. Levenston, *Influence of cyclic tension amplitude and frequency on chondrocyte matrix synthesis*, in *Transactions of the ORS*. 2004.
- 478. Uchino, T., et al., *Effect of squalene monohydroperoxide on cytotoxicity and cytokine release in a three-dimensional human skin model and human epidermal keratinocytes.* Biological and Pharmaceutical Bulletin, 2002. **25**(5): p. 605-610.
- 479. Gosset, M., et al., *Primary culture and phenotyping of murine chondrocytes*. Nature Protocols, 2008. **3**(8): p. 1253-1260.
- 480. Wong, M., M. Siegrist, and K. Goodwin, Cyclic tensile strain and cyclic hydrostatic pressure differentially regulate expression of hypertrophic markers in primary chondrocytes. Bone, 2003. **33**(4): p. 685-693.
- 481. Weiser, L., et al., *Effect of serum and platelet-derived growth factor on chondrocytes grown in collagen gels*, in *Tissue Engineering*. 1999. p. 533-544.
- 482. Iwasa, J., et al., *Effects of cell density on proliferation and matrix synthesis of chondrocytes embedded in atelocollagen gel*, in *Artificial Organs*. 2003. p. 249-255.
- 483. Akagi, M., et al., Cyclic tensile stretch load and oxidized low density lipoprotein synergistically induce lectin-like oxidized LDL receptor-1 in cultured bovine chondrocytes, resulting in decreased cell viability and proteoglycan synthesis. Journal of Orthopaedic Research, 2006. **24**(8): p. 1782-1790.
- 484. Chaipinyo, K., B.W. Oakes, and M.-P.I. Van Damme, *Effects of growth factors on cell proliferation and matrix synthesis of low-density, primary bovine chondrocytes cultured in collagen I gels*, in *Journal of Orthopaedic Research*. 2002. p. 1070-1078.
- 485. Galois, L., et al., Bovine chondrocyte behaviour in three-dimensional type I collagen gel in terms of gel contraction, proliferation and gene expression. Biomaterials, 2006. 27(1): p. 79-90.
- 486. Duprez, D., et al., Overexpression of BMP-2 and BMP-4 alters the size and shape of developing skeletal elements in the chick limb. Mechanisms of Development, 1996. 57(2): p. 145-157.
- 487. Xie, J., et al., Mechanical loading-dependence of mRNA expressions of extracellular matrices of chondrocytes inoculated into elastomeric microporous poly(L-lactide-co-b-caprolactone) scaffold. Tissue Engineering, 2007. **13**(1): p. 29-40.
- 488. Chokalingam, K., et al., 3D-In vitro Effects of Compression and Time in Culture on Aggregate Modulus and on Gene Expression and Protein content of Collagen Type II in Murine Chondrocytes

- Tensile Stimulation of Murine Stem Cell-Collagen Sponge Constructs Increases Collagen Type I Gene Expression and Linear Stiffness
- Using functional tissue engineering and bioreactors to mechanically stimulate tissue-engineered constructs. Tissue Eng Part A, 2009. **15**(4): p. 741-9.
- 489. Kim, Y.J., L.J. Bonassar, and A.J. Grodzinsky, *The role of cartilage streaming potential*, *fluid flow and pressure in the stimulation of chondrocyte biosynthesis during dynamic compression*. Journal of Biomechanics, 1995. **28**(9): p. 1055-1066.
- 490. Archambault, J., et al., *Stretch and interleukin-1? induce matrix metalloproteinases in rabbit tendon cells in vitro*. Journal of Orthopaedic Research, 2002. **20**(1): p. 36-39.
- 491. Nirmalanandhan, V.S., et al., *Effect of length of the engineered tendon construct on its structure-function relationships in culture*. Journal of Biomechanics, 2007. **40**(11): p. 2523-2529.
- 492. Erickson, I.E., et al., *Differential maturation and structure-function relationships in mesenchymal stem cell- and chondrocyte-seeded hydrogels*. Tissue Engineering Part A, 2009. **15**(5): p. 1041-1052.
- 493. Huang, A.H., M.J. Farrell, and R.L. Mauck, *Mechanics and mechanobiology of mesenchymal stem cell-based engineered cartilage*. Journal of Biomechanics, 2009.
- 494. Huang, A.H., et al., Identification of molecular antecedents limiting functional maturation of MSC-laden hydrogels, in Trans of the 54th Annual Meeting of the Orthopaedic Research Society. 2008.
- 495. Mauck, R.L., X. Yuan, and R.S. Tuan, *Chondrogenic differentiation and functional* maturation of bovine mesenchymal stem cells in long-term agarose culture, in Osteoarthritis and Cartilage. 2006. p. 179-189.
- 496. Chan, B.P., et al., *Effects of basic fibroblast growth factor (bFGF) on early stages of tendon healing: a rat patellar tendon model.* Acta Orthop Scand, 2000. **71**(5): p. 513-8.
- 497. Abrahamsson, S., G. Lundborg, and L. Lohmander, *Recombinant human insulin-like* growth factor-I stimulates in vitro matrix synthesis and cell proliferation in rabbit flexor tendon. J Orthop Res, 1991. **9**(Jul): p. 495-502.
- 498. Abrahamsson, S., G. Lundborg, and L. Lohmander, *Long-term explant culture of rabbit flexor tendon: effects of recombinant human insulin-like growth factor-I and serum on matrix metabolism.* J Orthop Res, 1991. **9**(Jul): p. 503-15.
- 499. Nirmalanandhan, V.S., et al., *Improving linear stiffness of the cell-seeded collagen* sponge constructs by varying the components of the mechanical stimulus. Tissue Engineering - Part A., 2008. **14**(11): p. 1883-1891.
- 500. Shearn, J.T., et al., Mechanical Stimulation of Tendon Tissue Engineered Constructs: Effects on Construct Stiffness, Repair Biomechanics and their Correlation. J Biomech Eng, in press.
- 501. Juncosa-Melvin, N., et al., *Effects of mechanical stimulation on the biomechanics and histology of stem cell-collagen sponge constructs for rabbit patellar tendon repair.* Tissue Eng, 2006. **12**(8): p. 2291-300.
- 502. Juncosa-Melvin, N., et al., *Effects of cell-to-collagen ratio in mesenchymal stem cell-seeded implants on tendon repair biomechanics and histology*. Tissue Eng, 2005. **11**(3-4): p. 448-57.
- 503. Ouyang, H.W., et al., *Knitted poly-lactide-co-glycolide scaffold loaded with bone marrow stromal cells in repair and regeneration of rabbit achilles tendon*. Tissue Engineering, 2003. **9**(3): p. 431-439.

- 504. Spalazzi, J.P., et al., *Mechanoactive scaffold induces tendon remodeling and expression of fibrocartilage markers*. Clin Orthop Relat Res, 2008. **466**(8): p. 1938-48.
- 505. Van Eijk, F., et al., *The effect of timing of mechanical stimulation on proliferation and differentiation of goat bone marrow stem cells cultured on braided PLGA scaffolds.* Tissue Engineering Part A., 2008. **14**(8): p. 1425-1433.
- 506. van Eijk, F., et al., *IN VIVO MATRIX PRODUCTION BY BMSCs SEEDED ON PLGA SCAFFOLDS FOR LIGAMENT TISSUE ENGINEERING*. Tissue Eng Part A, 2009.
- 507. Popov, E.P., *Engineering mechanics of solids*. 1990, Englewood Cliffs, N.J.: Prentice Hall. 727.